UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D. C. 20549

FORM 10-Q

(Mark	One)
×]

QUARTERLY REPORT PURSUANT TO SECTION 13 or 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended September 30, 2003

OR

TRANSITION REPORT PURSUANT TO SECTION 13 or 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
For the transition period from to
Commission File Number: 001-06605

EQUIFAX INC.

(Exact name of registrant as specified in its charter)

Georgia (State or other jurisdiction of incorporation or organization) 58-0401110 (I.R.S. Employer Identification No.)

1550 Peachtree Street, N.W. Atlanta, Georgia (Address of principal executive offices)

30309 (Zip Code)

404-885-8000 (Registrant's telephone number, including area code)

N/A

(Former name, former address and former fiscal year, if changed since last report)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15 (d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \boxtimes No \square

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act). Yes 🗷 No 🗆

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date.

Class

Common Stock, \$1.25 Par Value

Outstanding at September 30, 2003

139,076,780

Page No.

EQUIFAX INC. FORM 10-Q QUARTERLY PERIOD ENDED SEPTEMBER 30, 2003

INDEX

PART I	Financial Information (Unaudited)	
Item 1.	Financial Statements	
	Consolidated Balance Sheets—September 30, 2003 and December 31, 2002	3
	Consolidated Statements of Income—Three Months Ended September 30, 2003 and 2002	4
	Consolidated Statements of Income—Nine Months Ended September 30, 2003 and 2002	5
	Consolidated Statements of Cash Flows—Nine Months Ended September 30, 2003 and 2002	6
	Consolidated Statements of Shareholders' Equity and Comprehensive Income—Nine Months Ended September 30, 2003	7
	Notes to Consolidated Financial Statements	8-15
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	16-31
	·	

Item 3.	Quantitative and Qualitative Disclosures About Market Risk			
Item 4.	Controls and Procedures	32		
PART II	Other Information			
Item 1.	Legal Proceedings	33		
Item 6.	Exhibits and Reports on Form 8-K	34		
Signatures		35		
Index to Ex	xhibits	36		

PART I. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS.

EQUIFAX INC. CONSOLIDATED BALANCE SHEETS

	September 30, 2003			December 31, 2002	
In millions, except par values)	(Unau	dited)			
SSETS					
Current Assets:					
Cash and cash equivalents	\$	33.5	\$	30	
Trade accounts receivable, net of allowance for doubtful accounts of \$12.7 in 2003 and \$17.3 in 2002		191.4		179	
Other receivables		9.4		20	
Deferred income tax assets		19.6		20	
Other current assets		39.4		33	
Total current assets		293.3		285	
roperty and Equipment:					
Land, buildings and improvements		31.1		25	
Data processing equipment and furniture		121.4		11:	
. , ,					
		152.5		14	
Less accumulated depreciation		104.6		9.	
		47.9		5	
Goodwill		702.5		65	
urchased Data Files		262.8		26	
hther Assets Assets of Discontinued Operations (Note 4)		285.9 1.0		24	
	\$	1,593.4	\$	1,50	
LIBH WING AND GHADDHOLDEDGE DOLLTON					
IABILITIES AND SHAREHOLDERS' EQUITY urrent Liabilities:					
urrent Liabilities:	S	27 1	\$	23	
Short-term debt and current maturities	\$	27.1	\$		
urrent Liabilities: Short-term debt and current maturities Accounts payable	s	10.8	\$	1	
urrent Liabilities: Short-term debt and current maturities Accounts payable Accrued salaries and bonuses	s	10.8 28.6	s	1	
urrent Liabilities: Short-term debt and current maturities Accounts payable	s	10.8	s	1	
Short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities	s	10.8 28.6 154.7	s	23 1 3 14	
urrent Liabilities: Short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities Total current liabilities	s	10.8 28.6 154.7	s	1 3 14	
urrent Liabilities: Short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities Total current liabilities ong-Term Debt eferred Revenue	s	28.6 154.7 221.2 841.5 13.1	s	1 3 14 42 69 1	
urrent Liabilities: Short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities Total current liabilities ong-Term Debt eferred Revenue eferred Income Tax Liabilities	s	28.6 154.7 221.2 841.5 13.1 38.3	s	14 42 69 1	
short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities Total current liabilities ong-Term Debt eferred Revenue eferred Income Tax Liabilities ther Long-Term Liabilities	S	28.6 154.7 221.2 841.5 13.1	S	142 42 69 1 2	
urrent Liabilities: Short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities Total current liabilities Total current liabilities ong-Term Debt eleferned Revenue eferred Revenue eferred Income Tax Liabilities ther Long-Term Liabilities iabilities of Discontinued Operations (Note 4)	S	28.6 154.7 221.2 841.5 13.1 38.3 126.9 5.7	s	14 42 69 1 2	
urrent Liabilities: Short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities Total current liabilities ong-Term Debt eferred Revenue eferred Income Tax Liabilities eferred Income Tax Liabilities indicates of Discontinued Operations (Note 4)	S	28.6 154.7 221.2 841.5 13.1 38.3 126.9	s	14 42 69 1 2	
urrent Liabilities: Short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities Total current liabilities ong-Term Debt eferred Revenue eferred Income Tax Liabilities ther Long-Term Liabilities ther Long-Term Liabilities other Long-Term Liabilities of Discontinued Operations (Note 4)	S	28.6 154.7 221.2 841.5 13.1 38.3 126.9 5.7	S	14 42 69 1 2	
urrent Liabilities: Short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities Total current liabilities Total current liabilities ong-Term Debt eferred Revenue eferred Revenue eferred Income Tax Liabilities ther Long-Term Liabilities ther Long-Term Liabilities Total liabilities of Discontinued Operations (Note 4) Total liabilities Total liabilities Total liabilities Ommitments and Contingencies (Note 10) hareholders' Equity: Preferred stock, \$0.01 par value: Authorized—10.0; Issued—none Common stock, \$1.25 par value: Authorized shares—300.0 Issued shares—180.1 in 2003 and 180.1 in 2002 Outstanding shares—134.9 in 2003 and	S	28.6 154.7 221.2 841.5 13.1 38.3 126.9 5.7	S	1 3 14 42 69 1 2 12	
Short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities Total current liabilities ong-Term Debt eferred Revenue eferred Income Tax Liabilities ther Long-Term Liabilities abilities of Discontinued Operations (Note 4) Total liabilities Total liabilities Total liabilities Other current liabilities Other cu	S	10.8 28.6 154.7 221.2 841.5 13.1 38.3 126.9 5.7 1,246.7	S	1 3 14 42 69 1 2 12	
Short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities Total current liabilities Total current liabilities ong-Term Debt eferred Revenue eferred Income Tax Liabilities ther Long-Term Liabilities abilities of Discontinued Operations (Note 4) Total liabilities Total liabilities liabilities Total liabilities	S	28.6 154.7 221.2 841.5 13.1 38.3 126.9 5.7	S	42 42 69 12 12	
Short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities Total current liabilities ong-Term Debt eferred Revenue eferred Income Tax Liabilities ther Long-Term Liabilities abilities of Discontinued Operations (Note 4) Total liabilities Total liabilities Total liabilities Other current liabilities Other cu	S	10.8 28.6 154.7 221.2 841.5 13.1 38.3 126.9 5.7 1,246.7	S	14 42 69 1 2 12 1,28	
Short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities Total current liabilities mg-Term Debt eferred Revenue eferred Income Tax Liabilities her Long-Term Liabilities abilities of Discontinued Operations (Note 4) Total liabilities pumitments and Contingencies (Note 10) tareholders' Equity: Preferred stock, \$0.01 par value: Authorized—10.0; Issued—none Common stock, \$1.25 par value: Authorized shares—300.0 Issued shares—180.1 in 2003 and 180.1 in 2002 Outstanding shares—134.9 in 2003 and 135.7 in 2002 Paid-in capital	S	10.8 28.6 154.7 221.2 841.5 13.1 38.3 126.9 5.7 1,246.7 — 225.2 422.3	S	1 44 69 1 5 1 12 1,24 4 92	
Short-term debt and current maturities Accounts payable Accrued salaries and bonuses Other current liabilities Total current liabilities mg-Term Debt eftered Revenue efterred Income Tax Liabilities efter Long-Term Liabilities ther Long-Term Liabilities Total liabilities of Discontinued Operations (Note 4) Total liabilities Total representation of the state o	S	10.8 28.6 154.7 221.2 841.5 13.1 38.3 126.9 5.7 1,246.7 — 225.2 422.3 1,053.8	S	42 69 1 2 1,28	

 Total shareholders' equity
 346.7
 221.0

 \$
 1,593.4
 \$
 1,506.9

See Notes to Consolidated Financial Statements

3

EQUIFAX INC. CONSOLIDATED STATEMENTS OF INCOME (UNAUDITED)

		Three Months Ended September 30,		
	2003			2002
(In millions, except per share amounts)	\$ 200	9.8	¢.	289.7
Operating revenue Costs and expenses:	\$ 309	1.8	\$	289.7
Costs of services	129	9.0		113.2
Selling, general and administrative expenses		6.8		66.3
Depreciation Depreciation		4.1		3.7
Amortization		1.0		16.8
		_		
Total costs and expenses	220	0.9		200.0
20.00.000.000.000		_		
Operating income	88	8.9		89.7
Other income, net		3.0		2.0
Minority interest in earnings, net of tax	(1	0.9)		0.2
Interest expense	(3	8.6)		(10.4)
		_		
Income from continuing operations before income taxes		2.4		81.5
Provision for income taxes	(29	9.6)		(31.8)
I		_		40.7
Income from continuing operations	J.	2.8		49.7
Discontinued operations (Note 4):				
Loss from discontinued operations, net of income tax benefit of \$0 in 2003 and \$1.6 in 2002	(1.6)		(10.8)
		—		
Net income	\$ 5	1.2	\$	38.9
		_		
Per common share (basic):				
Income from continuing operations	\$ 0.	.39	\$	0.37
Discontinued operations	(0.	.01)		(0.08)
		_		
Net income	\$ 0.	.38	\$	0.29
		_		
Shares used in computing basic earnings per share	134	4.5		135.9
Per common share (diluted):				
Income from continuing operations	\$ 0.	.39	\$	0.36
Discontinued operations	(0.	.01)		(0.08)
		_		
Net income	\$ 0.	.38	\$	0.28
Shares used in computing diluted earnings per share	13	5.4		137.6
		_		
Dividends per common share	\$ 0.	.02	\$	0.02

See Notes to Consolidated Financial Statements.

4

EQUIFAX INC.

CONSOLIDATED STATEMENTS OF INCOME (UNAUDITED)

Nine Months Ended September 30,

2003 2002

(In millions, except per share amounts)

Operating revenue	\$	928.4	\$	816.7
Costs and expenses:				
Costs of services		393.7		323.0
Selling, general and administrative expenses		207.4		179.0
Depreciation		12.4		10.9
Amortization		60.1		47.2
Total costs and expenses		673.6		560.1
Operating income		254.8		256.6
Other income, net		12.3		5.3
Minority interest in earnings, net of tax		(2.6)		(1.3)
Interest expense		(31.1)		(30.5)
Income from continuing operations before income taxes		233.4		230.1
Provision for income taxes		(86.2)		(90.5)
110 vision for meonic taxes		(00.2)		(70.5)
Income from continuing operations	_	147.2		139.6
Discontinued operations (Note 4):				
Loss from discontinued operations, net of income tax benefit of \$0 in 2003 and \$2.2 in 2002		(10.3)		(11.5)
Net income	\$	136.9	\$	128.1
Per common share (basic):				
Income from continuing operations	\$	1.09	\$	1.02
Discontinued operations		(0.08)		(0.08)
Net income	\$	1.01	\$	0.94
Shares used in computing basic earnings per share		134.9		136.4
Per common share (diluted):				
Income from continuing operations	\$	1.08	\$	1.01
Discontinued operations	Ψ	(0.08)	Ψ	(0.08)
Net income	\$	1.00	s	0.92*
	_		_	
Shares used in computing diluted earnings per share		136.8		138.9
Dividends per common share	\$	0.06	\$	0.06

Does not total due to rounding.

See Notes to Consolidated Financial Statements.

5

EQUIFAX INC. CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED)

	Nine Months Ended September 30,			
		2003	2	002
(In millions)				
Cash flows from operating activities:				
Net income	\$	136.9	\$	128.1
Adjustments to reconcile net income to net cash provided by operating activities of continuing operations:				
Loss from discontinued operations		9.8		11.5
		72.5		58.1
Depreciation and amortization				
Deferred income taxes		10.5		9.8
Changes in assets and liabilities, excluding effects of acquisitions:				
Accounts receivable, net		(1.0)		13.1
Current liabilities, excluding debt		(13.8)		(33.8)
Other current assets		1.7		1.6
Other long-term liabilities, excluding debt		4.4		(5.7)
Other assets		(27.5)		(32.7)

Cash provided by operating activities	193.5	150.0
Investing activities:		
Additions to property and equipment	(11.4)	(8.5)
Additions to other assets, net	(25.3)	(26.9)
Acquisitions, net of cash acquired	(40.7)	(218.1)
Investments in unconsolidated affiliates	_	(0.1)
Proceeds on note receivable from sale of business	_	41.0
Deferred payments on prior year acquisitions	(1.4)	(4.9)
Cash used by investing activities	(78.8)	(217.5)
Financing activities:		
Net short-term payments	(11.5)	(15.2)
Additions to long-term debt	155.3	137.9
Payments on long-term debt	(202.5)	_
Treasury stock purchases	(65.3)	(78.7)
Dividends paid	(8.4)	(8.6)
Proceeds from exercise of stock options	16.7	31.6
Other	0.2	0.6
Cash (used) provided by financing activities	(115.5)	67.6
Effect of foreign currency exchange rates on cash	3.3	(3.3)
Cash provided by discontinued operations	0.5	0.1
Increase (decrease) in cash and cash equivalents	3.0	(3.1)
Cash and cash equivalents, beginning of period	30.5	33.2
Cash and cash equivalents, end of period	\$ 33.5 \$	30.1

See Notes to Consolidated Financial Statements.

6

EQUIFAX INC. CONSOLIDATED STATEMENTS OF SHAREHOLDERS' EQUITY AND COMPREHENSIVE INCOME

	Common	Common Stock:							Stock	
	Shares Outstanding	Amount	Paid-In Capital				ccumulated Other mprehensive oss) Income	Treasury Stock	Held By Employee Benefits Trusts	Total Shareholders' Equity
(In millions)										
Balance, December 31, 2002	135.7	\$ 225.1	\$ 412.	0 \$	925.4	\$	(359.4) \$	(899.7) \$	(82.4)	221.0
Net income	_	_	_	_	136.9		_	_	_	136.9
Other comprehensive income	_	_	_	_	_		43.7	_	_	43.7
Shares issued under stock plans	1.2	0.4	10.	0	_		_	_	8.5	18.9
Treasury stock purchased	(2.9)	_	-	-	_		_	(65.3)	_	(65.3)
Restricted stock cancelled	(0.3)	(0.3)) –	_	_		_	_	_	(0.3)
Cash dividends	_	_	_	_	(8.5)		_	_	_	(8.5)
Dividends from employee benefits trusts			0	3 _						0.3
Balance, September 30, 2003 (Unaudited)	133.7	\$ 225.2	\$ 422	3 \$	1,053.8	\$	(315.7) \$	(965.0) \$	(73.9)	346.7

Accumulated Other Comprehensive Loss consists of the following components:

	September 30, 2003		December 31, 2002
	(Unaudited)		
Foreign currency translation	\$ (196.0)	\$	(239.6)
Minimum pension liability, net of tax of \$70.2 in 2003 and 2002 Cash flow hedging transactions, net of tax of \$1.8 in 2003 and \$1.9 in 2002	(117.0)		(117.0) (2.8)
		_	
	\$ (315.7)	\$	(359.4)

Comprehensive Income is as follows:

Three Months Ended			Nine Mon	ths Ended
September 30,			Septen	aber 30,
2003	2002		2003	

	(Unaudited)	(Unaudited)		(Unaudited)		(Unaudited)
Net income	\$ 51.2	\$	38.9	\$ 136.9	\$	128.1
Other comprehensive income (loss):						
Foreign currency translation adjustment	1.2		(33.9)	43.5		(60.2)
Change in cumulative loss from cash flow hedging transactions	0.6		(1.4)	0.2		(2.0)
		_			_	
	\$ 53.0	\$	3.6	\$ 180.6	\$	65.9

See Notes to Consolidated Financial Statements.

7

EQUIFAX INC. NOTES TO CONSOLIDATED FINANCIAL STATEMENTS (UNAUDITED) September 30, 2003

As used herein, the terms "Equifax", "the Company", "we", "our", and "us" refer to Equifax Inc., a Georgia corporation, and its consolidated subsidiaries as a combined entity, except where it is clear that the terms mean only Equifax Inc.

1. BASIS OF PRESENTATION

We have prepared the accompanying unaudited consolidated financial statements in accordance with accounting principles generally accepted in the United States for interim financial information and with the instructions to Form 10-Q and Rule 10-01 of Regulation S-X. This information reflects all adjustments, which are in the opinion of management, necessary for a fair presentation of the statement of financial position as of September 30, 2003, and the results of operations for the three and nine month periods ending September 30, 2003 and 2002. All adjustments made have been of a normal recurring nature. Certain information and footnote disclosures normally included in financial statements prepared in accordance with accounting principles generally accepted in the U.S., or GAAP, have been condensed or omitted. We believe that the disclosures are adequate to make the information presented not misleading. Certain prior year amounts have been reclassified to conform to the current year presentation. These consolidated financial statements should be read in conjunction with the consolidated financial statements and the notes thereto included in our annual report on Form 10-K for the fiscal year ended December 31, 2002. That report includes a summary of our critical accounting policies. There have been no material changes in the accounting policies followed by us during fiscal 2003.

The preparation of financial statements in conformity with GAAP requires our management to make estimates and assumptions. These estimates and assumptions affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements as well as reported amounts of revenues and expenses during the reporting period. Actual results could differ from these estimates.

2. NATURE OF OPERATIONS

We collect, organize and manage various types of financial, demographic, and marketing information. Our products and services enable businesses to make credit and service decisions, manage their portfolio risk, and develop marketing strategies concerning consumers and commercial enterprises. We serve customers across a wide range of industries, including the financial services, mortgage, retail, telecommunications, utilities, automotive, brokerage, healthcare and insurance industries, as well as state and federal governments. We also enable consumers to manage and protect their financial health through a portfolio of products offered directly to individuals. We have approximately 4,500 employees worldwide, and manage our business globally through the following three operating segments: Equifax North America, Equifax Europe, and Equifax Latin America. Our operations are predominantly located within the United States, with foreign operations principally located in Canada, the United Kingdom, and Brazil.

Our products and services are categorized as follows: Information Services, Marketing Services, and Consumer Direct. Our Information Services products and services allow customers to make credit decisions about consumers and commercial enterprises. Our Marketing Services information products and databases enable customers to identify a target audience for marketing purposes, and our Consumer Direct products and services provide information to consumers which enable them to reduce their exposure to identity fraud and to monitor their credit health.

8

We develop, maintain, and enhance secured proprietary information databases through compilation of accounts receivable information about consumers and businesses that we obtain from a variety of sources, such as credit granting institutions, public record information, including bankruptcies, liens, and judgments, and marketing information from surveys and warranty cards. We process this information utilizing our proprietary information management systems and make it available to our customers in virtually any medium or format they choose.

3. ACQUISITIONS

During the first nine months of 2003, we acquired the credit files, contractual rights to territories, customer relationships and related businesses of four independent credit reporting agencies in the United States and one in Canada that house consumer information on our system ("Affiliate"). Additionally, in April 2003 we completed the purchase of a small eMarketing business for \$10.0 million. We acquired all of these businesses for \$42.9 million primarily in cash, allocating \$21.3 million of the purchase price to goodwill, \$15.5 million to purchased data files, and \$6.2 million to non-compete agreements. These allocations are preliminary estimates and will be finalized upon completion of the purchase valuation. In the case of the eMarketing business, the preliminary purchase price allocation does not include the effect of management's plan to involuntarily terminate certain employees of the acquired company. The results of operations for these acquisitions have been included in the consolidated statement of income from the date of acquisition, and have not been material.

4. DISCONTINUED OPERATIONS

During the third quarter of 2002, we made the decision to exit the commercial services business in Spain. The business is currently held for sale, with the expectation that it will be sold in the near future. In accordance with the Financial Accounting Standards Board's ("FASB") Statement of Financial Accounting Standards ("SFAS") No. 144 the net assets, results of operations and cash flows of the Spain commercial business have been classified as "discontinued operations." For the third quarter of 2003 and 2002, revenues totaled \$1.5 million and \$1.6 million respectively. Our loss on discontinued operations, net of income taxes, was \$1.6 million for the third quarter of 2003 and \$10.8 million for the third quarter of 2002. For the first nine months of 2003 and 2002, revenues totaled \$6.0 million and \$6.3 million respectively. Our loss on discontinued operations, net of income taxes, was \$10.3 million for the first nine months of 2003 and \$11.5 million for the first nine months of 2002. In the quarter ended September 30, 2003, we entered into a

letter of intent with a prospective buyer. We are still negotiating with the prospective buyer and will explore other options should our negotiation prove unsuccessful. The results of the Spain commercial business, for the nine months ended September 30, 2003, include our updated estimate of loss upon final disposition. The estimated loss on disposal recorded in the third quarter of 2002 totaled \$9.0 million.

5. RESTRUCTURING AND IMPAIRMENT CHARGES

In the fourth quarter of 2001, we recorded restructuring and impairment charges of \$60.4 million (\$35.3 million after tax or \$0.25 per diluted share). The restructuring charges, which total \$37.2 million, were associated with the alignment of our cost structure with changing market conditions to reduce expenses and improve efficiencies, particularly in international operations. The restructuring plan, which included headcount reductions of approximately 700 employees, consisted of severance costs and reserves to reflect our estimated exposure on facilities to be vacated or consolidated. The asset impairment charges, which total \$23.2 million, reflected our write-down of several technology investments. At December 31, 2002, our remaining reserve balance was \$16.3 million. During the first nine months of 2003, we paid \$2.9 million against the reserve, leaving a remaining balance of

9

\$13.4 million. An analysis of 2003 activity in the reserve through September 30 is as follows (in millions):

Total
\$ 16.3
(2.9)
\$ 13.4
\$

6. LONG-TERM DEBT AND SHORT-TERM BORROWINGS

In October 2002, we issued new 4.95% fixed rate five-year unsecured notes with a face value of \$250.0 million. The notes, which expire in 2007, were sold at a discount of \$0.5 million. The discount, and related issuance costs, will be amortized on a straight-line basis over the term of the notes. Our \$200.0 million 6.5% senior unsecured notes, originally issued in 1993, matured June 2003. We borrowed \$200.0 million under our revolving credit facility to retire the maturing notes. The indebtedness evidenced by our 4.95% notes, our 6.3% notes, and our 6.9% debentures, none of which has been guaranteed by any of our subsidiaries, is unsecured, and ranks on parity in right of payment with all of our other unsecured and unsubordinated indebtedness from time to time outstanding. Long-term debt at September 30, 2003 and December 31, 2002 was as follows:

		September 30, 2003		December 31, 2002
(In millions)			_	
Senior Notes, 6.5%, due 2003, net of unamortized discount of \$0.0 million in 2003 and \$0.1 million				
in 2002	\$	_	\$	199.9
Notes, 6.3%, due 2005, net of unamortized discount of \$0.3 million in 2003 and \$0.4 million in 2002		249.7		249.6
Notes, 4.95%, due 2007, net of unamortized discount of \$0.4 million in 2003 and \$0.5 million in				
2002		249.6		249.5
Debentures, 6.9%, due 2028, net of unamortized discount of \$1.3 million in 2003 and \$1.3 million in				
2002		148.8		148.7
Borrowings under revolving credit facilities, weighted average rate of 1.6% at September 30, 2003				
and 2.6% at December 31, 2002		178.1		21.8
Other		17.1		22.4
			_	
		843.3		891.9
Less current maturities		1.8		201.3
			_	
	\$	841.5	\$	690.6
	Ψ	041.5	Ψ	070.0

7. INCOME TAXES

Effective Tax Rates. The effective tax rates from continuing operations were 37% in 2003 and 39% in 2002. The decrease in the effective rate over time is attributable to the implementation of state tax planning strategies related to the apportionment of state income taxes and franchise taxes.

Deferred Tax Assets. We estimate levels of future taxable income and utilize prudent and feasible tax planning strategies in establishing and maintaining deferred tax assets. If we are unable to realize all or part of our deferred tax assets in the future, our effective tax rate could increase.

10

8. SHAREHOLDERS' EQUITY

Treasury Stock. During the first nine months of 2003, we repurchased 2,938,800 shares at an average price of \$22.22 through open market transactions at an aggregate investment of \$65.3 million. As of September 30, 2003, approximately \$156.9 million remained authorized for future share repurchases.

9. STOCK-BASED COMPENSATION

In accordance with the provisions of SFAS No. 123, "Accounting for Stock-Based Compensation" (SFAS No. 123), we have elected to apply APB Opinion No. 25 and related interpretations in accounting for our stock option and performance share plans. Accordingly, we do not recognize compensation cost in connection with our stock option plans and record compensation expense related to our performance share plan based on the current market price of our common stock and the extent to which performance criteria are being met. If we had elected to recognize compensation cost for these plans based on the fair value at grant date as prescribed by SFAS No. 123, net income and net income per share would have been reduced to the pro forma amounts indicated in the table below (in millions, except per share amounts):

		Third Quarter				Nine Months			
	2003 2002			2003		2002			
(In millions, except per share data)	_		_		_		_		
Net income, as reported	\$	51.2	\$	38.9	\$	136.9	\$	128.1	
Deduct: Total stock-based employee compensation expense determined under fair value based method for all awards, net of related tax effects		(2.0)		(2.8)		(11.3)		(12.0)	
Pro forma net income	\$	49.2	\$	36.1	\$	125.6	\$	116.1	
Earnings per share:									
Basic—as reported	\$	0.38	\$	0.29	\$	1.01	\$	0.94	
Basic—pro forma	\$	0.37	\$	0.27	\$	0.93	\$	0.85	
Diluted—as reported	\$	0.38	\$	0.28	\$	1.00	\$	0.92	
Diluted—pro forma	\$	0.36	\$	0.26	\$	0.92	\$	0.84	

10. COMMITMENTS AND CONTINGENCIES

Agreement with Computer Sciences Corporation. We have an agreement with Computer Sciences Corporation and certain of its affiliates ("CSC") under which CSC-owned credit reporting agencies utilize our computerized credit database services. CSC retains ownership of its credit files and the revenues generated by its credit reporting activity. We receive a processing fee for maintaining the database and for each report supplied. The agreement was renewed by CSC for a ten-year period beginning August 1, 1998. The agreement provides us with an option to purchase CSC's credit reporting business if CSC does not elect to renew the agreement or if there is a change in control of CSC while the agreement is in effect. Under the agreement CSC also has an option, exercisable at any time, to sell its credit reporting business to us. The option expires in 2013. The option exercise price will be determined by a third-party appraisal process and would be due in cash within 180 days after the exercise of the option. We estimate that if the option were exercised at this time, the price would fall within a range of approximately \$650.0 to \$700.0 million. This estimate is based solely on our internal analysis of the value of the businesses, market conditions at the time of our analysis, and other factors, all of which are subject to constant change. Therefore, the actual option exercise price could be materially higher or lower than the estimated amount. If CSC were to exercise its option, we would have to obtain additional sources of funding. We believe that this funding would be available from sources such as additional bank lines of credit and the issuance of public debt and/or equity. However,

11

the availability and terms of any such capital financing would be subject to a number of factors, including credit market conditions, the state of the equity markets, general economic conditions, and our financial performance and condition. See the "Risk Factors" section of the "Management's Discussion and Analysis" included in our annual report on Form 10-K for the year ended December 31, 2002.

Data Processing and Outsourcing Services Agreements. We have separate agreements with IBM, Polk/Acxiom, Seisint Inc., Xerox Connect, Inc., and Jones Lang LaSalle which outsource portions of our computer data processing operations and related functions and certain administrative functions. The agreements expire between 2004 and 2013. The estimated aggregate contractual obligation remaining under these agreements is \$478.8 million as of September 30, 2003, with no future year expected to exceed \$106.3 million. However, these amounts could be more or less depending on various factors such as the inflation rate, the introduction of significant new technologies, or changes in our servicing needs as a result of acquisitions or divestitures. Under certain circumstances (e.g., a change in control, or for our convenience), we may terminate these agreements. However, some of the agreements provide that we must pay a significant termination charge in the event of such a termination. See "Summary of Selected Recent Events" on page 18 for a description of the IBM agreement.

Guarantees. We will from time to time issue standby letters of credit, performance bonds or other guarantees in the normal course of business. The aggregate notional amount of all performance bonds and standby letters of credit is less than \$15.0 million and all have a maturity of less than one year. Guarantees are issued from time to time to support the needs of operating units. The only outstanding guarantee that is not reflected as a liability on our balance sheet was extended in connection with the sale of our risk management collections business to RMA Holdings, LLC ("RMA") in October 2000, at which time we guaranteed the operating lease payments of a partnership affiliated with RMA to a lender of the partnership pursuant to a term loan. The term loan, which had \$7.4 million outstanding as of September 30, 2003, expires December 1, 2011. Our obligations under such guarantee are not secured. We believe that the likelihood of demand for payment under these instruments is minimal and expect no material losses to occur in connection with these instruments.

Litigation. We are involved in various lawsuits, claims, and legal proceedings arising in the normal course of our business that are described in our 2002 Annual Report on Form 10-K and our quarterly reports on Form 10-Q for the periods ended March 31, 2003 and June 30, 2003. During the three months ended September 30, 2003, there were no material developments in such lawsuits, claims and legal proceedings.

11. SEGMENT INFORMATION

Our operations are primarily organized in four reportable segments, with three segments based on the provision of our three core product lines (Information Services, Marketing Services and Consumer Direct) within geographic regions (Equifax North America, Equifax Europe, and Equifax Latin America), and one segment based on other criteria (Other). We evaluate the segment performance based on operating income before unusual items (if any). Intersegment sales and transfers are not material. The measurements of segment profit or loss and segment assets for each reportable segment are substantially the same. All transactions between segments are accounted for at cost, and no timing differences occur between segments.

The operating results of our Commercial Services business in Spain have been reclassified to discontinued operations (Note 4) and are not included in Equifax Europe's segment results below

12

A description of segment products and services is as follows:

mortgage loan origination information; analytics and consulting; decisioning solutions, identity verification services, and commercial services, in the United States and Canada; Marketing Services, which consist of credit card marketing services and the provision of consumer demographic and lifestyle information; and Consumer Direct, which consists of credit and finance products sold directly to individuals.

Equifax Europe: Information Services, which includes Consumer and Commercial Services, such as credit, credit scoring and modeling services; and Credit Marketing Services, and products sold directly to consumers through Consumer Direct Services.

Equifax Latin America: Information Services, which includes Consumer and Commercial Services such as credit and other commercial, financial, and consumer information

Other: Lottery services relating solely to a contract to provide services to the state of California. No further revenue or operating income has been received since the second quarter 2002 or is expected to occur in this segment.

Operating revenue and operating income by segment for the third quarter and first nine months of 2003 and 2002 are as follows:

		Third Quarter					Nine Months			
	_	2003		2002		2003		2002		
(In millions)			_		_					
Operating Revenue:										
Equifax North America	\$	256.6	\$	238.6	\$	776.9	\$	657.8		
Equifax Europe		31.7		32.8		94.4		95.6		
Equifax Latin America		21.5		18.3		57.1		58.9		
Other		_		_		_		4.4		
	_		_				_			
	\$	309.8	\$	289.7	\$	928.4	\$	816.7		
	_									
Operating Income (Loss):										
Equifax North America	\$	89.4	\$	95.7	\$	270.2	\$	264.6		
Equifax Europe		4.9		0.9		13.4		8.4		
Equifax Latin America		6.0		5.1		13.9		14.7		
Other		_		_		_		4.4		
General Corporate Expense		(11.5)		(12.0)		(42.7)		(35.5)		
			_		_		_			
	\$	88.9*	\$	89.7	\$	254.8	\$	256.6		
	_									

Total assets by segment at September 30, 2003, and December 31, 2002, are as follows:

	Septembe 2003	r 30,	December 31, 2002		
Equifax North America	\$	1,105.1	\$	1,064.8	
Equifax Europe		174.6		174.4	
Equifax Latin America		196.1		161.8	
Other		3.5		3.5	
Corporate		113.2		94.9	
Discontinued Operations		0.9		7.5	
	\$	1,593.4	\$	1,506.9	

Does not total due to rounding

13

12. RECENT ACCOUNTING PRONOUNCEMENTS

In November 2002, the Emerging Issues Task Force ("EITF") reached a consensus on Issue No. 00-21, "Revenue Arrangements with Multiple Deliverables." EITF Issue No. 00-21 provides guidance on how to account for arrangements that involve the delivery or performance of multiple products, services and/or rights to use assets. The provisions of EITF Issue No. 00-21 apply to revenue arrangements entered into in fiscal periods beginning after June 15, 2003. EITF Issue No. 00-21 did not have a material impact on our financial position or on our results from operations for the quarter ended September 30, 2003. We will continue to evaluate EITF Issue No. 00-21 and do not expect that it will have a material impact on our financial position or results of operations.

In December 2002, the FASB issued SFAS No. 148, "Accounting for Stock-Based Compensation, Transition and Disclosure." SFAS No. 148 provides alternative methods of transition for a voluntary change to the fair value based method of accounting for stock-based employee compensation. In addition, SFAS No. 148 amends the disclosure provisions of SFAS No. 123 "Accounting for Stock Based Compensation" to currently require disclosure in the summary of significant accounting policies of the effects of an entity's accounting policy with respect to stock-based employee compensation on reported net income and earnings per share in annual and interim financial statements. SFAS No. 124 to require companies to account for their employee stock-based awards using the fair value method. However, the disclosure provisions are required for all companies with stock-based employee compensation, regardless of whether they utilize the fair value method of accounting described in SFAS No. 123 or the intrinsic value method described in APB Opinion No. 25, "Accounting for Stock Issued to Employees." We adopted SFAS No. 148 on January 1, 2003 and have included the required disclosures in the Notes to the Consolidated Financial Statements.

In January 2003, the FASB issued FASB Interpretation No. 46 ("FIN 46"), "Consolidation of Variable Interest Entities, an Interpretation of ARB No. 51." FIN 46 requires certain variable interest entities to be consolidated by the primary beneficiary of the entity if the equity investors in the entity do not have the characteristics of a controlling financial interest or do not have sufficient equity at risk for the entity to finance its activities without additional subordinated financial support from other parties. FIN 46 is

effective for all new variable interest entities created or acquired after January 31, 2003. For variable interest entities created or acquired prior to February 1, 2003, the provisions of FIN 46 must be applied for the first interim or annual period beginning after June 15, 2003. On October 8, 2003, the FASB deferred the effective date of FIN 46, such that we will be required to apply the provisions beginning with the 4th quarter of 2003. There has been no impact on our financial position or results of operations for the first nine months of 2003 under FIN 46. We will continue the process of evaluating FIN 46, and its prospective impact on our financial position and results of operations, for variable interest entities created or acquired prior to February 1, 2003.

In April 2003, the FASB issued SFAS No. 149 "Amendment of Statement 133 on Derivative Instruments and Hedging Activities". SFAS 149 requires that contracts with comparable characteristics be accounted for similarly. In particular, SFAS 149 clarifies under what circumstances a contract with an initial net investment meets the characteristic of a derivative, clarifies when a derivative contains a financing component, amends the definition of an underlying to conform it to language used in FIN 45, and amends certain other existing pronouncements. SFAS 149 is effective for contracts entered into or modified after June 30, 2003, and for hedging relationships designated after June 30, 2003. In addition, provisions of SFAS 149 should be applied prospectively. We do not expect the application of SFAS 149 to have a material impact on our financial position or results of operations.

In May 2003, the FASB issued SFAS No. 150 "Accounting for Certain Financial Instruments with Characteristics of both Liabilities and Equity". SFAS 150 establishes standards for how an issuer classifies and measures certain financial instruments with characteristics of both liabilities and equity. It requires that an issuer classify a financial instrument that is within its scope as a liability (or an asset in

14

some circumstances). Many of those instruments were previously classified as equity. SFAS 150 is effective for financial instruments entered into or modified after May 31, 2003, and otherwise is effective at the beginning of the first interim period beginning after June 15, 2003. It is to be implemented by reporting the cumulative effect of a change in an accounting principle for financial instruments created before the issuance date of SFAS 150 and still existing at the beginning of the interim period of adoption. Restatement is not permitted. We do not expect the application of SFAS 150 to have a material impact on our financial position or results of operations.

13. DERIVATIVE INSTRUMENTS AND HEDGING ACTIVITIES

Effective January 1, 2001, we adopted SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities." SFAS No. 133 requires that a company recognize derivatives as assets or liabilities on its balance sheet, and also requires that the gain or loss related to the effective portion of derivatives designated as cash flow hedges be recorded as a component of other comprehensive income.

We enter into hedging transactions in order to reduce financial volatility and manage the mix of fixed versus floating interest rate obligations of our debt portfolio. As of September 30, 2003, the only hedging transactions to which we were a counterparty consisted of interest rate swap agreements.

At September 30, 2003, we have a \$29.0 million notional amount floating-to-fixed interest rate swap agreement in place with a bank counterparty that fixes the interest rate on the \$29.0 million synthetic lease related to our corporate headquarters through its maturity in 2010. This hedge has been designated as a cash flow hedge under SFAS 133, is fully effective, and at September 30, 2003, was valued as a liability totaling \$4.5 million. This liability is included with other current liabilities in the accompanying consolidated balance sheets, and the related loss was recorded, net of income tax, as a component of accumulated other comprehensive loss.

At September 30, 2003, we also have interest rate swap agreements in place with bank counterparties to float the interest rate on \$250.0 million of our fixed rate senior unsecured notes through their maturity date in 2005. These derivatives have been designated as fair value hedges and are fully effective. The value of these swaps was \$15.3 million at September 30, 2003, and was recorded as an asset along with a corresponding increase in long-term debt.

Our maximum exposure to loss due to credit risk on these interest rate swap agreements approximates \$10.8 million if all of the bank counterparties default. We mitigate this exposure by monitoring the concentration of risk exposure that we have with any one bank, and through the use of minimum credit quality standards for all counterparties.

15

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS.

The following discussion of our financial condition and results of operations should be read in conjunction with our annual report on Form 10-K for the fiscal year ended December 31, 2002 and our unaudited consolidated financial statements and the related notes in Part I, Item 1—"Financial Statements," above.

Forward-Looking Statements

The following discussion contains forward-looking statements within the "safe harbor" provisions of the Private Securities Litigation Reform Act of 1995, Section 27A of the Securities Act of 1933, and Section 21E of the Securities Exchange Act of 1934. Words such as "may," "could," "should," "would," "believe," "expect," "anticipate," "estimate," "intend," "seeks," "plan," "project," "continue," "predict," and other words or expressions of similar meaning are intended to identify forward-looking statements, although not all forward-looking statements contain these identifying words. These statements are based on our current expectations about future events or results and information that is currently available to us, involve assumptions, risks and uncertainties, and speak only as of the date on which such statements are made. Our actual results may differ materially from the results discussed in such forward-looking statements. Important factors that could cause actual results to differ materially from our forward-looking statements include, but are not limited to, those discussed under the heading "Government Regulation and Pending Legislation" in this Item 2, those discussed under the heading "Forward Looking Statements" and in Part II, Item 7, "Management's Discussion and Analysis of Financial Condition and Results of Operations—Risk Factors," in our 2002 annual report on Form 10-K, and those set forth in our other filings with the SEC and our press releases. We caution that the risks and factors discussed in our 2002 annual report on Form 10-K and in such other filings and press releases are not exclusive. We disclaim any intention or obligation to update or revise any forward-looking statements, whether as a result of new information, future events, or otherwise.

Product Line Overview

We collect, organize, and manage various types of financial, demographic, and marketing information. Our products and services enable businesses to make credit and service decisions, manage their portfolio risk, and develop marketing strategies concerning consumers and commercial enterprises. We serve customers across a wide range of industries, including the financial services, mortgage, retail, telecommunications, utilities, automotive, brokerage, healthcare, and insurance industries, as well as state and federal governments. We also enable consumers to manage and protect their financial health through a portfolio of products offered directly to individuals. We have approximately 4,500 employees worldwide, and manage our business globally through the following three operating segments: Equifax North America, Equifax Europe, and Equifax Latin America. Our operations are predominantly located within the United States, with foreign operations principally located in Canada, the United Kingdom, and Brazil. Our products and services are further categorized into the following three product lines: Information Services, Marketing Services, and Consumer Direct.

In 2002, we redefined our Information Services product line, established Marketing Services as a separate product line to accommodate and reflect the strategic expansion of our

direct marketing products and services resulting from organic growth and acquisitions.

Information Services

Information Services now includes our Consumer Services, Commercial Services, and Mortgage Services, which we had historically referred to collectively as our "credit information services."

16

Consumer Services are products and services derived from the credit information that we maintain regarding individual consumers. These products and services include credit reporting and scores, identity verification, risk management, fraud detection, modeling services, and certain of our decisioning products, which are systems that automate a variety of decision making processes. Customers utilize the information we provide to make decisions for a wide range of credit and business purposes, such as whether, and on what terms, to approve loans, credit card applications, identity verification, and similar business uses.

Commercial Services are products and services derived from our database of information about businesses. These products and services include credit reporting, credit scoring, and risk management and are provided to a wide variety of customers primarily in Canada, the United Kingdom, and Brazil. We are continuing to develop our limited Commercial Services offerings in the United States, recently introducing our Small Business suite of products, which are based on a database which now contains comprehensive credit information on nearly 20 million small businesses. These products include a risk rating tool for account origination and portfolio management and our Small Business Credit Report. These solutions are designed to help our customers make informed decisions about their customers and prospects.

Mortgage Services provide specialized credit reports to mortgage lenders in the United States. These products combine the reports of the three major credit reporting agencies into one comprehensive report.

Information Services also includes our Fraud, Safety & Security product offerings, providing businesses and governments with solutions to help acquire, manage, and distribute information about high-risk individuals and entities, aid in the prevention of terrorist activity across the globe, help our customers meet regulatory compliance requirements, and uncover money-laundering risks.

Marketing Services

Marketing Services include our Credit Marketing and Direct Marketing Services.

Credit Marketing Services are derived from our consumer credit information databases, and include prescreen and account review services. Direct Marketing Services are comprised of the postal and email marketing ("eMarketing") services that we provide in the U.S., and include the products and services that were formerly referred to as our Consumer Information Services. Customers use our compiled, self-reported, permission-based consumer marketing databases and services to solicit consumers via postal and email communications. Our targeted high quality demographic and lifestyle information lists and list performance services, including data enhancement, list hygiene (which verifies a customer's information against other data), modeling and analytical consulting, are designed to facilitate improved mail and email response, and increased customer loyalty. These products also enable customers to target specifically defined market segments, individuals, and businesses, and design more efficient and cost effective marketing campaigns.

Consumer Direct

Consumer Direct includes individual credit information and other financial products and services offered directly to consumers in the United States primarily through a secure Internet channel. Consumer Direct's products and services provide information to consumers which enable them to reduce their exposure to identity fraud and to monitor their credit health. In 2002, we began offering Consumer Direct products to Canadian consumers. During the quarter just ended, we introduced Consumer Direct products in the UK. Revenues from these products are included in the revenues for our Europe business segment.

17

Product Lines and Segments

Information Services products are provided in varying degrees through each of our operating segments. Marketing Services and Consumer Direct products are provided through our Equifax North America and Europe segments. When we refer to our "U.S. Consumer and Commercial Services" from time to time, we mean the Consumer and Commercial Services that we provide in the United States, which we previously referred to as "U.S. Credit Information Services." Similarly, when we refer to our "Canadian Operations," we mean the Consumer Services, Commercial Services, Credit Marketing, and Consumer Direct services that we provide in Canada. References to "North American Information Services" include U.S. Consumer and Commercial Services, Mortgage Services, and Canadian Operations.

Databases

We develop, maintain, and enhance secured proprietary information databases through compilation of accounts receivable information about consumers and businesses that we obtain from a variety of sources, such as credit granting institutions, public record information, including bankruptcies, liens, and judgments, and marketing information from surveys and warranty cards. We process this information utilizing our proprietary information management systems and make it available to our customers in any medium or format they choose.

Summary of Selected Recent Events

Acquisitions. During the first nine months of 2003, we acquired the credit files, contractual rights to territories, customer relationships and related businesses of four independent credit reporting agencies in the United States and one in Canada and a small email marketing business, for \$42.9 million, primarily in cash. See Note 3 to the Consolidated Financial Statements.

IBM Agreement—In August 2003 we entered into an agreement with International Business Machines Corporation (IBM) to provide on demand technology services for our operations in the United States, Canada, and Europe (the "Agreement"). The Agreement renews and expands an existing contract pursuant to which IBM provided to us mainframe and network operations services, which was to expire in 2008. The 10-year Agreement provides \$400 million in additional estimated revenue to IBM for a total contract value of \$800 million. We expect to realize significant cost savings from the Agreement.

Government Regulation and Pending Legislation

A significant portion of our business involves the collection of consumer credit information and the distribution of such information to businesses which make credit, service,

and marketing decisions. As a result, certain of our activities and products and services are subject to regulation under various local, state, and federal laws in the United States, including the Fair Credit Reporting Act, or "FCRA", which regulates the use of consumer credit information. The United States Senate and the House of Representatives have passed separate bills amending the FCRA, which are now before a joint conference committee. Action by both houses on a final version of the legislation is expected by the end of November. Both the House and Senate bills would make permanent the FCRA's preemption of state laws regulating certain aspects of consumer reporting that would otherwise expire at the end of this year. The preservation of national uniform standards is important to us, and that aspect of the proposed legislation is a favorable development. However, provisions of both bills also would impose on us new requirements to provide credit scores and free credit reports annually to consumers on request. Details of the credit report and score requirements and other provisions of the proposed legislation, including provisions that would mitigate potential adverse effects of the legislation, differ in important respects between the two bills. These mitigation provisions include terms and procedures that avoid peaks in demand for free reports, specify methods of requesting the free reports that permit efficient handling, give the ability to charge reasonable fees for credit scores, provide twelve months or more to prepare for implementation, preempt further state free report laws and expand the periods

18

within which such reports must be delivered and information reverified. This legislation could result in additional expense that could have a material adverse effect on our financial condition and results of operations. We cannot predict the net impact of the proposed legislation on our business, however, until we know its final terms and have the opportunity to evaluate the effects of countervailing factors, including the nature of mitigation provisions included in the final legislation, internal steps to lessen its impact, and opportunities to market fee-generating products to consumers requesting free reports.

Components of Income Statement

Revenues from our three operating segments, Equifax North America, Equifax Europe and Equifax Latin America, are generated from a variety of products and services categorized into three groups: Information Services, Marketing Services, and Consumer Direct. In the third quarter of 2003, our Equifax North America segment generated 83% of our worldwide revenues and 89% of our operating profit before corporate expenses. For the nine months ended September 30, 2003, our North America segment generated 84% of our worldwide revenues and 91% of our operating profit before corporate expenses. We provide information regarding our non-US based operations in local currency to demonstrate the economic effect of such operations without the impact of currency translation. We believe this is useful to one's understanding of the business.

Information Services revenues are principally transaction related, and are derived from our sales of the following products (virtually all delivered electronically): credit reporting and scoring, mortgage reporting, identity verification, fraud detection, decisioning, and modeling services. Revenues from Marketing Services are derived from our sales of products that help customers acquire new customers, and manage existing customers, and include our Credit Marketing and Direct Marketing Services. Consumer Direct revenues are transaction related, and are derived from our sales of credit reporting products and identity theft monitoring services, which we deliver to consumers electronically via the Internet and via mail. Our revenues are sensitive to a variety of factors, such as demand for, and price of, our services, technological competitiveness, our reputation for providing timely and reliable service, competition within our industry, federal, state, and foreign regulatory requirements governing privacy and use of data, and general economic conditions. Our operating expenses include costs of services and selling, general, and administrative, or SG&A, expense. Costs of services consist primarily of data acquisition and royalties; customer service costs, which include personnel costs to collect, maintain and update our proprietary databases, to develop and maintain software application platforms, and to provide consumer and customer call center support; hardware and software expense associated with transaction processing systems; telecommunication and computer network expense; and occupancy costs associated with facilities where these functions are performed. SG&A expenses consist primarily of personnel costs for compensation paid to sales and administrative employees and management. Depreciation and amortization expense includes amortization of acquired intangible assets.

Consolidated Results of Operations

Our consolidated results for the three and nine month periods ended September 30, 2003 and 2002 are as follows:

	Third Quarter				Nine Months			
	2003		2002		2 2003			2002
(In millions, except per share data)	_		_		_			
Revenue	\$	309.8	\$	289.7	\$	928.4	\$	816.7
Operating Income	\$	88.9	\$	89.7	\$	254.8	\$	256.6
Income from Continuing Operations	\$	52.8	\$	49.7	\$	147.2	\$	139.6
Net Income	\$	51.2	\$	38.9	\$	136.9	\$	128.1
Diluted EPS:								
Income from Continuing Operations	\$	0.39	\$	0.36	\$	1.08	\$	1.01
Net Income	\$	0.38	\$	0.28	\$	1.00	\$	0.92

19

All references to earnings per share data in this MD&A are to diluted earnings per share unless otherwise noted.

Highlights for Third Quarter 2003 Consolidated Financial Results

- Consolidated revenues of \$309.8 million increased 7% over the same period a year ago.
- Income from continuing operations was \$52.8 million compared to \$49.7 in the prior year.
- Earnings per share from continuing operations increased 8%.
- Total debt outstanding at September 30, 2003 was \$868.6 million, with an average interest rate of 3.9% compared to \$895.0 million with an average interest rate of 4.9% in the prior year.
- Cash provided by operations was \$91.4 million up \$33.6 million from the same period a year ago.
- Repurchase of 1.5 million shares of common stock for a total investment of \$35.5 million.

Third Quarter 2003 Compared With Third Quarter 2002.

Our consolidated operating revenues of \$309.8 million for the three months ended September 30, 2003 grew by \$20.1 million, or 7%, over the same period one year ago. This

growth was provided primarily by our Equifax North America segment and was driven by volume increases for Information Services and Consumer Direct more than offsetting lower volume for Marketing Services. Information Services growth was fueled by 10% growth in our US consumer and commercial services business and by 14% growth in our Mortgage Services due to slightly higher refinancing volumes when comparing the same quarter this year to last year. Marketing Services was off 11% when compared to the same quarter a year ago due mainly to volume decreases. Consumer Direct revenues grew 69% over the same period one year ago due to continued growth in consumer subscriptions and number of transactions. Equifax Latin America grew 11% in local currency despite difficulties in the economic environment. Our revenues for Equifax Latin America grew by 17% compared to the prior period including a favorable foreign currency impact of \$1.2 million. Our European segment reported third quarter revenues of \$1.7 million, which is \$1.1 million less than the third quarter of 2002. Foreign exchange rates provided a benefit of \$1.8 million to Europe's operating revenues.

Consolidated operating expenses for the third quarter of 2003 of \$220.9 million grew by \$20.9 million, or 10%, over the same period in 2002. This increase is primarily the result of volume increases in correlation to revenue growth, expenditures related to the integration of our eMarketing business into our Direct Marketing Services product line, operations of our eMarketing business and new growth initiatives. Volume growth in our Consumer Services, Mortgage Services, and Consumer Direct businesses accounted for \$11.9 million or 57% of our expense growth. Expenses associated with our eMarketing business accounted for \$6.5 million or 31% of our total expense growth. eMarketing operating expenses and business integration costs outpaced reduced eMarketing revenue. If eMarketing personnel reductions (109 full time equivalents in Q3 of 2003) and operating efficiencies, that have been initiated in an attempt to bring eMarketing operating costs in line with eMarketing revenue, are not enough to produce acceptable results, management may have to take additional action, including greater personnel reductions and facility consolidations. Growth initiatives for which current expenditures will create future revenue streams are our Fraud, Safety and Security and our Small Business Enterprise businesses. These businesses were the predominate source of remaining expense growth for the quarter. Compared to the third quarter of 2002, we reduced consolidated headcount by 400 full time equivalents across business units in a continuing drive to generate incremental operating efficiencies on a consolidated basis.

Costs of services increased by \$15.8 million or 14% compared to the same quarter of 2002. Costs of services represent 42% of total operating revenues for the quarter, an increase from 39% for the same

20

quarter last year. Volume increases in our U.S. Consumer and Commercial Services, Mortgage Services, and Consumer Direct businesses continue to drive increased expenditures in royalty expenses, data costs, and production expenses.

SG&A grew by \$0.5 million or 1% over the same quarter a year ago. As a percent of total operating revenues, SG&A decreased to 22% for the third quarter of 2003 from 23% from the same quarter in 2002. The increase in dollars is due primarily to growth in our business.

Operating income for the third quarter of 2003 decreased by \$0.8 million or 1% as investment in growth initiatives, growth in our lower margin Marketing Services and Consumer Direct businesses, expenses related to the integration of our eMarketing business into our Direct Marketing Services, operations of our eMarketing business and volume related increases in costs of sales combined to produce operating margins of 29% for the quarter compared to 31% for the same period of 2002. Our business mix is shifting as we continue to diversify our marketing and consumer businesses to drive incremental revenues and profits.

Other income, net

Other income, net of \$3.0 million for the third quarter of 2003, is comprised primarily of income from our share of collections amounting to \$2.7 million related to the sale of our risk management collections business in 2000 and \$0.5 million in interest income. This compares favorably to other income, net of \$2.0 million for the third quarter of 2002, which is comprised primarily of interest income.

Interest expense

Interest expense decreased \$1.8 million in the third quarter 2003 compared to the third quarter 2002. These decreases were due to lower interest rates. Total debt outstanding of \$868.6 million as of September 30, 2003 is \$26.4 million lower than the balance of \$895.0 million at September 30, 2002.

Effective Tax Rates

Our effective tax rates from continuing operations were 36% and 39% for the third quarters ended 2003 and 2002, respectively. The decrease in the effective rate over time is due to the implementation of state tax planning strategies related to the apportionment of state income taxes and franchise taxes.

2

Segment Results

Our segment results for the third quarter 2003 and 2002 are as follows:

		Third (Juart	Variance		
	_	2003		2002	\$	%
(In millions)	_					
Operating Revenue:						
Equifax North America	\$	256.6	\$	238.6	\$ 18.0	8%
Equifax Europe		31.7		32.8	(1.1)	(3)%
Equifax Latin America		21.5		18.3	3.2	17%
Other		_		_	_	0%
	_					
	\$	309.8	\$	289.7	\$ 20.1	7%
	_					
Operating Income:						
Equifax North America	\$	89.4	\$	95.7	\$ (6.3)	(7)%
Equifax Europe		4.9		0.9	4.1	487%
Equifax Latin America		6.0		5.1	0.9	18%
Other		_		_	_	0%
General Corporate Expense		(11.5)		(12.0)	0.5	4%
	_					

Third Oneston

Vanionas

Does not total due to rounding

Equifax North America

In the third quarter of 2003, Equifax North America generated 83% of our revenue and 89% of our operating profit before corporate expense. This segment's revenue increased 8% compared to the same period last year.

As shown in the following table, our Equifax North America segment includes revenues from our:

- U.S. Consumer and Commercial Services, which are comprised of the Consumer and Commercial Services that we provide in the U.S., which we previously referred to as U.S. Credit Information Services;
- Mortgage Services, which we previously included in U.S. Credit Information Services;
- Canadian Operations, which are comprised of the Consumer Services, Commercial Services and Credit Marketing Services that we provide in Canada;
- Credit Marketing Services provided in the U.S.;
- Direct Marketing Services which are comprised of the postal and email marketing services provided in the U.S. and include the products and services that we
 formerly referred to as our Consumer Information Services;
- Consumer Direct products and services that we provide in the United States, Canada, and the UK.

22

		Third Quarter					Variance 9/2		
		2003 2002		2002	<u> </u>		%		
(In millions)				_		_			
Equifax North America Revenue:									
U.S. Consumer and Commercial Services		\$	131.9	\$	119.9	\$	12.0	10%	
Mortgage Services			18.3		16.0		2.3	14%	
Canadian Operations			23.3		19.4		4.0	21%	
•				_					
Total North America Information Services	:	\$	173.5	\$	155.3	\$	18.3	12%	
Credit Marketing Services			38.2		42.2		(4.0)	(9)%	
Direct Marketing Services			26.2		30.1		(3.9)	(13)%	
				_		_			
Total Marketing Services			64.4		72.3		(7.9)	(11)%	
Č	,			_		_			
Consumer Direct			18.7		11.0		7.7	70%	
Consumer Direct			10.7		11.0		7.7	7070	
		\$	256.6	\$	238.6	\$	18.0	8%	

Third Quarter 2003 Compared With Third Quarter 2002.

U.S. Consumer and Commercial Services revenues for the third quarter 2003 grew by \$12.0 million, or 10%, over the same quarter last year. The increase was largely due to a combination of new accounts and increased volume in existing accounts. Mortgage refinancing slowed down in the third quarter of 2003 as mortgage interest rates came off of their 40 year low. As a result, Mortgage Services revenue growth slowed to 14% when compared to the same three months last year. Our business mix shifted in the third quarter of 2003 toward increased volumes in our marketing and consumer related businesses which have a lower price per unit of online activity. Canadian Operations revenues increased by \$4.0 million or 21% due to increased volumes in consumer line and decision solutions businesses.

Credit Marketing Services revenues decreased \$4.0 million, or 9%, compared to the third quarter of last year. The same quarter comparison reflects price compression exacerbated by net volume decreases. Direct Marketing Services revenues decreased by \$3.9 million or 13%, due mainly to the deterioration in the performance of our eMarketing business. As discussed above, we have reduced headcount in our eMarketing business by 109 full time equivalents compared to the same quarter last year. Our marketing services business does not involve the sending of unsolicited commercial email (spam). However, increased business and legislative efforts to reduce and or regulate the use of spam continues to negatively affect marketing services in general and to adversely impact sales of our consumer requested email marketing (opt-in) services. Numerous Federal and State legislative proposals would regulate the use of spam and we believe such legislation would result in greater market acceptance of opt-in email marketing. In addition, the establishment of the national telemarketing do-not-call registry and efforts by marketers to utilize non-telephonic means of marketing may benefit Direct Marketing Services through increased demand for its non-telephonic products and services. However, the national do-not-call list registry is currently being challenged in court. We cannot predict the terms of any anti-spam legislation that might be enacted or the results of the challenge to the national do-not-call list or their possible impact on our business.

Consumer Direct revenues grew \$7.6 million or 69% compared to the same quarter last year due to new subscriptions and an increase in renewals of our Credit Watch product.

Equifax North America operating profit for the third quarter 2003 was \$89.4 million, a decrease of \$6.3 million, or 7%, over the third quarter of the previous year. Because of increased expenses resulting from investment in future growth initiatives, such as our Fraud, Safety and Security and Small Business Enterprise businesses, and the ongoing expenses from the integration of eMarketing services into Direct Marketing Services, Equifax North America produced margins of 35% compared to 40% one year ago. See our discussion above regarding expenses related to volume growth, eMarketing integration, eMarketing operations and investment in future growth initiatives that also apply here. We expect operating margins will improve as our new initiatives continue to grow and margins in these new initiatives improve with maturity.

Equifax Europe

Third Quarter 2003 Compared With Third Quarter 2002.

Equifax Europe includes the results of our operations in the United Kingdom, Spain, Portugal, and Italy, and our support operations in Ireland. Our commercial operations in Spain for the third quarter 2003 and 2002 are classified as discontinued operations, while our consumer operations in Spain are included in this segment as continuing operations. Our revenue growth for the segment declined 9% in local currency but was only down 3% from the same quarter last year after foreign currency translation.

For the combined Equifax Europe segment, operating margins increased from 3% in the third quarter of 2002 to 15% in the third quarter of 2003, due to extensive cost reductions driven by lower headcount. Relatively flat revenue in the United Kingdom, Spain and Portugal offset margin erosion in Italy. Equifax Europe reported operating income of \$4.9 million for the third quarter of 2003 and \$0.8 million for the third quarter of 2002.

Equifax Latin America

Third Quarter 2003 Compared With Third Quarter 2002.

Equifax Latin America includes results of our operations in Brazil, Argentina, Chile, Peru, Uruguay, and El Salvador. In local currency, combined operating revenues for this segment improved by 11% over third quarter of the previous year, and were up by 17% after foreign exchange translation.

Operating income for the Equifax Latin America segment increased from \$5.1 million for the third quarter of 2002 to \$6.0 million for the third quarter of 2003. Operating margins were 28% for both periods.

Other

In our Other segment, we report information about the former lottery business, which consists solely of an agreement between a subsidiary of ours and GTECH Corporation. Pursuant to this subcontract, GTECH assumed obligations of our subsidiary under a contract with the State of California to install a system to automate the processing of instant lottery tickets, provide terminals and related security hardware, and license various software applications developed to support the system. We have exited the lottery business, and all previously deferred revenue related to this subcontract has now been recognized, and no further revenue or operating income is expected to occur in this segment.

General Corporate

General corporate expense decreased by \$0.5 million, or 4%, to \$11.5 million in the third quarter 2003 compared to the same period a year ago. The large scale productivity project that had generated significant consulting fees in the first two quarters of 2003 has been completed and resulted in a new agreement with IBM to provide global on demand technology services. See Summary of Selected Recent Events on page 18.

First Nine Months 2003 Compared With First Nine Months 2002.

Our operating revenues of \$928.4 million for the nine months ended September 30, 2003 grew by \$111.7 million, or 14%, over the same period one year ago. This growth was provided primarily by our Equifax North America segment and was driven by volume increases across all product lines: Information Services, Marketing Services and Consumer Direct. Information Services growth was fueled by 16% growth in our US consumer and commercial services business and by 45% growth in our Mortgage Services due to stronger refinancing volumes in the first nine months of 2003 compared to the same period in 2002. Marketing Services grew 9% overall. Our eMarketing business provided

24

significant revenue growth for Direct Marketing Services during the first nine months of 2003 compared to the first nine months of 2002. Consumer Direct revenues grew 85% over the same period one year ago due to continued growth in consumer subscriptions and transactions. Equifax Latin America grew 9% in local currency despite continuing difficulties in the economic environment. Our revenues for Equifax Latin America declined by 3% compared to the prior period due to an unfavorable foreign currency impact. Equifax Europe reported revenue declines of 11% in local currency, yet only a 1% decline including favorable currency benefit to Europe's operating revenues. Our European segment reported first nine month revenues of \$94.4 million, which is relatively even with first nine months of 2002.

Consolidated operating expenses for the first nine months of 2003 of \$673.6 million grew by \$113.5 million, or 20%, over the same period in 2002. This increase is primarily the result of volume increases in correlation to revenue growth, expenditures primarily related to the integration of our eMarketing business into our Direct Marketing Services product line and new growth initiatives. Expenses associated with our eMarketing business account for \$47.6 million or 42% of our total expense growth. These are a combination of the expenses of a new product line and the costs associated with the headcount reduction and integration efforts discussed above. Growth initiatives for which current expenditures will create future revenue streams are our Fraud, Safety and Security and Small Business Enterprise businesses. Volume growth in our Consumer Services, Mortgage Services and Consumer Direct businesses accounted for \$44.9 million or 40% of our expense growth. During the first nine months of 2003, we have reduced headcount by 379 full time equivalents across business units in a continued drive to provide operating efficiencies.

Costs of services during the first nine months of 2003 increased by \$70.7 million or 22% compared to the first nine months of 2002. Costs of services represent 42% of total operating revenues for the period, an increase from 40% for the same period last year. Volume increases in our U.S. Consumer and Commercial Services, Mortgage Services and Consumer Direct businesses continue to drive increased expenditures in royalty expenses, data costs, and production expenses. In addition, incremental eMarketing costs for the compared periods were exacerbated by the integration expenditures discussed above.

SG&A grew by \$28.4 million or 16% over the same period a year ago. As a percent of total operating revenues, SG&A decreased to 22% for the first nine months of 2003 from 23% from the same period in 2002. The increase in dollars is due primarily to eMarketing's SG&A.

Operating income for the first nine months of 2003 decreased by \$1.8 million or 1% as investment in growth initiatives, growth in our lower margin Marketing Services and Consumer Direct businesses, expenses related to eMarketing and volume related increases in costs of sales combined to produce operating margins of 27% for the period compared to 31% for the same period of 2002. Our business mix continues to shift as we diversify into marketing and consumer businesses for the purpose of driving incremental revenues, profits and cash flow.

Other income, net

Other income, net of \$12.3 for the first nine months of 2003 is comprised primarily of income from our share of collections amounting to \$9.0 million related to the sale of our risk management collections business in 2000, \$1.8 million of exchange gains on foreign currency, and \$1.4 million in interest income. This compares favorably to other income, net of \$5.3 million for the first nine months of 2002 which is comprised primarily of interest income of \$5.7 million and gains on foreign currency of \$1.2 million.

Interest expense

Interest expense increased \$0.6 million for the first nine months compared to 2002. These increases were due to higher average borrowings. Total debt outstanding of \$868.6 million as of September 30, 2003 is \$26.3 million lower than the balance of \$895.0 million at September 30, 2002. The increase in debt from September 30, 2002 to September 30, 2003 provided cash for business acquisitions and for stock repurchases.

Effective Tax Rates

Our effective tax rates from continuing operations were 37% and 39% for the first nine months of 2003 and 2002, respectively. The decrease in the effective rate over time is due to the implementation of state tax planning strategies related to the apportionment of state income taxes and franchise taxes.

Segment Results

Our segment results for the first nine months ended 2003 and 2002 are as follows:

		Nine M	Ionths		;		
		2003		2002	s		%
(In millions)							
Operating Revenue:							
Equifax North America	\$	776.9	\$	657.8	\$	119.1	18%
Equifax Europe		94.4		95.6		(1.2)	(1)%
Equifax Latin America		57.1		58.9		(1.8)	(3)%
Other		_		4.4		(4.4)	(100)%
	\$	928.4	\$	816.7	\$	111.7	14%
	_						
Operating Income:							
Equifax North America	\$	270.2	\$	264.6	\$	5.6	2%
Equifax Europe		13.4		8.4		5.1	61%
Equifax Latin America		13.9		14.7		(0.8)	(6)%
Other		_		4.4		(4.4)	(100)%
General Corporate Expense		(42.7)		(35.5)		(7.2)	(20)%
	_				_		
	\$	254.8	\$	256.6	\$	(1.8)	(1)%

26

Equifax North America

In the first nine months of 2003, Equifax North America generated 84% of our revenue and 91% of our operating profit before corporate expense. This segment's revenue increased 18% compared to the same period last year.

	Nine Months					Variance		
	2	2003		2002	s		%	
(In millions)								
Equifax North America Revenue:								
U.S. Consumer and Commercial Services	\$	391.2	\$	338.6	\$	52.6	16%	
Mortgage Services		56.4		38.9		17.5	45%	
Canadian Operations		66.7	_	58.2	_	8.5	15%	
Total North America Information Services	\$	514.3	\$	435.7	\$	78.6	18%	
Credit Marketing Services		120.2		122.7		(2.5)	(2)%	
Direct Marketing Services		91.4		71.8		19.6	27%	
Total Marketing Services		211.6		194.6		17.0	9%	
Consumer Direct		50.9		27.5		23.4	85%	
	\$	776.9*	\$	657.8	\$	119.1	18%	

^{*} Does not total due to rounding

First Nine Months 2003 Compared With First Nine Months 2002.

U.S. Consumer and Commercial Services revenues for the first nine months of 2003 grew by \$52.6 million, or 16%, over the same period last year. The increase was largely due to a combination of new accounts and increased volume in existing accounts. Additional momentum resulted from mortgage refinancing with Mortgage Services delivering 45% growth as mortgage interest rates reached 40 year lows. Overall pricing remained stable, e.g. the average price decrease per online unit was only 5% through the first nine

months of 2003. Canadian Operations revenues increased by \$8.5 million or 15% due to increased volumes in the first nine months of 2003 compared to the same period in 2002.

Credit Marketing Services revenues marginally increased compared to the first nine months of last year.

Direct Marketing Services revenues increased by \$19.7 million, or 27%, due mainly to the acquisition of our eMarketing business in the third quarter of 2002. Our marketing services business does not involve the sending of unsolicited commercial email (spam). However, increased business and legislative efforts to reduce and or regulate the use of spam continues to negatively affect marketing services in general and to adversely impact sales of our consumer requested email marketing (opt-in) services. Numerous Federal and State legislative proposals would regulate the use of spam and we believe such legislation would result in greater market acceptance of opt-in email marketing. In addition, the establishment of the national telemarketing do-not-call registry and efforts by marketers to utilize non-telephonic means of marketing may benefit Direct Marketing Services through increased demand for its non-telephonic products and services. However, the national do-not-call list registry is currently being challenged in court. We cannot predict the terms of any anti-spam legislation that might be enacted or the results of the challenge to the national do-not-call list or their possible impact on our business.

Consumer Direct revenues grew \$23.4 million or 85% compared to the same period last year due to new subscriptions and an increase in renewals of our Credit Watch product.

Equifax North America grew operating profit to \$270.2 million, an increase of \$5.6 million, or 2%, over the first nine months of the previous year based on strong revenue growth across all businesses. While investing in future growth initiatives, such as our Fraud, Safety and Security and Small Business

27

Enterprise businesses, and absorbing business integration costs, Equifax North America produced margins of 35% compared to 40% one year ago. We anticipate that operating margins will continue to shift while we continue to invest in new electronic marketing, safety & security, and consumer initiatives, and also expect that margins in these new initiatives will improve with maturity.

Equifax Europe

First Nine Months 2003 Compared With First Nine Months 2002.

Equifax Europe includes the results of our operations in the United Kingdom, Spain, Portugal, and Italy, and our support operations in Ireland. Our commercial operations in Spain for the first nine months of 2003 and 2002 are classified as discontinued operations, while our consumer operations in Spain are included in this segment as continuing operations. Revenue growth for the segment declined 9% in local currency but was down only 3% with the same period last year after foreign currency translation.

For the combined Equifax Europe segment, operating profit improved 60% compared to the first nine months of 2002. This profit improvement was due to extensive cost reductions driven by lower headcount. Modest growth in the United Kingdom, Spain and Portugal mitigated margin erosion in Italy. Equifax Europe reported operating income of \$13.4 million for the first nine months of 2003 and \$8.4 million for the first nine months of 2002.

Equifax Latin America

First Nine Months 2003 Compared With First Nine Months 2002.

Equifax Latin America includes results of our operations in Brazil, Argentina, Chile, Peru, Uruguay, and El Salvador. In local currency, combined operating revenues for this segment improved by 9% over the first nine months of the previous year, but were lower by 3% after foreign exchange translation.

Operating income for the Equifax Latin America segment declined from \$14.7 million for the first nine months of 2002 to \$13.9 million for the first nine months of 2003 as did operating margins which declined from 25% to 24% over the same period.

Other

In our Other segment, we report information about our former lottery business, which consists solely of an agreement between a subsidiary of ours and GTECH Corporation. Pursuant to this subcontract, GTECH assumed obligations of our subsidiary under a contract with the State of California to install a system to automate the processing of instant lottery tickets, provide terminals and related security hardware, and license various software applications developed to support the system. We have exited the lottery business, and all previously deferred revenue related to this subcontract has now been recognized, and no further revenue or operating income is expected to occur in this segment. We reported revenues and operating income of \$4.4 million for the first six months of 2002 for this segment.

General Corporate

General corporate expense increased by \$7.2 million or 20% to \$42.7 million in the first nine months of 2003 compared to the same period a year ago. We had greater legal expenses during the first nine months of 2003 due to our rapid business volume expansion and integration and cost optimization related expenditures of acquisitions. In addition, we have incurred consulting fees in conjunction with the signing of our new agreement with IBM, which will drive future expense efficiencies in the maintenance and operation of our technology resources. See Summary of Selected Recent Events on page 18.

28

Liquidity and Capital Resources

Overview

Our principal sources of liquidity are cash flows provided by our operating activities, our revolving credit facilities, and cash and cash equivalents on hand.

Our ability to generate cash from our operations is one of our fundamental financial strengths. We use cash flows from operations, along with borrowings, to fund growth initiatives, to fund capital expenditures, to make acquisitions, to retire outstanding indebtedness, to pay dividends, and to make share repurchases.

Cash from Operations

For the nine months ended September 30, 2003, we generated \$193.5 million of cash flow from operating activities compared to \$150.0 million for the nine months ended September 30, 2002. The major source of cash flow for the first nine months of 2003 was net income of \$136.9 million, net of loss from discontinued operations in Spain of \$9.8 million and \$72.5 million for depreciation and amortization. Total working capital at September 30, 2003, was \$72.0 million versus a negative \$142.3 million at

December 31, 2002. Significant changes in working capital for the first nine months of 2003 included an increase in trade receivables, a decrease in current maturities and a slight increase in other current liabilities. The increase in trade receivables is due to increased sales. Current maturities decreased significantly as our matured notes were paid in June 2003. The change in other current liabilities is due to payments for restructuring reserves that were booked at the end of 2001 being made primarily in the first nine months of 2002.

Investing Activities

Investing activities for the first nine months of 2003 and 2002 used cash of \$78.8 million and \$217.5 million, respectively. Capital expenditures used cash in the amounts of \$36.7 million and \$35.4 million for the first nine months of 2003 and 2002, respectively. Our capital expenditures are used for developing, enhancing, and deploying new and existing technology platforms, replacing or adding equipment, updating systems for regulatory compliance, the licensing of software applications and investing in disaster recovery systems.

In addition, we used cash of \$40.7 million and \$218.1 million for the first nine months of 2003 and 2002, respectively, for acquisitions. We acquired consumer credit files, contractual rights to territories, and customer relationships and related businesses from four Affiliates in the United States and one in Canada and a small email marketing business all for \$42.9 million primarily in cash for the first nine months of 2003 and for seven Affiliates in the United States and three Affiliates in Canada, the remaining 20% interest of our information services company in Brazil, a small technology development company and an email marketing business, primarily in cash, in 2002.

Financing Activities

Cash used by financing activities was \$115.5 million for the first nine months of 2003 contrasted to \$67.6 million provided for in the first nine months of 2002. Net short-term payments were \$11.5 and \$15.2 million for the first nine months of 2003 and 2002, respectively. Additions to our long-term debt were \$155.3 million and \$137.9 million for the first nine months of 2003 and 2002, respectively. Payments to our long-term debt were \$202.5 million for the first nine months of 2003.

We used cash of \$65.3 million during the first nine months of 2003 for the purchase of 2,938,800 of our common shares at an average price of \$22.22. Share purchases during the same period in 2002 were \$78.7 million.

29

Our dividend policy has remained consistent, we used cash of \$8.4 million for the first nine months of 2003 and \$8.6 million for the first nine months of 2002. We received cash of \$16.7 million and \$31.6 million for the same periods, respectively, due to the exercise of stock options by employees.

We increased the amount outstanding under our \$465.0 million credit facility in 2003 for purposes of retiring the \$200.0 million aggregate principal amount of our outstanding 6.5% senior notes that matured in June 2003.

Cash and Cash Equivalents

Our cash and cash equivalents balance was \$33.5 million and \$30.5 million at September 30, 2003 and December 31, 2002, respectively.

Revolving Credit Facilities

Our \$465.0 million revolving credit facility, which we entered into with Bank of America, N.A. and certain other lenders on October 4, 2001, provides for a variable interest rate tied to Base Rate, LIBOR and competitive bid options. The weighted average interest rate of borrowings outstanding under this facility was approximately 1.6% as of September 30, 2003. The credit facility consists of a \$160.0 million 364-day portion and a \$305.0 million multi-year portion, which expire on September 30, 2004 and October 3, 2004, respectively. The current amount outstanding, \$178.1 million is all borrowed under the \$305.0 multi-year portion of our facility. The agreement governing this facility contains various covenants and restrictions, including, among other things, limitations on liens, subsidiary debt, mergers, liquidation, asset dispositions, acquisitions, and maintenance of certain financial covenants. Our borrowings under this facility, which have not been guaranteed by any of our subsidiaries, are unsecured and will rank on parity in right of payment with all of our other unsecured and unsubordinated indebtedness from time to time outstanding. As of September 30, 2003 we had \$286.9 million of borrowing capacity available under our \$465.0 million revolving credit facility. Our variable interest facilities increased by \$156.3 million during the first nine months of 2003 due primarily to the funding necessary to pay the \$200.0 million senior note that matured in September 2003.

One of our Canadian subsidiaries has an unsecured, 364-day C\$100.0 million revolving credit facility that will expire on September 30, 2004. The agreement provides for borrowings tied to Prime, Base Rate, LIBOR and Canadian Bankers' Acceptances, and contains financial covenants related to interest coverage, funded debt to cash flow, and limitations on subsidiary indebtedness. We have guaranteed the indebtedness of our Canadian subsidiary under this facility. As of September 30, 2003, \$54.8 million of borrowing capacity was available under this credit facility, from which no amount was outstanding.

Contractual Obligations and Commercial Commitments

We believe that future cash flows provided by our operating activities, together with current cash and cash equivalent balances, will be sufficient to meet our projected cash requirements for the next 12 months, and the foreseeable future thereafter, although any projections of future cash needs and cash flows are subject to substantial uncertainty. For instance, Computer Sciences Corporation and certain of its affiliates (collectively "CSC") has an option, exercisable at any time prior to 2013, to sell its credit reporting business to us. The option exercise price will be determined by a third-party appraisal process and would be due in cash within 180 days after the exercise of the option. We estimate that if CSC were to exercise the option today, the option price would fall within a range of approximately \$650.0 to \$700.0 million. This estimate is based solely on our internal analysis of the value of the business, market conditions at the time of our analysis, and other factors, all of which are subject to constant change. If CSC were to exercise its option, we would have to obtain additional sources of funding. We believe that this funding would be available from sources such as additional bank lines of credit and the issuance of public debt and/or equity. However, the availability and terms

30

of any such financing would be subject to a number of factors, including credit market conditions, the state of the equity markets, general economic conditions, and our financial performance and condition. Because we do not control the timing of CSC's exercise of its option, we could be required to seek such financing and increase our indebtedness at a time when market or other conditions are unfavorable.

We continually evaluate opportunities to sell additional equity or debt securities, obtain credit facilities from lenders, and restructure our long-term debt for strategic reasons, or to further strengthen our financial position. The sale of additional equity or convertible debt securities could result in additional dilution to our shareholders. In addition, we will, from time to time, consider the acquisition of, or investment in, complementary businesses, products, services and technologies, and the repurchase and retirement of debt, which might affect our liquidity requirements or cause us to issue additional equity or debt securities. There can be no assurance that financing will be available in amounts or on terms acceptable to us, if at all.

Off Balance Sheet Transactions

Other than facility leasing arrangements, we do not engage in off-balance sheet financing activities. We have entered into a synthetic lease on our Atlanta corporate headquarters building in order to provide us with favorable financing terms with regard to this facility. This \$29.0 million lease was entered into in 1998 and expires in 2010. Total lease payments for the remaining term total \$12.1 million. Under this synthetic lease arrangement, we have also guaranteed the residual value of the leased property to a lessor. In the event that the property were to be sold by the lessor at the end of the lease term, we would be responsible for any shortfall of the sales proceeds, up to a maximum amount of \$23.2 million, which equals 80 percent of the value of the property at the beginning of the lease term. We believe that the fair market value of this property exceeds the amount of the guarantee.

Letters of credit and guarantees

We will, from time to time, issue standby letters of credit, performance bonds or other guarantees in the normal course of our business. The aggregate notional amount of all performance bonds and standby letters of credit is less than \$15.0 million and they all have a maturity of one year or less. We provide these guarantees from time to support the needs of our operating units. Except for our guarantee of the synthetic lease referred to above, our only outstanding guarantee that is not reflected as a liability on our balance sheet was extended in connection with the sale of our risk management collections business to RMA Holdings, LLC, or RMA, in October 2000, at which time we guaranteed the operating lease payments of a partnership affiliated with RMA to a lender of the partnership pursuant to a term loan. The term loan, which had \$7.4 million outstanding as of September 30, 2003, expires December 1, 2011. Our obligations under the RMA guarantee are not secured. We believe that the likelihood of demand for payment under these instruments is minimal and expect no material losses to occur in connection with these instruments.

Subsidiary funds transfer limitations

The ability of certain of our subsidiaries and associated companies to transfer funds is limited in some cases by foreign government regulations. At September 30, 2003, the amount of equity subject to such restrictions for consolidated subsidiaries was not material.

Inflation

We do not believe that the rate of inflation has had a material effect on our operating results. However, inflation could adversely affect our future operating results if it were to result in a substantial weakening in economic conditions.

31

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK.

For information regarding our exposure to certain market risk, see Item 7A, "Quantitative and Qualitative Disclosures About Market Risk," in our Annual Report on Form 10-K for the year ended December 31, 2002. There have been no material changes to our market risk exposure during the first nine months of 2003.

ITEM 4. CONTROLS AND PROCEDURES.

Our management, with the participation of our principal executive officer and principal financial officer, has evaluated the effectiveness of our disclosure controls and procedures (as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934, as amended (the "Exchange Act")), as of the end of the period covered by this Quarterly Report on Form 10-Q. Based on such evaluation, our principal executive officer and principal financial officer have concluded that as of such date, our disclosure controls and procedures were effective in ensuring that information required to be disclosed by us in reports that we file or submit under the Exchange Act is recorded, processed, summarized and reported, within the time periods specified in applicable SEC's rules and forms. No system of controls, no matter how well designed and operated, can provide absolute assurance that the objectives of the system of controls are met, and no evaluation of controls can provide absolute assurance that all control issues and instances of fraud, if any, within a company have been detected.

32

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

We are involved in various lawsuits, claims and legal proceedings arising in the normal course of our business that are described in our 2002 Annual Report on Form 10-K and our quarterly reports on Form 10-Q for the periods ended March 31, 2003 and June 30, 2003. During the three months ended September 30, 2003, there were no material developments in such lawsuits, claims and legal proceedings.

33

ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K

(a) Exhibits

The following is a complete list of Exhibits included as part of this Report. A list of those documents filed with this Report is set forth on the Index to Exhibits appearing elsewhere in this Report and is incorporated by reference:

Exhibit No.	Description
10.1	Agreement for Operations Support Services ⁽¹⁾
31.1	Certification of Thomas F. Chapman, Chairman and Chief Executive Officer of Equifax Inc., Pursuant to Rule 13a-14(a).
31.2	Certification of Donald T. Heroman, Chief Financial Officer of Equifax Inc., Pursuant to Rule 13a-14(a).

- Certification of Thomas F. Chapman, Chairman and Chief Executive Officer of Equifax Inc., Pursuant to 18 U.S.C. Section 1350, as adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
 Certification of Donald T. Heroman, Chief Financial Officer of Equifax Inc., Pursuant to 18 U.S.C. Section 1350, as adopted Pursuant.
- 32.2 Certification of Donald T. Heroman, Chief Financial Officer of Equifax Inc., Pursuant to 18 U.S.C. Section 1350, as adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
- (1) Document omits information pursuant to a Request for Confidential Treatment under Rule 24b-2 of the Securities Exchange Act of 1934.
- (b) Reports on Form 8-K

Registrant furnished two reports on Form 8-K during the quarter for which this report is filed.

On July 17, 2003 the Registrant furnished a report on Form 8-K furnishing to the Securities and Exchange Commission a copy of a press release dated July 17, 2003, announcing its financial results for the quarter ended June 30, 2003.

On July 18, 2003 the Registrant furnished a report on Form 8-K furnishing to the Securities and Exchange Commission certain information relating to the quarter ended June 30, 2003 disclosed during the registrant's conference call with investment analysts on July 17, 2003 and certain information required by Regulation G.

34

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

	EQUIFAX INC. (Registrant)	
Date: November 12, 2003	/s/ THOMAS F. CHAPMAN	
	Thomas F. Chapman, Chairman and Chief Executive Officer	
Date: November 12, 2003	/s/ DONALD T. HEROMAN	
	Donald T. Heroman Chief Financial Officer	

35

INDEX TO EXHIBITS

The following documents are being filed with this Report.

Exhibit No.	Description
10.1	Agreement for Operating Support Services ⁽¹⁾
31.1	Certification of Thomas F. Chapman, Chairman and Chief Executive Officer of Equifax Inc., Pursuant to Rule 13a-14(a).
31.2	Certification of Donald T. Heroman, Chief Financial Officer of Equifax Inc., Pursuant to Rule 13a-14(a).
32.1	Certification of Thomas F. Chapman, Chairman and Chief Executive Officer of Equifax Inc., Pursuant to 18 U.S.C. Section 1350, as adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
32.2	Certification of Donald T. Heroman, Chief Financial Officer of Equifax Inc., Pursuant to 18 U.S.C. Section 1350, as adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

(1) Document omits information pursuant to a Request for Confidential Treatment under Rule 24b-2 of the Securities Exchange Act of 1934.

36

QuickLinks

INDEX

PART I. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK.

ITEM 4. CONTROLS AND PROCEDURES.

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS
ITEM 6. EXHIBITS AND REPORTS ON FORM 8-K

SIGNATURES INDEX TO EXHIBITS Equifax / IBM Confidential

AGREEMENT

FOR

OPERATIONS SUPPORT SERVICES

This Agreement is entered into as of July 1, 2003 (the "Effective Date"), between

1. International Business Machines Corporation, a New York corporation ("IBM"),

AND

2. Equifax Inc., a Georgia corporation ("Equifax").

This Agreement supersedes and replaces certain existing agreements between Equifax (or its Affiliates) and IBM (or its Affiliates), which are listed in Schedule M (Existing IBM-Equifax Agreements Superseded By This Agreement).

The Parties agree to the terms and conditions set forth in this Agreement (which are those set forth in the main body of this Agreement, and the various Schedules, Exhibits, Attachments, Appendices and Supplements attached to and referenced in this Agreement) and in each Statement of Work executed by the Parties referencing this Agreement.

The Commencement Date of the Services under this Agreement will be August 7, 2003. This Agreement is subject to ratification by the Equifax Board of Directors at its meeting to be held August 6, 2003. If such ratification does not occur, this Agreement shall be void *ab initio* and of no further force or effect.

Signed for and on behalf of IBM:

INTERNATIONAL BUSINESS MACHINES CORPORATION

Signature: /s/ Arthur G. Gopfert

Title: /s/ Global Senior Project Executive (July 31, 2003)

Signed for and on behalf of Equifax:

EQUIFAX INC.

Signature: /s/ Owen V. Flynn

Title: /s/ Chief Technology Officer (July 31, 2003)

Equifax / IBM Confidential

TABLE OF CONTENTS

			Page
1.	PURPOSE/S	STRUCTURE/TERM OF AGREEMENT	1
	1.1	Purpose of Agreement	1
	1.2	Structure of Agreement	2
	1.3	Term of Agreement	3
	1.4	Extension of Services	3
2.	DEFINITIO	ons	4
	2.1	Certain Definitions	4
	2.2	Other Terms	13
3.	THE SERV	ICES	13
	3.1	Obligation to Provide Services	13

	3.2	Performance	14
	3.3	Disaster Recovery Services	14
	3.4	Audits	1:
	3.5	IBM Cooperation with Authorized User Examinations	10
	3.6	Facilities	10
	3.7	Security	1
	3.8	Technology Refresh	1
	3.9	Machines And Third Party Service Agreements	1'
	3.10	Equifax Owned Software — Existing	19
	3.11	Third Party Provider Software — Existing	19
	3.12	New Software Added During the Term	19
	3.13	Changes to the Software	20
	3.14	Terms of Acquisition by IBM of Third Party Provider Software	2
	3.15	Affiliates	2
	3.16	Viruses	2
4.	WARRA	ANTIES/REPRESENTATIONS/COVENANTS	2.
	4.1	Work Standards	2.
	4.2	Non-infringement	22
	4.3	Disabling Code	22
	4.4	Authorization and Enforceability	22
	4.5	Maintenance	2.
	4.6	Efficiency and Cost Effectiveness	2:
	4.7	Software Ownership or Use	2.
	4.8	Inducements	2.
	4.9	Disclaimer	2.
	4.10	Legal and Regulatory Compliance	2:
	4.11	Year 2000 Warranty	20
	4.12	Covenant of Cooperation and Good Faith	20
5.		FER, TRANSITION AND TRANSFORMATION	20
	5.1	Transition/Transformation Plans	20
	5.3	Resources and Facilities	2'
6.	GOVER		29
	6.1	Relationship and Contract Governance Model	25
	6.2	Meetings	29
	6.3	Procedures Manual.	29
	6.4	Change Management Process	31
7.			
7.	IBM Pers	rsonnel	30

	7.2	Replacement of Personnel	30
	7.3	Key IBM Personnel Positions	31
	7.4	Retention of Experienced Personnel	32
8.	RELATIO	ONSHIP PROTOCOLS	32
	8.1	Annual Updating of Schedules to this Agreement	32
	8.2	Required Consents	33
	8.3	Appointment as Attorney In Fact	34
	8.4	Conflicts of Interests	35
	8.5	Alternate Providers	35
	8.6	Use of Subcontractors	36
		ii	
	8.7	Equifax Approvals and Notification	37
9.	CHARGE	S	37
10.	INTELLE	CTUAL PROPERTY RIGHTS	37
	10.1	Ownership of Materials	38
	10.2	Obligations Regarding Materials	39
	10.3	Authorized Users	39
11.	CONFIDE	ENTIALITY/DATA SECURITY	39
	11.1	Confidential Information	39
	11.2	Obligations	40
	11.3	Exclusions	41
	11.4	Loss of Company Information	41
	11.5	Limitation	42
	11.6	Equifax Data	42
	11.7	Data Privacy	42
12.	TERMINA		43
	12.1	Termination By Equifax	43
	12.2	Termination by IBM	44
	12.3	Services Transfer Assistance	44
	12.4	Equitable Remedies	45
	12.5	Other Rights Upon Termination	45
	12.6	Effect of Termination/Survival of Selected Provisions	47
	12.7	Savings Clause	47
13.	LIABILIT		47
	13.1	Liability Caps	47
	13.2	*	48
	13.3	Direct Damages and Cover Charges	48
	13.4	Dependencies	48
	13.5	Remedies	48

Omitted pursuant to Rule 24b-2 of the Securities Exchange Act of 1934, as amended (the "Exchange Act").

14.	INDEMNITIE	S	49
	14.1	Indemnity by IBM	49
	14.2	Indemnity by Equifax	50
	14.3	Employment Actions	52
	14.4	Exclusive Remedy	52
	14.5	Infringement	52
	14.6	Indemnification Procedures	52
	14.7	Limitation	53
15.	INSURANCE	AND RISK OF LOSS	53
	15.1	IBM Insurance	53
	15.2	Risk of Property Loss	54
	15.3	Mutual Waiver of Subrogation	54
16.	DISPUTE RES	SOLUTION	55
	16.1	Dispute Resolution Procedures	55
	16.2	Continued Performance	56
17.	GENERAL		56
	17.1	Relationship of Parties	56
	17.2	Entire Agreement, Updates, Amendments and Modifications	57
	17.3	Force Majeure	57
	17.4	Waiver	58
	17.5	Severability	58
	17.6	Counterparts	58
	17.7	Governing Law	58
	17.8	Binding Nature and Assignment	59
	17.9	Notices	59
	17.10	No Third Party Beneficiaries	60
	17.11	Other Documents	60
	17.12	Consents and Approvals	60
	17.13	Headings	60
	17.14	Remarketing	60
	17.15	Commencement of Actions	61
		iv	
	17.16	IBM Logo Products Warranties	61
	17.17	Local Enabling Agreements	61
		v	

Form of Local Enabling Agreement

Schedule

A	Services
В	Service Levels
C	Charges
D	Human Resources
E	Equipment
F	Software
G	Third Party Agreements
Н	Facilities
I	Transition/Transformation
J	IT Management Process Improvement Program
K	Operational Reports
L	Governance
M	Existing IBM-Equifax Agreements Superseded By The Agreement
N	Projects
О	Services Transfer Assistance
P	Data Protection
Q	Country-Specific Regulatory and Legal Requirements
R	Listed Subcontractors
S	Disaster Recovery Services
T	Security Procedures

^{**} Copies of the above attachments will be provided to the Commission upon request.

V

Title

Equifax / IBM Confidential

PURPOSE/STRUCTURE/TERM OF AGREEMENT

1.1 Purpose of Agreement

- (a) IBM is a provider of a broad range of operations support services for financial services companies including, without limitation, information technology, information management, communications and related services, and is experienced and skilled in the administration, management, provision and performance of such services and the business functions, responsibilities and tasks attendant with such services. IBM desires (i) to continue to provide certain of these operations support services to the Equifax Group for the Equifax Business, and to continue to perform and assume the functions, responsibilities and tasks attendant with such operations support services as currently performed by IBM for the Equifax Business and the Equifax Group; and (ii) to provide additional quantities and elements of these and other operations support services to the Equifax Group for the Equifax Business and to perform and assume the functions, responsibilities and tasks attendant with such operations support services as currently performed by the Equifax Group or as envisioned to be required for the Equifax Business and the Equifax Group, all as specifically set forth in this Agreement. Equifax desires that such operations support services for the Equifax Business and the Equifax Group and the attendant functions, responsibilities and tasks, be performed and assumed by IBM. This Agreement documents the terms and conditions under which (i) the Equifax Group will obtain such operations support services from IBM and (ii) IBM will administer, manage, support, provide and perform such services and the functions, responsibilities and tasks attendant with such services, for the Equifax Group.
- (b) Subject to Section 1.1(d), the Parties have identified goals and objectives that they intend that IBM's performance pursuant to this Agreement will assist the Parties to achieve. These goals and objectives include the following:
- (i) Realigning of the scope of IBM's services with Equifax Group's business needs;
- (ii) Providing world class service delivery based on industry best practices and standards, specifically aimed at improving information technology productivity and reliability, and speed to market of new products;
- (iii) Achieving significant cost savings, beginning in 2003, and continuing from year to year, enabling resources to be redirected from maintaining legacy systems to new growth initiatives, including the development of new products;
- (iv) Providing pricing structures that give Equifax Group better visibility into and control over its total life-cycle spending for information technology systems and platforms;
- (v) A commitment by IBM to maintaining the technological currency of the information technology systems and resources used to perform the Services;

- (vi) Providing market competitive pricing throughout the Term of this Agreement;
- (vii) Establishing a global relationship and Agreement governance structure that shall facilitate the use of consistent approaches and processes across all the countries in which Equifax Group operates;
- (viii) Achieving seamless service across the U.S., Canada and Europe that leverages IBM competency centers;

- (ix) Establishing a single, integrated delivery model across geographies that minimizes the number touch-points between IBM and Equifax Group and eliminates the need for Equifax Group to function as an integrator;
- (x) Delivering a solution that places risk on the Party in the best position to manage and control the risk;
- (xi) Providing value-added strategic thought, vision and leadership from IBM;
- (xii) Providing an opportunity to transition the Services back to the Equifax Group or to another service provider from IBM with minimal disruption;
- (xiii) Delivering a solution that will enable IBM to be successful;
- (xiv) Securing favorable rates for current and additional resource consumption and for reductions in resource consumption and increasing flexibility regarding resources chargeable and available to the Equifax Group and committed by IBM to the Equifax Group;
- (xv) Enhancing the current functionality of the Equifax Group's processes, systems and service levels covered under this Agreement;
- (xvi) Proactively defining and proposing cost-effective solutions to improve the efficiency and functionality of the information management systems operations of the Equifax Group in support of the Equifax Business;
- (xvii) Ensuring the efficiency, stability and security of existing and future processes, systems and service levels; and
- (xviii) Evolving the support services, processes, systems and service levels to meet the dynamic requirements of the Equifax Group and Equifax Business.
 - (c) IBM recognizes that the Equifax Group expects to be treated as a valued customer and agrees that the definition of customer satisfaction goes beyond IBM's performance against established service levels and requires that IBM exhibit a customer service attitude focused on assisting Equifax where commercially reasonable to attain the goals and objectives described in Section 1.1(b), including, without limitation, reducing the operations support costs of and improving service levels to the Equifax Group and the customers of the Equifax Group.
 - (d) The provisions of this Section 1.1 are intended to be a statement of the purpose of this Agreement and are not intended to alter the plain meaning of the terms and conditions of this Agreement or to require either Party to undertake performance obligations beyond those set forth in this Agreement. To the extent that the terms and conditions of this Agreement are unclear or ambiguous, such terms and conditions are to be interpreted and construed consistent with the purposes set forth in this Section 1.1.

1.2 Structure of Agreement

(a) For certain purposes under this Agreement, the Services will be grouped around the following technology Platforms: (i) Mainframe, Midrange, Parallel Systems, MicroLAN and Service Desk, which collectively comprise and are sometimes referred to as the "Operations" Service Tower; (ii) Voice and Data (including WAN, LAN and MAN), which collectively comprise and are sometimes referred to as the "Network" Service Tower; and (iii) Disaster Recovery, which is a cross-functional Service Tower that spans both the Operations and Network Service Towers.

2

- (b) Although the scope of the Services does not include * as of the Execution Date, Equifax has agreed to purchase a certain amount of * from IBM under this Agreement. IBM has agreed to supply such services to Equifax and has included in the charges under Schedule C (Charges) a portion of the cost to Equifax of such work. Any * services ordered from IBM under this Agreement shall be authorized by a Statement of Work prepared and issued under this Agreement for such services, which Statement of Work will contain special terms and conditions applicable to the performance of the * work.
- (c) This Agreement is comprised of the provisions set forth in this Agreement and the various Schedules, Exhibits, Attachments, Appendices and Supplements referenced herein.
- (d) IBM and Equifax will be the primary contracting parties under this Agreement. Non-U.S. Affiliates of the Parties shall sign Local Enabling Agreements in order for the Services to be provided and received in their respective home countries in accordance with Section 17.17 of this Agreement.

1.3 Term of Agreement

(a) The term of this Agreement will begin as of the Effective Date and will terminate upon the later to occur of (a) the tenth (10th) anniversary of the Execution Date, or (b) the completion of the Services Transfer Assistance (the "Term"), unless earlier terminated in accordance with the provisions of this Agreement or extended pursuant to Section (a). The Commencement Date shall be August 7, 2003 for the Canada, Ireland, Spain, U.K and U.S. Country Locations. Equifax will remain responsible for the performance of the Services from the Execution Date to the Commencement Date for all Country Locations, and IBM will provide assistance to Equifax in managing the delivery of the Services during that period.

1.4 Extension of Services

(a) Equifax may request and IBM will extend the provision of the Services or the Services Transfer Assistance for up to one (1) year ("Extension Period") upon not less than sixty (60) days prior written notice before the scheduled termination or expiration of the provision of the Services or Services Transfer Assistance, or if applicable, notice given within thirty (30) days after the effective date of a notice of termination for any reason by either Party, other than Termination for Convenience. Equifax shall have three (3) such optional Extension Periods. The charges set forth in Schedule C (Charges), subject to adjustment and change as provided in Schedule C (Charges), shall apply during any Extension Periods. However, in the event Equifax is in default with respect to the payment of any

amounts under this Agreement at the start of the Extension Period, IBM will extend the provision of such Services or Services Transfer Assistance as described in this Section 1.4, only if Equifax cures such default and prepays three (3) months of the Monthly Charges during such Extension Period and a reasonable projection of other charges due for such three (3) calendar months period. Equifax will be credited any unused portions of such prepayment for the remaining part of such Extension Period covered by such unused portion of such prepayment.

* Information deleted pursuant to Rule 24b-2 of the Exchange Act.

3

2. DEFINITIONS

2.1 Certain Definitions

In this Agreement, the following terms will have the following meanings:

1998 Agreement Means that certain Master Agreement for Operations Support Services, dated January 1, 1998, as amended, between

IBM and Equifax.

Action has the meaning given in Section 17.7.

*

Affiliates means, with respect to a Party, any entity at any time Controlling, Controlled by or under common Control with such

Party.

Agreement means this Agreement for Operations Support Services, including all of the various Schedules, Exhibits, Attachments,

Appendices and Supplements attached to and referenced herein and any Statements of Work issued under and

referencing this Agreement.

*

Applications Software means those programs and programming, including all supporting documentation and media, that perform specific

user related data processing, data management and telecommunications tasks, including updates, enhancements, modifications, releases and Derivative Works thereof. Applications Software as of the Execution Date is listed in

Schedule F (Software).

Applications Software — Equifax means the Applications Software provided by or through Equifax as of the Execution Date, as listed on Schedule F

(Software) under such heading, or during the Term of this Agreement in accordance with Section 6.4. If additional Applications Software is provided by or through Equifax during the Term or if previously provided Applications Software is removed, Schedule F (Software) will be amended to reflect the addition or removal of such Applications

Software in accordance with Sections 8.1 and 17.2.

Applications Software —IBM means the Applications Software provided by or through IBM as of the Execution Date, as listed on Schedule F

(Software) under such heading, or during the Term of this Agreement in accordance with Section 6.4. If additional Applications Software is provided by or through IBM during the Term or if previously provided Applications Software is removed, Schedule F (Software) will be amended to reflect the addition or removal of such Applications

Software in accordance with Sections 8.1 and 17.2.

* Information deleted pursuant to Rule 24b-2 of the Exchange Act.

4

Authorized User means a person or entity (which shall be either an Affiliate or a customer of Equifax or an entity in which an Equifax

Group member is a joint venturer, partner, member or equity owner) authorized to use the Services, including without

limitation the System, by Equifax.

Baseline(s) has the meaning given in Schedule C.

Cable or Cabling means the wires or cables that interconnect Machines and/or connect a Machine to a facility connection.

Change of Control means the transfer of the Control of a Party, or a sale of substantially all of the assets of a Party, from the persons or

persons who hold such Control on the Execution Date to another person or persons, but shall not include a transfer of

the Control of a Party to an Affiliate of such Party.

Claim has the meaning given in Section 14.6(a).

Code has the meaning given in Article 10.

Commencement Date means the date(s) on which IBM's becomes responsible for performance the Services, which date may be different

from one Country Location to another.

Commercially Reasonable Efforts Whether or not capitalized, means taking such steps and performing in such a manner as a well managed company

would undertake where such company was acting in a determined, prudent and reasonable manner to achieve a

particular desired result for its own benefit.

Company Information has the meaning given in Section 11.1.

Confidential Information has the meaning given in Section 11.1.

Contract Change means any change(s) to any of the documents comprising this Agreement, which shall be carried out as provided in

Sections 17.2 and 8.1.

Contract Year means any consecutive twelve (12) month period commencing on the Execution Date or any anniversary thereof

during the Term.

Control, Controlling, or Controlled means possessing, directly or indirectly, the power to direct or cause the direction of the management and policies of

an entity through contractual right or ownership of greater than fifty (50%) percent of the voting securities of such

entity.

Country Location has the meaning given in Schedule C (Charges).

Data Center means the data centers from which the Services are provided located in the Facilities listed in Schedule H (Facilities).

Derivative Work means a work based on one or more pre-existing works, including without limitation, a condensation, transformation,

expansion or adaptation, which would constitute a copyright infringement if prepared without authorization of the

owner of the copyright of such pre-existing work.

has the meaning given in Section 13.3.

Develop has the meaning given in Article 10.

Direct Damages

Direct Damages Caps has the meaning given in Section 13.1(b).

5

Disabling Code means Code which is designed for the purpose and has the effect of disabling or otherwise shutting down one or more

software programs or systems and/or hardware or hardware systems.

Disaster Recovery means the cross-functional Service Tower that is composed of the Disaster Recovery Services as specified in

Schedule S for the environment that spans both the Operations and Network Service Towers.

Disaster Recovery Services described in Schedule A (Services) and/or Schedule S (Disaster Recovery

Services).

Effective Date means the date set forth on the initial page of this Agreement.

Elements of the Services has the meaning given in Section 17.14.

EMU Matters means the failure of any product or service to correctly process or properly exchange monetary data in euro

denominations accurately.

Equifax Means one or more members of the Equifax Group, as the context requires, unless it is clear from the context that the

term "Equifax" is instead intended to mean Equifax Inc. in that particular instance.

Equifax Business means the businesses engaged in by the Equifax Group.

Equifax Code means Code Developed by IBM and/or its subcontractors independently or jointly with the Equifax Group and/or

their contractors, as part of the Services. Equifax Code shall not include any IBM Derivative Code.

Equifax Data means all information, whether or not Confidential Information, entered in Software or Machines by or on behalf of

Equifax and information derived from such information, including as stored in or processed through the Machines or

Software.

Equifax Direct Damages Cap has the meaning given in Section 13.1(b).

Equifax Derivative Code means Code Developed by IBM and/or its subcontractors independently or jointly with the Equifax Group and/or

their contractors, as part of the Services, which constitutes Derivative Work of software for which the copyright is

owned by the Equifax Group and/or their contractors.

Equifax Group means individually and collectively Equifax and its existing and future Affiliates that are using and/or receiving any

portion of the Services.

Equifax Owned Software means Software that is owned by Equifax.

Equifax Provided Hardware

means the computer equipment peripheral devices, storage media, Cabling, connectors, the Data Network, the LAN, telephone equipment and other equipment (however described) provided from time to time by the Equifax Group for use by IBM to perform and deliver the Services and fulfill its obligations under this Agreement. The Equifax Provided Hardware as of the Execution Date is listed on and/or referred to in Schedule E (Machines). If additional Equifax Provided Hardware is added or previously provided Equifax Hardware is removed during the Term of this Agreement in accordance with Section 6.4, Schedule E (Machines) shall be updated pursuant to Sections 8.1 and 17.2 to reflect the then-current Equifax Provided Hardware.

Equifax Provided Office Furnishings

means the desks, chairs, filing cabinets, office cube partitions and other office furniture (however described) provided from time to time by the Equifax Group for use by IBM to perform and deliver the Services and fulfill its obligations under this Agreement. The Equifax Provided Office Furnishings as of the Execution Date are listed on and/or referred to in Schedule E (Machines). If additional Equifax Provided Hardware is added or previously provided Equifax Hardware is removed during the Term of this Agreement in accordance with Section 6.4, Schedule E (Machines) shall be updated pursuant to Sections 8.1 and 17.2 to reflect the then-current Equifax Provided Office Furnishings.

Equifax Retained Function

means any Process-Element intersection in the Scope Models in Schedule A (Services) in which Equifax is designated as the Actor or for which IBM is not designated as the Actor.

Equifax Software

means Applications Software-Equifax and Systems Software-Equifax.

Equifax Works

means literary works of authorship (other than Code) Developed by IBM and/or its subcontractors independently or jointly with the Equifax Group and/or its contractors under this Agreement, specifically for the Equifax Group or the Equifax Business or specifically for the purpose of providing the Services, including without limitation user manuals, charts, graphs and other written documentation, and machine-readable text and files, but shall not include any Derivative Works of any works in which the copyright is owned by IBM, its Affiliates or subcontractors.

Euro-Ready

means that a product or service, when used in accordance with its associated documentation and used properly in accordance with its specifications, will correctly process monetary data in the euro denomination and will perform in accordance with the euro currency formatting conventions including the euro sign assuming that all other products (i.e. hardware, software, firmware, etc.) that are used with this product are also Euro-Ready.

Execution Date

means the date this Agreement is signed by both Parties.

Extension Period

has the meaning given in Section 1.4.

Facilities

means the facilities listed in Schedule H (Facilities).

Force Majeure Event

has the meaning given in Section 17.3(a).

7

IBM

Means any or all of International Business Machines Corporation and its Affiliates, as the context requires, unless it is clear from the context that the term "IBM" is instead intended to mean International Business Machines Corporation in that particular instance.

IBM Code

means Code Developed by IBM personnel at IBM's expense and not as part of the Services, but used to provide the Services, which code does not constitute a Derivative Work of any software owned by the Equifax Group, IBM, or their respective Affiliates or contractors or subcontractors. IBM Code shall not include any Equifax Derivative Code.

IBM Derivative Code

means Code Developed under this Agreement, which constitutes Derivative Works of software for which the copyright is owned by IBM, its Affiliates or its subcontractors.

IBM Direct Damages Cap

has the meaning given in Section 13.1(a)(i).

IBM Euro-Ready

means that an IBM Logo product, when used in accordance with its associated documentation and used properly in accordance with its specifications, will correctly process monetary data in the euro denomination and will perform in accordance with the euro currency formatting conventions including the euro sign assuming that all other products (i.e. hardware, software, firmware, etc.) that are used with this product are Euro-Ready. IBM hardware products that are IBM Euro-Ready may or may not have an engraved Euro sign key on their keyboards.

IBM Indemnitees

has the meaning given in Section 14.2.

IBM Interfaces

means Code and/or literary works of authorship created at IBM's expense, by IBM personnel and/or its contractors and not as part of the Services, but used to provide the Services, and interface or describe and instruct regarding the interface, between and among Applications Software and the Systems Software, which does not constitute a Derivative Work of any software or literary works of authorship owned by the Equifax Group, IBM, or their respective Affiliates or contractors, including without limitation, user manuals, charts, graphs and other written documentation, and machine-readable text and files.

IBM Logo Products

has the meaning given in Section 4.11.

IBM Machines

means the computer equipment, peripheral devices, storage media, cabling, connectors, extenders and other equipment (however described) including without limitation, modems, routers and termination boxes for the Network located in the Facilities and other Equifax Group Sites, including without limitation Data Center and at the Network Locations, provided by or through and used from time to time by IBM to perform and deliver the Services and fulfill its obligations under this Agreement. The IBM Machines as of the Execution Date are listed on Schedule E (Machines), which schedule shall be updated pursuant to Section 8.1 during the Term to reflect the then current IBM

Machines.

IBM Software

means the Applications Software—IBM and Systems Software—IBM.

8

IBM Year 2000 Compliance or Compliant

means that the product will, subject to the provisions of Section 4.9(b)), when used in accordance with its associated documentation, (i) accurately process and handle date data (including but not limited to, calculating, comparing and sequencing, to the extent that the product's specifications provide for such processing or handling of date data) within, from, into and between the twentieth and twenty-first centuries, and the years 1999 and 2000, including leap year calculations, to the extent that all other products used in combination with such product properly exchange date data with it, and (ii) will properly exchange date data with other IBM Logo Products that are IBM Year 2000 Compliant, provided that such IBM Logo Products are specified by IBM to operate together as part of a system.

IBM Works

means literary works of authorship (other than Code) Developed at IBM's expense, by IBM personnel and/or its contractors and not specifically for the Equifax Group or the Equifax Business or not specifically for the purpose of providing the Services, but used to provide the Services, including without limitation user manuals, charts, graphs and other written documentation and machine-readable text and files, but shall not include any Derivative Works of any works in which the copyright is owned by Equifax or its Affiliates or subcontractors.

Indemnified Party has the meaning given in Section 14.4.

Indemnifying Party has the meaning given in Section 14.5(a).

Indemnitee has the meaning given in Section 14.1.

Key IBM Personnel Positions

means those personnel furnished by IBM to perform the Services who occupy positions designated as Key IBM

Personnel Positions in Schedule D (Human Resources) or pursuant to Section 7.3.

Knowledge Retention Personnel Has the meaning given in Section 7.4(a).

Listed Subcontractors has the meaning given in Section 8.6(a).

Local Enabling Agreements has the meaning given in Section 17.17.

Edeal Endotting / Edecition 17.17.

Losses means all losses, liabilities, damages, penalties and claims (including taxes and all related interest and penalties

incurred directly with respect thereto), and all related costs, expenses and other charges *.

Machines Means the IBM Machines and Equifax Provided Hardware.

Maintenance Release Means those Software fixes and updates provided by the Software vendors as part of normal maintenance service for

the Software for which there is no charge by such vendors in addition to periodic maintenance charges, if any.

Materials Means the Equifax Code, the Equifax Derivative Code, the Equifax Works, the IBM Code, the IBM Derivative Code,

the IBM Works and the IBM Interfaces.

* Information deleted pursuant to Rule 24b-2 of the Exchange Act.

9

MicroLan means desktop support. For the avoidance of doubt, the terms "MicroLan," and "Desktop" are used interchangeably in

this Agreement.

Network means the Service Tower that is composed of the Voice and Data (i.e., WAN, LAN and MAN) Platforms, as such

Platforms are defined in Schedule C.

New Services has the meaning given in Section 10.1 of Schedule C (Charges).

Notice has the meaning given in Section 16.1(b).

Operations means the Service Tower that is composed of the Mainframe, Mid-Range, Parallel Systems, MicroLan and Service

Desk Platforms, as such Platforms are defined in Schedule C.

Parties means IBM and Equifax.

Party means IBM or Equifax.

Platform is a term used in certain circumstances under this Agreement to subdivide the Services comprising a Service Tower

(i.e., Operations, Network or Disaster Recovery) into sub-groups/clusters of Services.

Project has the meaning given in Schedule N (Projects).

Project Executive has the meaning given in Section 7.1.

Required Consents means any consents or approvals required to be obtained (a) to allow IBM, its Affiliates and its approved

subcontractors to assume financial and/or support, operational, management and administrative responsibility for the Equifax Software, the Equifax Provided Hardware and the Equifax Provided Office Furnishings in connection with the Services; (b) for the licensing, transfer and/or grant of the right to the Equifax Group, as permitted by each license, to use the IBM Software and IBM Machines as contemplated by this Agreement; and (c) for the Equifax Group and IBM, its Affiliates and its approved subcontractors to have access to and use of the space, equipment, software and/or third party services provided under the Third Party Agreements in connection with the Services as

contemplated by this Agreement.

Resource Unit ("RU") Has the meaning given in Schedule C (Charges).

Service Level Credits has the meaning set forth in Schedule B (Service Levels).

Service Employees has the meaning given in Section 12.5(g).

Service Levels means the standards of performance to be met or exceeded by IBM in providing the Services. The Service Levels are

set forth in Schedule B (Service Levels).

Service Tower is a term used in certain circumstances under this Agreement to subdivide the Services into three high-level groups of

Services: Operations, Network and Disaster Recovery.

10

Services

means all functions, responsibilities, tasks and activities: (a) described in this Agreement that are to be performed by IBM under this Agreement; or (b) that are directly related to information technology services and were performed for the Equifax Group in the immediately preceding twelve (12) months before the Execution Date by the Transferred Employees or by Equifax Group employees whose functions are assumed by IBM or displaced under this Agreement; or (c) that were performed or required to be performed by IBM and/or its Affiliates and subcontractors for the Equifax Group in the immediately preceding twelve (12) months prior to the Execution Date under the 1998 Agreement or any of the other agreements identified in Schedule M as being superseded by this Agreement; or (d) that were performed in the immediately preceding twelve (12) months before the Execution Date by assets conveyed or made available to IBM or displaced as a result of this Agreement. For the avoidance of doubt, the Parties agree that references in this definition to the 1998 Agreement and other agreements identified in Schedule M are for purposes of defining the scope of the Services under this Agreement—such references are not intended to incorporate by reference into this Agreement any legal terms and conditions of such agreements, all of which are superseded by this Agreement.

Services Transfer Assistance

has the meaning given in Section 12.5.

Software

means IBM Software and Equifax Software.

Span Elements

has the meaning provided in Schedule A (Services).

Statement of Work

has the meaning given in Schedule N (Projects).

System

means the Machines, Software and Network covered under this Agreement and the operating environment therefor.

Systems Software

means those programs and programming (including all supporting documentation and media) that perform tasks related to the functioning of the data processing, and telecommunication equipment which is used to operate the Applications Software or otherwise to support the provision of the Services by or through IBM under this Agreement, whether or not licensed to IBM. Systems Software may include but is not limited to, database creation and management software, application development tools, operating systems, software utilities, data security software, data network software, communications monitors and data base managers. Systems Software as of the Execution Date is listed in Schedule F (Software), which schedule shall be updated pursuant to Sections 8.1 and 17.2 during the Term to reflect the then current Systems Software. In Schedule A (Services), Systems Software is subdivided into and includes two groups of Software identified as "Platform Software" and "Infrastructure Software."

11

Systems Software -Equifax

means the Systems Software and general purpose software such as the database creation and management software, utility software and applications development tools software provided by or through Equifax on the Execution Date, as listed in Schedule F (Software) under such heading, or during the Term of this Agreement in accordance with Section 6.4. If additional Systems Software is provided by or through Equifax during the Term or if previously provided Systems Software is removed, Schedule F (Software) will be amended to reflect the addition or removal of such Systems Software in accordance with Sections 8.1 and 17.2

Systems Software—IBM means the Systems Software provided by or through IBM on the Execution Date, as listed in Schedule F (Software)

under such heading, or during the Term of this Agreement in accordance with Section 6.4. If additional Systems Software is provided by or through IBM during the Term or if previously provided Systems Software is removed, Schedule F (Software) will be amended to reflect the addition or removal of such Systems Software in accordance

with Sections 8.1 and 17.2

Term has the meaning given in Section 1.3 and any extension and renewal term described in this Agreement.

Termination Charges means the charges designated as such that are set forth in Exhibit C-8.

Third Party Agreements means those contractual, leasing and licensing arrangements to which one or more members of the Equifax Group is a

party and pursuant to which a member of the Equifax Group receives any third party products, software and/or services that IBM will need to access or use in providing the Services and for which IBM is assuming financial, management and/or administrative responsibility under this Agreement. Third Party Agreements in effect as of the Execution Date are listed on Schedule G (Third Party Agreements), which schedule shall be updated pursuant to

Sections 8.1 and 17.2 during the Term to reflect the then-current Third Party Agreements.

Third Party Provider means a business or entity other than a member of the Equifax Group or IBM and its Affiliates that provides

products, software and/or services under a Third Party Agreement, in support of the provision of the Services by IBM.

Third Party Service Agreements means those Third Party Agreements, excluding leases and license agreements, pursuant to which a member of the

Equifax Group receives third party services that IBM will need to access or use in providing the Services and for which IBM is assuming financial, management and/or administrative responsibility under this Agreement. Third Party Services Agreements in effect as of the Execution Date are listed on Schedule G (Third Party Agreements), which schedule shall be updated pursuant to Sections 8.1 and 17.2 during the Term to reflect the then-current Third

Party Service Agreements.

Transferred Employees has the meaning given in Section 5.2.

Trade Secrets has the meaning given in Section 11.1.

Transition Cover Costs has the meaning given in Section 13.3(b).

12

Transition Personnel has the meaning given in Section 5.1(e).

Transformation/ Transformation Plan has the meaning given in Section 5.1(a).

Use means, in the context of Software, to use, copy, maintain, modify, enhance, distribute or create Derivative Works as

permitted under the license for such Software.

Version means those Software updates that generally add function to the existing Software and may be provided by the

Software vendors at a fee over and above the standard periodic software maintenance costs.

Virus or Viruses means computer instructions that are intended, designed and have the effect of adversely affecting the specified

operation, security or integrity of a computing, telecommunications or other digital operating or processing system or

environment.

Wind-Down Expenses means the net amount, after IBM takes commercially reasonable action to mitigate the amount thereof, that will

reimburse IBM for the actual reasonable costs for severance, relocation and unbillable time that IBM incurs in the placement on other customer accounts or severance of IBM personnel primarily employed to provide the Services;

provided, however, Equifax shall have the right to mitigate such costs by hiring such IBM personnel.

2.2 Other Terms

Other capitalized terms used in this Agreement are defined where they are used and have the meanings there indicated. Those terms, acronyms and phrases not defined in this Agreement but in common usage in the information technology ("IT") industry or other pertinent business context shall have their generally understood meanings in the IT industry.

3. THE SERVICES

3.1 Obligation to Provide Services

- (a) Starting on the Commencement Date in each Country Location and continuing during the Term, IBM shall provide the Services to, and perform the Services for, the Equifax Group, for use by the Equifax Group and other Authorized Users.
- (b) In performing and providing the Services, the relationship of IBM with the members of the Equifax Group will be as an independent contractor. However, as a result of its position in providing and performing the Services, the Parties acknowledge that certain employees of IBM and each of its Affiliates providing portions of the Services may have a unique knowledge of the information technology operations of the members of the Equifax Group that no employee of a member of the Equifax Group will have in full, and employees of IBM and each of its Affiliates providing portions of the Services will be interacting with the employees, executive management and accountants to the Equifax Group and the members thereof, and will be performing functions that would otherwise be performed by employees of the Equifax Group.

activities and tasks either (i) were performed during the twelve (12) months preceding the Commencement Date by Equifax personnel (employees and contractors) or assets transitioned or made available to IBM or displaced as a result of this Agreement or (ii) are determined to be required for the proper performance and provision of the Services or are an inherent part, or a necessary sub-part included within, the Services, such functions, responsibilities, activities and tasks shall be deemed to be implied by and included within the scope of the Services to the same extent and in the same manner as if specifically described in this Agreement. Each such determination shall be made by agreement of the Parties or resolved pursuant to the dispute resolution provisions of Article 16. For the avoidance of doubt: (i) this Section 3.1(c) shall not be interpreted to include as deemed parts of the Services any functions, responsibilities, activities or tasks that are not related to the provision of information technology services — for example, managing the company's annual charitable fund-raising campaign; and (ii) IBM's obligations with respect to the interpretation of or compliance with laws and regulations as they apply to the Equifax Business shall be limited to those obligations expressly set forth in this Agreement, the criteria in this Section 3.1(c) notwithstanding.

(d) IBM shall refresh and supplement the infrastructure, tools, and other factors of production used by IBM in providing its services in order to keep pace with technological advances and advances in the methods of delivering services, where such advances are at the time pertinent and in general use within the information technology industry to enable Equifax to take advantage of technological advancements in its industry and support Equifax's efforts to maintain competitiveness in the markets in which it competes. For the avoidance of doubt, this provision does not apply to Span Elements for which a specific refresh schedule has been provided or for which Equifax has retained financial responsibility for refreshment.

3.2 Performance

IBM agrees to perform the Services in a manner that will meet or exceed each of the applicable Service Levels set forth in Schedule B (Service Levels), subject to the limitations and in accordance with the provisions set forth in this Agreement, and to standards satisfied by well-managed operations performing services similar to the Services. IBM further agrees that it shall perform the Services at least at the same level and with at least the same degree of accuracy, quality, completeness, timeliness, responsiveness and efficiency as was provided prior to the Commencement Date by or for Equifax provided Equifax can demonstrate to IBM such levels and degrees utilizing data for the twelve (12) months immediately preceding the Commencement Date.

3.3 Disaster Recovery Services

IBM will provide Disaster Recovery Services in accordance with Schedules A and S. If IBM fails to provide Disaster Recovery Services to the extent and in accordance with the time table set forth in Schedule A (Services) or Schedule S (Disaster Recovery Services) for a period as set forth in Schedule A (Services) or Schedule S (Disaster Recovery Services), Equifax will be entitled, at its election, to terminate the portion of this Agreement (including Schedule S (Disaster Recovery Services)) relevant to the affected Service Tower within the affected country pursuant to Section 12.1(a) (without giving the notices and observing the cure periods set forth in Section 12.1(a)) upon written notice to IBM. If the Services for the Operations and/or Network

14

Service Towers are terminated, Equifax shall have the right to terminate the corresponding Disaster Recovery Services described in Schedules S and A on the same basis. If Equifax elects to terminate as described in this Section 3.3, Equifax shall give notice to IBM of such election within thirty (30) days after the occurrence of the event on which such termination is based. In the event termination of this Agreement is authorized under this Section 3.3, Equifax shall not be required to pay any Termination Charges or Wind-Down Expenses to IBM. Such termination shall not constitute the sole and exclusive remedy of Equifax for such failure of performance by IBM.

3.4 Audits

- (a) IBM shall maintain a complete audit trail of all financial and non-financial transactions resulting from this Agreement. IBM will assist the Equifax Group in meeting their respective audit and regulatory requirements, including providing access to the Facilities and IBM's and its Affiliates' books and records, to enable the Equifax Group and its auditors and examiners, to conduct appropriate audits and examinations of the Equifax Group's operations and IBM's and its Affiliates' operations relating to the performance of the Services, and to: (i) verify the accuracy of IBM's charges and credits to Equifax; (ii) verify the integrity of Equifax Group Data and examine the systems that process, store, support and transmit that data; (iii) verify that the Services are being provided in accordance with this Agreement; and (iv) as otherwise necessary to enable Equifax to meet, or to confirm that IBM is meeting, applicable regulatory and other legal requirements. Without limiting the generality of the preceding sentence, in conducting such audits and examinations, the Equifax Group and its auditors and examiners shall have the right to audit or examine relevant practices and procedures; systems, Machines and Software; and supporting information and calculations regarding compliance with Service Levels; general controls and security practices and procedures; disaster recovery and back-up procedures; provided, however, that neither Equifax nor its auditors will be allowed access to other IBM or IBM Affiliates customers' records or IBM confidential and proprietary data; but provided further that nothing in this Agreement shall limit or restrict Equifax's or IBM's rights in discovery proceedings pursuant to any civil litigation. Such access will require forty-eight (48) hour written notice to IBM and will be provided at reasonable hours. If any audit or examination reveals that IBM's invoices for the audited period are not correct (other than amounts in dispute pursuant Schedule C (Charges)), IBM shall promptly reimburse Equifax for the amount of any overcharges, or Equifax shall promptly pay IBM for the amount of any undercharges. If any such audit activities interfere with IBM's ability to perform the Services in accordance with the Service Levels, IBM shall be relieved of such performance obligations to the extent caused by such audit activity. If the assistance required of IBM shall cause IBM to expend resources and incur additional costs to provide such assistance that are not within the scope of the Services and Resource Unit Baselines, Equifax shall reimburse IBM for such costs.
- (b) If the exercise by the Equifax Group of its audit rights under Section 3.4(a) relates to functions performed by IBM's subcontractors, IBM shall supply or cause to be provided to the Equifax Group (or its auditors or examiners, as applicable) the necessary information and, as required, arrange for the Equifax Group (or its auditors or examiners, as applicable) to have suitable access to IBM's subcontractor's facilities (under IBM's accompaniment); provided, however, Equifax may request IBM to arrange such access only if part of the Services are performed at such subcontractor's facility.
- (c) Subject to Section 4.10, IBM agrees to make any changes to the Services and take other actions which are necessary in order to maintain compliance with laws or regulations applicable to its performance and provision of the Services. Subject to Section 4.10, Equifax may submit to IBM

15

findings and recommendations regarding changes to the Services necessary for the compliance by Equifax with applicable laws and regulations which IBM will analyze and consider in good faith. IBM shall promptly respond to Equifax regarding IBM's evaluation and activity plan for such findings and recommendations.

(d) IBM shall conduct audits of or pertaining to the Services in such manner and at such times as is consistent with the audit practices of well managed operations performing services similar to the Services. IBM shall support Equifax's conduct of SAS 70 (or similar or successor) audits to the extent the control objectives of

such audits relate to functions performed by IBM under this Agreement.

- (e) IBM shall make available promptly to Equifax the relevant portions of results of any * conducted by IBM or its Affiliates * or by inspectors or regulators, relating to the effectiveness of IBM's * to the extent such * is relevant to the Services and indicates an impact to Equifax.
- (f) Until the later of (i) three (3) years after expiration or termination of this Agreement; (ii) all pending matters relating to this Agreement (e.g., disputes) are closed; or (iii) the information is no longer required to meet Equifax's records retention policy as disclosed by Equifax to IBM in writing and as such policy may be adjusted from time to time, IBM shall maintain and provide access upon request to the records, documents and other information required to meet Equifax's audit rights under this Agreement.

3.5 IBM Cooperation with Authorized User Examinations

IBM acknowledges that Authorized Users frequently require the right to conduct security, disaster recovery and regulatory compliance examinations of Equifax information processing facilities and resources used to provide services to them or for their benefit. IBM agrees to cooperate with Equifax on behalf of Equifax's Authorized Users in the conduct of such examinations and to provide them access to IBM facilities, personnel and systems used to provide and perform the Services as reasonably necessary for such Authorized Users (or their designated representatives) to conduct such examinations. All such access to such IBM facilities and resources used by IBM to provide and perform the Services shall be subject to (i) reasonable data and records protection and physical security measures and (ii) such Authorized Users' employees, agents and representatives undertaking reasonable confidentiality requirements relating to such examinations.

3.6 Facilities

- (a) IBM will not relocate the portion of the Services provided from the Facilities set forth in Schedule H (Facilities) without the prior written consent of Equifax as described in Section 5.3(g).
- (b) During the Term, IBM will provide the Equifax Group with access upon prior notice to the portion of the Facilities used by IBM to provide and perform the Services (including, without limitation, the Data Center) in order for Equifax to provide tours of such portions of the Facilities and such tours will be conducted in a manner reasonably calculated not to interfere with IBM's provision of Services.
- * Information deleted pursuant to Rule 24b-2 of the Exchange Act.

16

- (c) IBM will provide reasonable access to the portion of the Facilities used by IBM to provide and perform the Services as necessary or appropriate for the performance, delivery and use of the Services by the Equifax Group and for the operation, maintenance, upgrade, support and use of any other Equifax hardware, software and other resources located in the Facilities (i) to the Equifax Group's authorized employees, agents and representatives, and (ii) to Third Party Providers and third party vendors and suppliers of installation, maintenance, support and upgrade services, technology and hardware for the System and any other Equifax hardware, software and other resources located in the Facilities serviced thereby. To the extent practical in light of such installation, maintenance, support and upgrade requirements, Equifax will provide twenty-four (24) hours notice to IBM prior to any visits by such Third Party Providers and third party vendors and suppliers.
- (d) All access to the portion of the Facilities under the control of IBM and used by IBM to provide and perform the Services (including, without limitation, the Data Center) shall be subject to (i) reasonable data and records protection and physical security measures (including Equifax physical security requirements) and (ii) such Equifax Group employees, agents and representatives and Third Party Providers and third party vendors and suppliers undertaking reasonable confidentiality requirements relating to such visits.

3.7 Security

IBM shall comply with Equifax's standard policies and procedures and with applicable leases as these are made available to IBM in writing regarding access to and use of the Equifax Data and Equifax Facilities, including procedures for the physical security of the Equifax Facilities. Equifax will authorize all access to all Software operated by, and Company Information and other records of the Equifax Group in the possession of, IBM in support of the Services through the data and records security procedures as described in Schedule T (Security Procedures). IBM shall notify Equifax of the identity of each of the entities and personnel working with IBM to provide and perform the Services that are to be authorized access to Equifax Data or the Software utilized in support of the Services and the level of security access required by each. The Parties shall cooperate in administering security procedures regarding such access, in accordance with such Schedule. IBM will enable such access by persons as designated by Equifax and deny such access to all other persons, in accordance with such Schedule.

3.8 Technology Refresh

IBM will refresh the information technologies components of the Services (including both hardware and software components) as specifically provided in this Agreement. This Section 3.8 shall not affect or limit IBM's obligations or authority to perform the repair, maintenance and upgrade functions and services as set forth in this Agreement.

3.9 Machines And Third Party Service Agreements

This Section 3.9 addresses the Parties' respective rights in Machines and Third Party Service Agreements. Grant by Equifax to IBM of rights of use pursuant to this Section 3.9 shall be deemed to include, subject to the other provisions of this Agreement, grant of such rights to IBM's Listed Subcontractors. Equifax Provided Hardware and Third Party Service Agreements made available to IBM are made available on an "as is, where is" basis, with no warranties whatsoever.

17

- (a) Leased Existing Equifax Provided Hardware. With respect to leased Equifax Provided Hardware so designated in Schedule E (Machines) and subject to Section 8.2, Equifax grants to IBM during the remaining term of the applicable lease solely to the extent necessary for performing the Services, the rights of use of such Equifax Provided Hardware that Equifax has with respect to such Equifax Provided Hardware. IBM shall comply with the duties imposed on Equifax by such leases as made known to IBM in writing.
- (b) Machine Acquisitions During the Term.

- (i) IBM is not acquiring any existing non-personnel assets of Equifax Group as of the Commencement Date.
- (ii) Subject to Section 3.9(a) and except as otherwise provided in the Financial Responsibilities Matrix set forth as Exhibit C-14 (Financial Responsibilities Matrix) of Schedule C (Charges), IBM shall acquire and shall be financially responsible for new Machines including modifications, upgrades, enhancements, additions and replacements of existing Machines in IBM's name as necessary or appropriate to provide the Services. Such Machines shall be acquired in the name of IBM except as set forth in Section 3.9(b)(iii) and subject to modifications, upgrades and enhancements of leased Machines being treated in accordance with the governing lease.
- (iii) With respect to Machine acquisitions for which Equifax is financially responsible as provided in the Financial Responsibilities Matrix set forth as Exhibit C-14 (Financial Responsibilities Matrix) of Schedule C (Charges), the Machines shall either be acquired by Equifax and made available to IBM for installation and operation or, alternatively, at Equifax's election, such Machines shall be purchased by IBM, at the applicable vendor's (third party or IBM) published list price at which such product may be purchased by the general public, plus a mark-up of five (5%) percent, in the name of Equifax unless Equifax expressly designates otherwise. With respect to Machines acquired in the name of Equifax pursuant to this Section 3.9(b)(iii), Equifax grants to IBM, during the Term, and (subject to Section 3.9(a) if such Machine is leased) solely to the extent necessary for performing the Services, the rights of use of such Machines that Equifax has with respect to such Machines.
 - (c) Third Party Service Agreements.
- (i) In the case of any Third Party Agreement listed in Schedule G (Third Party Agreements) for which IBM is designated as having 'Legal' responsibility, such agreement shall either be assigned to IBM or, alternatively, IBM shall assume legal responsibility for such agreement and have the same responsibility for managing such agreement as if it were an IBM subcontract. The determination as to which Third Party Service Agreements shall be shall be assigned to IBM will be made on a country-by-country basis. In either case, IBM shall comply with the duties imposed on Equifax by such Third Party Agreements.
- (ii) In the case of any other Third Party Agreements listed in Schedule G (e.e., those for which Equifax is designated as having 'Legal' responsibility), subject to Section 8.2, Equifax grants to IBM, during the Term, and solely to the extent necessary for performing the Services, the rights of use of the services covered by such Third Party Service Agreements that Equifax has with respect to such Third Party Service Agreements. IBM shall comply with the duties imposed on Equifax by such Third Party Service Agreements.
 - (d) Exercise of Rights. To the extent IBM has financial responsibility for a Machine lease or Third Party Service Agreement, but such lease or Third Party Service Agreement remains in Equifax's name, Equifax shall exercise termination or extension rights thereunder as IBM, after consultation with Equifax, reasonably directs; provided that IBM shall be responsible for the costs, charges and

18

fees to be paid to such Third Party associated with the exercise of such rights to the extent Equifax has acted in accordance with IBM's direction.

3.10 Equifax Owned Software — Existing

Equifax grants to IBM, and to IBM's Affiliates and Listed Subcontractors as required for IBM to provide the Services, a worldwide, fully paid-up, nonexclusive license during the Term to Use Equifax Owned Software solely for the purpose of and to the extent necessary for performing the Services. Equifax Owned Software will be made available to IBM in such form and on such media as exists on the Commencement Date or as is later obtained by Equifax, together with available documentation and any other related materials. IBM shall not be permitted to Use Equifax Owned Software for the benefit of any entities other than members of the Equifax Group and their Authorized Users without the prior written consent of Equifax, which may be withheld at Equifax's discretion. IBM shall install, operate and support (and otherwise treat in the same manner as Equifax Owned Software existing as of the Commencement Date) additional Equifax Owned Software that Equifax may make available to IBM from time to time during the Term. Except as otherwise requested or approved by Equifax, IBM shall cease all Use of Equifax Owned Software upon expiration or termination of this Agreement. Equifax Owned Software is made available to IBM on an "as is, where is" basis, with no warranties whatsoever, other than the intellectual property infringement indemnification obligations described in this Agreement.

3.11 Third Party Provider Software — Existing

- (a) <u>Grant of Rights.</u> With respect to the Third Party Provider Software licensed by Equifax, subject to the Parties having obtained any Required Consents for Third Party Provider Software in the manner provided in Section 8.2, Equifax grants to IBM, and to IBM's Affiliates and Listed Subcontractors as required for IBM to provide the Services, solely for the purposes of and to the extent necessary for performing the Services, the rights of Use of such Software that Equifax has as of the Commencement Date or later obtains with respect to such Software. Except as otherwise requested or approved by Equifax, IBM shall cease all Use of such Software upon expiration or termination of this Agreement. At Equifax's election, IBM shall promptly return to Equifax or destroy any such Software and related documentation.
- (b) IBM will comply with all license obligations under all licenses and maintenance agreements for the Software, including without limitation, the obligations of nondisclosure and scope of use; provided, however, that IBM will only be obligated under this Section 3.11(b) with regard to the licenses and maintenance agreements for Equifax Software to the extent the obligations thereunder are disclosed to and accepted by IBM. To the extent provided to IBM by Equifax prior to execution of this Agreement, IBM shall be deemed to have reviewed and accepted the obligations under the licenses and maintenance agreements for the Equifax Software listed on Schedule F (Software) as of the Commencement Date.

3.12 New Software Added During the Term

(a) IBM shall not introduce any additional IBM Software into Equifax's dedicated IT Environment in providing the Services without Equifax's prior written approval, which approval Equifax may withhold in its discretion. As and to the extent necessary for Equifax or a third party to perform work as permitted under this Agreement, IBM grants to Equifax, Authorized Users or such third party, after being notified by Equifax, a non-exclusive license to Use such Software solely to enable Equifax Group and its Authorized Users to receive and use the Services during the Term.

19

(b) All IBM Software provided by IBM in connection with the Services and any Equifax Software licensed under a Third Party Agreement shall be licensed (and the attendant maintenance arrangements contracted) in the name of the Equifax Group member designated by Equifax as the licensee with IBM having the right to access and use such Software in performing the Services, unless IBM can procure such Software (and/or attendant maintenance arrangement) on a more cost effective basis licensed in its own name on terms permitting IBM to assign the license and maintenance agreements for such Software to Equifax without charge to Equifax at the expiration or termination of this Agreement or on terms otherwise accepted by Equifax in writing, in which case IBM may procure such Software (and/or attendant maintenance arrangement) in IBM's name.

3.13 Changes to the Software

- (a) IBM shall not, and shall not have the right to, direct the Equifax Group to, terminate, extend, replace, amend or add licenses for the Software and/or the maintenance arrangements attendant therewith, contracted in the name of a member of the Equifax Group without notifying Equifax in writing of the proposed action by IBM and obtaining Equifax's prior written agreement; moreover, IBM shall provide to Equifax a written report of the reasons for, and the impact and ramifications on the Services of, such proposed action concurrently with such notification. IBM may terminate, replace, amend or add licensees for the IBM Software as it chooses so long as IBM continues to perform the Services in the manner required by this Agreement; provided, however, IBM agrees to provide twenty-one (21) business days written notification to Equifax prior to each such termination, replacement, amendment or addition and concurrently with such notification, deliver to Equifax a written report of the reasons for, and the impact and ramifications on the Services of, IBM's proposed action. In addition, if such action by IBM with respect to a license and/or maintenance arrangement for the IBM Software will have an impact on the Services or the monitoring and/or evaluation of the Services in a manner that in turn will have a financial and/or operational impact on the Equifax Group or the ability of IBM or Equifax to monitor and/or evaluate the performance and delivery of the Services, and IBM is notified in writing by Equifax of its estimate of such financial and/or operational impact prior to IBM's implementation of such action and IBM elects to proceed, IBM will provide or cause to be provided the programs, services, rights and other benefits and resources that are the subject of such licenses and maintenance agreements to the Equifax Group on terms no less favorable than the terms of such license and maintenance agreements and ensure that there shall be no negative impact on the ability of IBM or Equifax to monitor and/or evaluate the performance and delivery of the Services. If Equifax in connection with or resulting from IBM's termination, replacement, amendment or addition of any license for IBM Software and/or maintenance arrangement incurs additional expenses, costs or Losses, including but not limited to personnel costs, and IBM has been notified in writing by Equifax of its estimate of such financial impact prior to IBM's implementation of such action and IBM elects to proceed, IBM shall promptly reimburse Equifax for such amounts actually incurred by Equifax; provided, however, that in each instance in this Section 3.13(a)) that Equifax provides IBM an estimate of the financial impact of an action by IBM on Equifax, the amounts recoverable from IBM by Equifax in each such instance shall not exceed the amount of the written estimate provided to IBM for each such instance.
- (b) IBM will provide to Equifax, and update as changes occur, a listing of all Software by name, Maintenance Release and Version promoted into production on each Machine at each location of the Machines.
- (c) If Equifax requests a substitution of any Software for which IBM has financial responsibility, according to Exhibit C-14 (Financial Responsibilities Matrix) of Schedule C (Charges), Equifax

20

shall pay or receive a credit in the amount by which the periodic license or maintenance fees attributable to the substituted Software exceeds or is less than the then-current periodic license or maintenance fees being paid by IBM attributable to the Software being replaced. If Equifax requests deletion of any Software for which IBM has financial responsibility and does not immediately substitute any other new Software therefor, Equifax may utilize an amount equal to the then-current applicable periodic license and/or maintenance fees attributable to such deleted Software to offset the fees attributable to any new Software or receive a credit in such amount. IBM will provide Equifax with the requisite license and/or maintenance fees support documentation to assist Equifax in evaluating the decision to replace such Software. Equifax will be responsible for any other fees payable to the Software vendor associated with such substitutions or additions.

- (d) Equifax may add Software to, or delete Software. IBM agrees to promote into or remove from production, use and operate any Software selected by Equifax, subject to the provisions of Section 10.1 (Prices and Charges for New Services) of Schedule (C) if performance of such obligations requires IBM to perform New Services. Equifax shall be permitted by IBM to audit, control and approve all new Software prior to its promotion into production, and IBM shall provide the cooperation, information and access necessary or appropriate to permit Equifax to perform such functions.
- (e) If IBM timely notifies Equifax that any software requested by Equifax be substituted for, deleted from, or added to, the Software will have an adverse impact on the operation of the System before such action is effected and Equifax directs IBM to effect such action even in view of such notice, IBM shall be excused from of any failure to satisfy the Service Levels for the affected portion of the Services to the extent, and only to the extent, such action directly causes such failure to satisfy such Services Levels.

3.14 Terms of Acquisition by IBM of Third Party Provider Software

If IBM acquires Third Party Provider Software in IBM's name during the Term to provide the Services, IBM shall use Commercially Reasonable Efforts to acquire such software on commercially reasonable terms and conditions, including customary forms of protection of the licensee concerning actual or alleged claims of infringement of intellectual property rights.

3.15 Affiliates

If the Equifax Group acquires any additional Affiliates or other operations or assets during the Term and desires that IBM provide the Services for such Affiliates or other operations or assets, IBM will provide such Affiliates or other operations or assets with Services in accordance with this Agreement, subject to additional charges in accordance with Schedule C (Charges), including Section 7.3 (New Services) of Schedule C (Charges) if and to the extent acceptance of such responsibilities by IBM would require the performance of New Services.

3.16 Viruses

IBM shall take commercially reasonable measures to ensure that no Viruses or similar items are coded or introduced into the System and the operating environments used to provide the Services, initially continuing to perform the Virus protection and correction procedures and processes in place at the Equifax Group prior to the Execution Date, and subsequently continues to review, analyze and implement improvements to and upgrades of such virus prevention and correction programs and processes that are commercially reasonable and consistent with industry standards. If a Virus is found to have been introduced into the System and the operating

Virus.

4. WARRANTIES/REPRESENTATIONS/COVENANTS

4.1 Work Standards

IBM warrants, represents and covenants that (a) it has, and during the Term will have, and each of the IBM employees and subcontractors that it will use to provide and perform the Services has and during the Term will have, the necessary knowledge, skills, experience, qualifications and resources to provide and perform the Services in accordance with this Agreement; (b) it has successfully provided and performed the Services or services that are substantially similar to the Services for other customers of IBM; and (c) the Services will be performed for the Equifax Group in a diligent, workmanlike manner in accordance with industry standards applicable to the performance of such services.

4.2 Non-infringement

Each of the Parties warrants and covenants that it will perform its responsibilities under this Agreement in a manner that does not infringe, or constitute an infringement or misappropriation of, any patent, trade secret, copyright or other proprietary right of any third party; provided however that IBM shall not be in breach of this warranty and covenant * in the course of performing the Services is caused by *. Notwithstanding this provision or any other provision in this Agreement, Equifax makes no warranty or representation with respect to any claims for such infringement or misappropriation by virtue of its compliance with obligations herein to provide IBM access to, use of or benefits of any Third Party Agreements prior to receiving the necessary Required Consents; provided, however, that this Section 4.2 shall not relieve Equifax from any liability or obligation under Sections 8.2 and 14.2.

4.3 Disabling Code

IBM represents and warrants that, without the prior written consent of Equifax, IBM will not invoke any Disabling Code that may be part of the Software at any time, including upon expiration or termination of this Agreement for any reason.

4.4 Authorization and Enforceability

Each Party hereby represents and warrants that:

- (a) it has all requisite corporate power and authority to enter, and fully perform pursuant to, into this Agreement;
- * Information deleted pursuant to Rule 24b-2 of the Exchange Act.

22

- (b) the execution, delivery and performance of this Agreement and the consummation of the transactions contemplated hereby have been duly and properly authorized by all requisite corporate action on its part; and
- (c) this Agreement has been duly executed and delivered by such Party.

4.5 Maintenance

IBM warrants and covenants that it shall maintain the Machines and Software for which IBM is responsible so that they operate in accordance with their specifications.

4.6 Efficiency and Cost Effectiveness

IBM warrants and covenants that it shall take commercially reasonable actions (a) to efficiently administer, manage, operate and use the resources employed by IBM to provide and perform the Services that are chargeable to Equifax under this Agreement (b) to diligently and continuously improve the performance and delivery of the Services by IBM and the elements of the policies, processes, procedures and System that are used by IBM to perform and deliver the Services, including, without limitation, re-engineering, tuning, optimizing, balancing or reconfiguring the processes, procedures and systems used to perform, deliver and track the Services; and (c) use Commercially Reasonable Efforts to perform the Services in a cost-effective manner consistent with the required level of quality and performance.

4.7 Software Ownership or Use

IBM warrants that it is either the owner of, or authorized to Use, the Software utilized pursuant to Sections 3.12(a) any Software developed by IBM as part of the Services pursuant to Section 10.1.

4.8 Inducements

IBM represents and warrants that it has not violated any applicable laws or regulations or any Equifax policies of which IBM has been given notice regarding the offering of unlawful inducements in connection with this Agreement.

4.9 Disclaimer

- (a) IBM does not warrant the accuracy of any advice, report, data or other product delivered to Equifax to the extent any inaccuracies are caused by data and/or software provided by Equifax. IBM will promptly notify Equifax of any such inaccuracies of which IBM becomes aware and the cause therefore if known by IBM. IBM will provide commercially reasonable assistance to Equifax to remedy such problems.
- (b) Subject to the obligations of IBM to satisfy the Service Levels and provide the Services as set forth in this Agreement without material denigration or interruption, IBM does not assure uninterrupted or error-free operations of the Software and Machines.
- (c) EMU Transition
- (i) IBM is not providing Equifax with any EMU assessment, conversion or testing services under this Agreement. IBM will not be responsible for the assessment of Equifax's current systems or for taking any appropriate actions to migrate them to Euro-Ready systems. However, nothing in this Section 4.9(c)(i) shall relieve IBM of its obligations under Section 4.9(c)(ii).

(ii) The Parties shall have the duties, obligations and responsibilities with respect to EMU Matters as set forth in the following chart for "Equifax Products" (comprised of the categories of assets listed in the chart below); "IBM Logo Products" provided by IBM under this Agreement and used to provide the Services; "IBM Third Party Products" (defined to mean third party products owned or licensed by IBM and provided by IBM under this Agreement, in both cases used to provide the Services); and "Equifax Provided Assets" (defined to mean all hardware, software contracts and licenses with respect to which ownership or title is or was transferred by Equifax to any of its Affiliates to IBM or any of its Affiliates prior to the Commencement Date). The Schedules referenced in this Section shall be updated pursuant to Section 8.1.

	Maintenance	Financial	Management	SLA Relief	
	Note 1	Note 2	Note 3	Note 4	
(i) Equifax Products					
"Applications Software — Equifax" (Schedule F)	E	E	Е	Yes	
"Equifax Provided Hardware" Used by IBM	I	E	I	Yes	
Equifax hardware Not Used by IBM	E	E	Е	Yes	
"Contracts" Used by IBM (Schedule G)	N/A	E	E	Yes	
Equifax agreements Not Used by IBM	N/A	E	E	Yes	
"Systems Software — Equifax" Used by IBM (Schedule F)	I	E	E	Yes	
Equifax software Not Used by IBM	E	E	E	Yes	
(ii) IBM Logo Products provided by IBM under this					
Agreement and used to provide the Services	I	I	I	No	

E = Equifax Responsibility

I = IBM Responsibility

- (d) With respect to EMU Matters only, the following definitions will apply:
- (i) Note 1 "Maintenance" shall mean maintenance responsibility, including but not limited to applying fixes, corrections and other enhancements (but not financial responsibility for such) and/or using reasonable efforts to cause a third party vendor to perform the foregoing.
- (ii) Note 2 "Financial" shall mean the financial responsibility for all fees, charges and costs incurred in connection with correcting an EMU Matter, including but not limited to software upgrades, new versions or releases, hardware upgrades or replacements and IBM Project Office support.

24

- (iii) Note 3 "Management" shall mean responsibility for coordinating EMU Matter corrective actions (but not financial responsibility for such actions).
- (iv) Note 4 "Service Level Relief" shall mean that IBM then IBM shall not be held responsible for any failure to meet a Service Level if, and only to the extent, such failure is directly attributable to a failure of the items covered by the description in the chart above to be Euro-Ready.
- (v) The IBM Logo Products which are introduced after January 1, 1999 and that are used directly by IBM (and not by IBM subcontractors) and that have a dependency to process monetary data in the euro denomination to provide the Services will be IBM Euro-Ready.
- (vi) For the IBM Third Party Provider Products, to the extent possible, IBM will seek to determine from the third party manufacturers or providers whether such third party products which have a Euro dependency and are designated by such third party manufacturers or providers as Euro-Ready. To the extent that IBM is permitted by law or contract to do so, IBM will pass through any such representations or warranties for IBM Third Party Products to Equifax. Where IBM has obtained information that an IBM Third Party Product is designated as not Euro-Ready, IBM will notify Equifax as soon as reasonably practicable after IBM becomes aware of such fact. Nothing herein shall, however, be construed as a representation or warranty by IBM that IBM Third Party Products are Euro-Ready.
- (vii) Notwithstanding that the IBM Third Party Products are considered to be Euro-Ready, IBM does not represent or warrant uninterrupted or error free operation of the Services.
- (viii) For the IBM Logo Products, IBM will determine whether such IBM Logo Products which have a Euro dependency, are designated as IBM Euro-Ready. Where IBM has determined that an IBM Logo Product is designated as not IBM Euro-Ready, IBM will notify Equifax as soon as reasonably practicable after IBM becomes aware of such fact.
- (ix) Notwithstanding that the IBM Logo Products are considered to be IBM Euro-Ready, IBM does not represent or warrant uninterrupted or error free operation of the Services.

MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE.

4.10 Legal and Regulatory Compliance

Each Party agrees at its cost and expense to obtain all necessary regulatory approvals applicable to its business, to obtain any necessary permits for its business, and to comply with all laws and regulatory requirements applicable to the performance of its obligations under this Agreement. To the extent Equifax directs IBM, in accordance with the Regulatory Compliance Interaction Model set forth as Exhibit A-4-1 to Schedule A (Services), to comply with Equifax corporate compliance policies that describe the processes and procedures to be followed in order for Equifax to be and remain in compliance with legal and regulatory requirements (including consent decrees to which Equifax is a party) that apply to Equifax's business and operations and provides IBM a written copy of such corporate compliance policies, IBM shall comply with such corporate compliance policies as and to the extent applicable to functions for which IBM is responsible; provided, however, that IBM's compliance with such requirements may constitute a New Service.

25

4.11 Year 2000 Warranty

IBM warrants that products manufactured or distributed by IBM and bearing a logo of IBM and/or an IBM Affiliate ("IBM Logo Products") that are provided under this Agreement and installed after the Execution Date of a Transaction Document and used to provide Services under this Agreement will be IBM Year 2000 Compliant, unless IBM notifies Equifax of its intention to install a non-IBM Year 2000 Compliant IBM Logo Product and Equifax agrees in writing to such installation.

4.12 Covenant of Cooperation and Good Faith

The Parties covenant to timely and diligently cooperate, with due consideration of the goals, objectives and purposes of this Agreement, to facilitate the performance of their respective duties and obligations under this Agreement in a commercially reasonable manner. Further, the Parties agree to deal and negotiate with each other and their respective Affiliates in good faith in the execution and implementation of their duties and obligations under this Agreement.

5. TRANSFER, TRANSITION AND TRANSFORMATION

5.1 Transition/Transformation Plans

- (a) IBM and Equifax have developed and agreed upon the "Transition/Transformation Plan" as set forth in Schedule I (Transition/Transformation) and an "IT Management Process Improvement Program" as set forth in Schedule J (IT Management Process Improvement Program). With respect to the transition and transformation, IBM will:
- (i) perform the transition and transformation tasks as specified in Schedule I (Transition/Transformation) and Schedule J (IT Management Process Improvement Program), respectively; and
- (ii) maintain the Services with minimal disruption to Equifax's business operations in each country;
 - (b) No functionality of the information technology services or operations being transitioned or transformed shall be disabled or cut over to a new service or replacement functionality until the new service or functionality is demonstrated to Equifax's satisfaction to have equivalent capabilities for such functionality and Equifax has provided written notice of acceptance of such capability; provided, however, that IBM shall not be required to keep such dual services or operations enabled for more than thirty (30) days (or, in specific circumstances, if any, to be mutually agreed by the Parties in which it is not reasonably possible to test and verify the proper functioning of the new service or functionality within thirty (30) days, such longer period as is reasonable under the circumstances) at IBM's expense unless Equifax has identified and notified IBM within such period of reasonable concerns, in which case the Parties will agree to a reasonable timeframe for continuation of the identified services or operations at IBM's expense.
 - (c) Equifax reserves the right to monitor, test and otherwise participate in the transition and transformation. IBM shall immediately notify Equifax if such monitoring, testing or participation has caused (or in IBM's reasonable opinion may cause) a problem or delay in the transition and transformation and work with Equifax to prevent or circumvent such problem or delay.
 - (d) Equifax reserves the right to temporarily suspend the transition and/or the transformation by providing notice to IBM if IBM is responsible for the transition and/or transformation failing to fulfill the requirements set out in the approved Transition/Transformation Plan or otherwise causing a material disruption in Equifax's business environment occasioned by the carrying out of the transition and/or transformation, until such time as IBM can demonstrate to Equifax's satisfaction that it is ready to achieve such requirements and/or end such disruptions.

26

(e) During the transition period, Equifax will cooperate with IBM in implementing the Transformation/Transformation Plan by providing the personnel (or portions of the time of the personnel) set forth in the Transformation/Transformation Plan ("Transition Personnel") and performing the tasks described for Equifax in the Transition/Transformation Plan; provided, however, that unless otherwise agreed in writing by the Parties, Equifax will have a modest role and be responsible for incurring modest costs in performing such tasks. During the Transition Period, IBM will be responsible for the provision of the Services set forth in this Agreement (including within those Services the implementation of the Transformation/Transformation Plan).

5.2 *

(a) In order to facilitate the orderly assumption by IBM and its Affiliates of incremental scope of responsibility under this Agreement, * in accordance with the terms and conditions of * and applicable local law. All costs and expenses incurred by IBM in connection with * shall be the responsibility of IBM. Except as * expressly provides otherwise, IBM will promptly reimburse Equifax for the * on and after the Commencement Date for the period until *.

5.3 Resources and Facilities

(a) To enable IBM to provide the Services, the Parties may agree for Equifax to provide, at no charge to IBM, the use of the Equifax Provided Hardware, Equifax Provided Office Furnishings, Equifax facilities, and office services such as reasonable local analog telephone services for the sole purpose of providing and performing the Services for the Equifax Group. These obligations will generally not include the provision of (i) office, storage or equipment/Data Center space, parking facilities, or heat, light, power, air conditioning and other similar utilities which will be provided under a separate lease agreement between the members

of the Equifax Group as lessor and IBM or its Affiliates as lessee for a portion of any Equifax Group facilities leased to IBM (or its Affiliates), or (ii) office support services (e.g., janitorial and security), office supplies and similar services and consumables, unless already being provided to IBM by Equifax as of the Execution Date. All such items provided by Equifax shall comply with all applicable laws and regulations relating to safety and use. Subject to the satisfaction of Equifax's obligation with respect to compliance with applicable laws and regulations, IBM shall ensure a safe working environment is maintained for the Equifax Provided Hardware, Equifax Provided Office Furnishings and Equifax facilities in compliance with all applicable laws and regulations, and shall take no action that will compromise such safety of such working environment or violate such laws and regulations.

(b) Equifax agrees to negotiate in good faith with IBM with the objective of IBM and Equifax entering into an agreement within six months to extend, through March 1, 2012, the term of the existing sub-lease agreement between them pursuant to which IBM sub-leases from Equifax space located at 1505 and 1525 Windward Concourse, Alpharetta, Georgia. The Parties agree that the extension of the sub-lease would be on its existing terms for calculating the rent and other amounts due under the sub-lease and that IBM shall have the right to perform outsourcing services for other customers utilizing such space. The sub-lease extension will also give IBM the option to further extend the term of the sub-lease to be coterminous with the term of Equifax's facilities lease that includes the sub-leased space in the event Equifax exercises any or all of its current options to extend such lease at the end of its term — which options give Equifax the right to extend the term of its lease for up to six (6) periods of three (3) months each. If Equifax does not exercise its rights for such lease extension or otherwise provide for IBM to utilize the same space, Equifax shall be responsible for all costs incurred by IBM to move elements necessary for IBM to continue to perform the Services.

* Information deleted pursuant to Rule 24b-2 of the Exchange Act.

27

- (c) If the Parties fail to reach agreement on the terms of the sub-lease extension, the sub-leased space at 1505 and 1525 Windward Concourse will become an Equifax facility upon the expiration of the sub-lease, the continuous use of which Equifax will make available to IBM during the Term, and IBM will credit Equifax with an amount equal to the rent and services amounts specified in the former sub-lease for so long as IBM continues to provide the Services from 1505 and 1525 Windward Concourse.
- (d) IBM shall notify Equifax whenever any Equifax Provided Hardware, Equifax Provided Software and/or Equifax Provided Office Furnishings are no longer required by IBM to perform the Services. Equifax's obligations set forth in this Section with respect to each such item of resources shall terminate thereafter.
- (e) Except as otherwise provided in this Agreement, IBM will have the responsibility and obligation to provide and administer, manage, support, maintain and pay for all resources (including, without limitation, personnel, hardware, software, facilities, services and other items, however described) necessary or appropriate for IBM to provide, perform and deliver the Services as described in this Agreement. When the Equifax resources are no longer required for performance of the Services, IBM shall return them to Equifax in substantially the same condition as when IBM began use of them, subject to reasonable wear and tear.
- (f) IBM will provide and have on site its Global Project Executive prior to the Commencement Date and for the duration of the Term, and will timely provide additional trained and qualified personnel as necessary or appropriate to facilitate and ensure the timely and proper definition, provision, performance and delivery of the Services in accordance with this Agreement.
- (g) IBM will have the right to change the location of the IBM activities associated with the Services with the prior written consent of Equifax (which consent shall not be unreasonably withheld) or upon the occurrence of a Force Majeure Event (in which event IBM shall, if possible, provide the Services from the disaster recovery facility designated for that purpose). Equifax has agreed that IBM may change the location of the Network Operations Center (NOC) from Equifax's facility in Alpharetta, Georgia to an IBM facility located in the U.S. Among the factors Equifax may consider in determining whether to grant any such consent, Equifax may consider whether any and all changes in the location of such IBM activities may result (i) in a reduction of IBM's ability to perform the Services and the Business and Operations Support Plan; (ii) in any reduced accessibility to IBM and/or the Services by the Equifax Group; (iii) in any deterioration of the Services; (iv) any decrease in the security or integrity of operations and Company Information of the Equifax Group; and (v) in any additional cost to Equifax.

6. GOVERNANCE

6.1 Relationship and Contract Governance Model

- (a) IBM acknowledges that it is a key business requirement of Equifax's that IBM provide the Services in a consistent, integrated manner globally across all Service Towers and Country Locations. To meet that requirement, IBM has proposed and will adhere to a global relationship and governance model and processes as described in Schedule L (Governance).
- (b) The IBM organization responsible for IBM's relationship with Equifax and delivery of the Services will be led by an IBM Global Project Executive (Global PE), whose counterpart will be the Equifax Global Program Manager (Global PM).

28

- (c) The IBM Global PE will be supported by IBM Regional Project Executives (Regional PEs), one in each Country Location, and Delivery Project Executives (DPEs), whose roles and responsibilities are set forth in Schedule L (Governance). The IBM Regional PE's will serve as the primary points of contact with their respective Equifax counterparts in their home Country Locations, the Equifax Regional Program Managers (Regional PMs).
- (d) For each Project undertaken under this Agreement, IBM will also designate in the applicable Statement of Work a Project manager who will be assigned full-time or part-time as specified in that Statement of Work and who will have responsibility, working under the direction of the applicable IBM Regional PE, for the successful completion and delivery of the Project.

6.2 Meetings

The Parties shall determine an appropriate set of meetings to be held between their representatives, which shall include at least a quarterly meeting of the IBM Global PE with the Equifax Global PM and at least a monthly meeting of each IBM Regional PE with his or her counterpart Equifax Regional PM. IBM shall prepare and circulate an agenda sufficiently in advance of each such meeting to give participants an opportunity to prepare for the meeting. IBM will make such changes to the agenda as Equifax may request. Equifax will chair all such meetings. At Equifax's request, IBM shall prepare and circulate minutes promptly after each meeting. Such minutes shall not be binding on either Party if they are in any way inconsistent with this Agreement.

6.3 Procedures Manual.

- (a) The "Procedures Manual" shall describe how IBM shall perform and deliver the Services under this Agreement, the Machines and Software being used, and the documentation (e.g., operations manuals, user guides, specifications) which provide further details of such activities. The Procedures Manual shall describe the activities IBM proposes to undertake in order to provide the Services, including those direction, supervision, monitoring, staffing, reporting, planning and oversight activities normally undertaken to provide services of the type IBM is to provide under this Agreement. The Procedures Manual also shall include descriptions of the acceptance testing and quality assurance procedures approved by Equifax, IBM's problem management and escalation procedures, and the other standards and procedures of IBM pertinent to Equifax's interaction with IBM in obtaining the Services. The Procedures Manual shall be suitable for use by Equifax to understand the Services.
- (b) Within ninety (90) days after the Execution Date, IBM shall deliver a draft Procedures Manual to Equifax, for Equifax's comments and review. IBM shall incorporate reasonable comments or suggestions of Equifax and shall finalize the Procedures Manual within one hundred and twenty (120) days after the Commencement Date. The final Procedures Manual shall be subject to the approval of Equifax. IBM shall periodically update the Procedures Manual to reflect changes in the operations or procedures described therein. Updates of the Procedures Manual shall be provided to Equifax for review, comment and approval. IBM shall perform the Services in accordance with the most recent Equifax-approved version of the Procedures Manual. In the event of a conflict between the provisions of this Agreement and the Procedures Manual, the provisions of this Agreement shall control. The Procedures Manual shall be considered an operational document, which the IBM Global Project Executive and the Equifax Global Program Manager may revise by mutual written agreement without the need to amend this Agreement.

29

6.4 Change Management Process

- (a) The Parties will follow an agreed Change Management process to control Changes to the IT Environment in a controlled manner with minimum disruption. "Change," as defined in Exhibit A-2, means the addition, modification or removal of any aspect of the IT Environment. The purposes and objectives of the Change Management process are (i) to determine whether a Change to the IT Environment is within the scope of the Services or constitutes a New Service, (ii) to prioritize all requests for Changes, (iii) to minimize the risk of exceeding time and/or cost estimates associated with the Change by identifying, documenting, quantifying, controlling, managing and communicating Change requests, their disposition and, as applicable, implementation; and (iv) to identify the different roles, responsibilities and actions that shall be assumed and taken by the Parties to define and implement the Changes. The Change Management process covers activities from receipt of a request for a Change to assessment, scheduling, implementation and, finally, review. The Change Management process will produce approval (or otherwise) for any proposed Change. Equifax will not be obliged to approve any Change requested by IBM if implementation of the change would increase IBM's charges to Equifax under this Agreement or Equifax's internal costs, or would otherwise adversely affect Equifax's business. IBM shall not be obliged to carry out any Change that is not approved by Equifax. The Change Management process will be included as part of the Procedures Manual.
- (b) If an approved Change would result in a change in the scope of the Services, IBM's charges under this Agreement or terms and conditions, then such Change must be authorized via a Contract Change made pursuant to Section 17.2.
- (c) The Change Management process shall be considered an operational document, which the IBM Global Project Executive and the Equifax Global Program Manager may revise by mutual written agreement without the need to amend this Agreement.

7. IBM Personnel

7.1 Global Project Executive

Unless otherwise provided in this Agreement, IBM shall cause the person assigned as the IBM Global Project Executive to devote substantially all of his or her working time and effort in the employ of IBM to his or her responsibilities for the provision of the Services, subject to IBM's reasonable holiday, vacation and medical leave policies and subject to occasional, short-term, non-recurring work on other assignments by IBM related to the Project Executive's areas of expertise. The IBM Global Project Executive shall (i) serve as the single point of accountability for IBM for the Services; (ii) have day-to-day authority for undertaking to ensure customer satisfaction; (iii) work directly with Equifax's Chief Technology Officer and other information technology executives; (iv) develop and maintain a keen understanding of Equifax's business requirements and applications; and (v) serve as Equifax's technology advisor as well as an advocate for Equifax's interests within IBM. The IBM Global Project Executive's compensation shall include meaningful financial incentives based on Equifax's satisfaction with the Services.

7.2 Replacement of Personnel

(a) IBM shall assign an adequate number of IBM personnel to perform the Services. IBM personnel shall be properly educated, trained and fully qualified for the Services they are to perform.

30

(b) If Equifax reasonably and in good faith determines that it is not in Equifax's best interests for any IBM or subcontractor employee to be appointed to perform or to continue performing any of the Services, Equifax shall give IBM written notice specifying the reasons for its position and requesting that such employee not be appointed or be removed from the IBM or IBM subcontractor employee group servicing Equifax and be replaced with another IBM employee or IBM subcontractor employee. Promptly after its receipt of such a notice, IBM shall investigate the matters set forth in the notice, discuss with Equifax the results of the investigation, and resolve the matter in a mutually agreeable manner. If, following such period, Equifax requests the replacement of such person, IBM shall replace that person with another person of suitable ability and qualifications.

7.3 Key IBM Personnel Positions

- (a) The following positions shall be considered to be the initial Key IBM Personnel Positions:
 - (i) The Global Project Executive;
 - (ii) Regional Project Executives; and
 - (iii) Delivery Project Executives.

- (b) IBM shall cause the IBM Personnel who occupy each of the Key IBM Personnel Positions to devote substantially full time and effort to the provision of the Services, except that Equifax agrees that the individual appointed as the IBM Global Project Executive as of the Commencement Date may continue to serve in that role for both Equifax and the company he serves as of the Commencement Date that was formerly owned by Equifax.
- (c) Equifax may from time to time change the designated Key IBM Personnel Positions under this Agreement, provided that without IBM's consent, the number of Key IBM Personnel pPpositions shall not exceed the number specified above as of the Commencement Date.
- (d) Before the initial and each subsequent assignment of an individual to a Key IBM Personnel Position, IBM shall notify Equifax of the proposed assignment, introduce the individual to appropriate representatives of Equifax and, consistent with IBM's personnel practices, provide Equifax with a resume and any other information about a prospective individual reasonably requested by Equifax. If Equifax in good faith objects to the proposed assignment, the Parties shall attempt to resolve Equifax's concerns on a mutually agreeable basis. If the Parties have not been able to resolve Equifax's concerns within five (5) working days, IBM shall not assign the individual to that position and shall propose to Equifax the assignment of another individual of suitable ability and qualifications.
- (e) IBM will give Equifax, where reasonably possible, at least ninety (90) days advance notice of a change of the person appointed to a Key IBM Personnel Position, and will discuss with Equifax any objections Equifax may have to such change. Where reasonably possible, IBM will arrange for the proposed replacement for an individual appointed to a Key IBM Personnel Position to work side-by-side with the individual being replaced during the notice period to effectuate an effective transfer of knowledge prior to the incumbent leaving his or her position. IBM shall not reassign or replace any person assigned to a Key IBM Personnel Position during the first year of his or her assignment to the Equifax service team, nor shall IBM assign more than five (5) different individuals to a single Key IBM Personnel Position during the Term, unless Equifax consents to such reassignment or replacement, or the IBM employee voluntarily resigns from IBM, requests a transfer, is terminated by IBM or is unable to work due to his or her death or disability. Individuals

31

filling Key IBM Personnel Positions may not be transferred or re-assigned until a suitable replacement has been approved by Equifax, and no such re-assignment or transfer shall occur at a time or in a manner that would have an adverse impact on delivery of the Services. IBM shall establish and maintain an up-to-date succession plan for the individuals serving in Key IBM Personnel Positions.

7.4 Retention of Experienced Personnel

- (a) Equifax has identified certain of the Transferred Employees as "Knowledge Retention Personnel" due to their possession of knowledge that Equifax believes will be critical to IBM in providing the Services. A list of the Transferred Employees who are designated as Knowledge Retention Personnel is set forth as Exhibit D-5 (Knowledge Retention Personnel) to Schedule D (Human Resources). IBM will not transfer or reassign any Knowledge Retention Personnel from his or her current function prior to the successful completion of the transition and transformation projects set forth in Schedule I (Transition/Transformation) in the applicable Country Locations that relate to his or her current function. Thereafter, IBM will use commercially reasonable efforts to keep all Knowledge Retention Personnel as members of the Equifax service team subject to IBM's ability to make reasonable opportunities for promotion available to them within the Equifax service team. The provisions of Section 7.3(e) shall also apply to changes in the assignment and/or employment status of Knowledge Retention Personnel, except that the provision of Section 7.3(e) that limits the number of individuals who may occupy a Key Personnel Position during the Term shall not apply to Knowledge Retention Personnel.
- (b) Equifax and IBM agree that it is in their best interests to keep the turnover rate of IBM personnel to a reasonably low level. Accordingly, if IBM fails to meet the Service Levels persistently or continuously and if Equifax reasonably believes such failure is attributable in whole or in part to IBM's reassignment, movement, or other changes in the human resources allocated by IBM to the performance and delivery of the Services and/or to the IBM subcontractors assigned to the Equifax service team, Equifax will notify IBM of such belief and the basis for such belief. Upon receipt of such notice from Equifax, IBM (a) will promptly provide to Equifax a report setting forth IBM's position regarding the matters raised by Equifax in its notice; (b) will meet with Equifax to discuss the matters raised by Equifax in its notice and IBM's positions with regard to such matters; and (c) will promptly and diligently take Commercially Reasonable Efforts to modify or eliminate any IBM practices and/or processes identified as adversely impacting the performance and delivery of the Services.

8. RELATIONSHIP PROTOCOLS

8.1 Annual Updating of Schedules to this Agreement

The Parties agree to review and update the Schedules to this Agreement (including their respective Exhibits, Attachments, Appendices and Supplements) at least once a year during the Term to accurately reflect the implementation of agreed Changes. The preceding sentence is not intended, nor is it authorization, to expand the scope of the Services except as provided pursuant to Section 10.1 (Prices and Charges for New Services) of Schedule C (Charges). If the Schedules to this Agreement are not otherwise updated during any Contract Year, the Parties will review and update them during the first month after the end of such Contract Year

32

8.2 Required Consents

- (a) Equifax shall have the legal responsibility for timely obtaining all Required Consents under the Third Party Agreements to which a member of the Equifax Group is a party on the Execution Date. Unless Schedule G (Third Party Agreements) expressly provides otherwise, the applicable Equifax Group member will remain the contracting party of record on each such Third Party Agreement.
- (b) With regard to obtaining any Required Consents, IBM will provide Equifax with advice and counsel regarding IBM's experience and agreements with the Third Party Providers with respect to the agreements in Schedule G (Third Party Agreements), and the benefit of any relationship of IBM with each such Third Party Provider to the extent permitted under the IBM-Third Party Provider arrangement. IBM shall also have management and administrative responsibilities for obtaining Required Consents under such Third Party Agreements, subject to the consent of Equifax to the terms of each such Required Consent. Subject to the provisions of Section 8.3, IBM will use commercially reasonable efforts to obtain, and will act as Equifax's attorney in fact in connection with obtaining, such Required Consents, and any other Third Party Agreements that are entered into after the Execution Date. Upon obtaining a Third Party Provider's agreement to terms for a Required Consent, the Required Consent shall be provided to Equifax for review, approval, and signature. If IBM is unable to obtain the Required Consent within a reasonable time in a form acceptable to Equifax, then Equifax may pursue such Required Consent directly. If Equifax is unable to obtain the

Required Consent, Equifax and IBM will determine a reasonable alternative arrangement for the portion of the Services affected by the absence of such Required Consent. The cost of achieving such reasonable alternative arrangement shall be borne by IBM if caused by Required Consents needed from (i) IBM or Affiliates of IBM, (ii) from the licensors of the IBM Software, and/or (iii) from Third Party Provider under any Third Party Agreements treating outsourcing arrangements involving IBM as the services provider differently than their standard policies afforded to other outsourcing services providers generally, and in all other instances such cost shall be borne by Equifax.

- (c) The provisions of this paragraph shall govern financial responsibility for fees payable for or in connection with obtaining Required Consents under Equifax Group licenses for Third Party Provider Software listed on Schedule F (Software). Equifax is financially responsible for payment of such consent fees in an aggregate amount up to *, and IBM shall be financially responsible for any such consent fees in excess of *. IBM will pay the required fees to the Third Parties and may invoice Equifax for reimbursement for such payments (up to the * aggregate cap) in January 2004. If the actual aggregate amount of such fees is less than *, Equifax will reimburse IBM for the actual amount of such fees and, in addition, pay IBM a bonus in an amount equal to one-half $\binom{1}{2}$ of the difference between * and the actual amount of such fees. For the avoidance of doubt, Equifax will be financially responsible for all payments for any Required Consents for Third Party Software not listed on Schedule F (Software) as of the Execution Date.
- (d) Equifax is financially responsible for all payments for any Required Consents under Third Party Agreements listed in Schedule G (Third Party Agreements) other than the Equifax Group licenses for Third Party Provider Software listed on Schedule F (Software), which are subject to Section 8.2(c) above.
- * Information deleted pursuant to Rule 24b-2 of the Exchange Act.

33

(e) For all Third Party Agreements allocable to this Agreement entered into after the Execution Date, the Party having financial responsibility for the product or service to which the Third Party Agreement relates, as indicated in Exhibit C-14 (Financial Responsibilities Matrix) to Schedule C (Charges), shall bear the costs, if any, of obtaining any associated Required Consents. The provisions of this paragraph shall be applicable to New Services unless otherwise provided by the Parties in the Contract Change documentation governing New Services.

8.3 Appointment as Attorney In Fact

- (a) Equifax appoints IBM as the attorney in fact of the members of the Equifax Group, and IBM accepts such appointment as a part of the Services, for the limited purposes of administering, managing, supporting, operating under and paying under the Third Party Agreements to which one or more members of the Equifax Group is a party, and to obtain Required Consents as provided in Section 8.2, in connection with the Services as contemplated by this Agreement. Equifax does not appoint IBM as the attorney in fact of the members of the Equifax Group for the purposes of entering into oral or written agreements with any individual or business entity for or in the name of the Equifax Group or their Affiliates, without the prior express written approval of Equifax. Equifax agrees to promptly notify all Third Party Providers under the Third Party Agreements to which one or more members of the Equifax Group is a party of such appointment. Subject to its obligation to indemnify Equifax for any applicable penalties, damages, termination or other charges under Section 14.1, IBM may direct that the Equifax Group cancel, substitute, terminate, change or add to the Third Party Providers under the Third Party Agreements as it chooses so long as IBM continues to perform the Services in the manner required by this Agreement; provided, however, IBM must submit written notification to Equifax and obtain Equifax's written agreement prior to the cancellation, substitution, termination, change or addition of any Third Party Agreement to which one or more members of the Equifax Group is or will be a party. If Equifax does not respond to such notice from IBM within twenty-one (21) business days of Equifax's receipt of such notice, Equifax shall be deemed to have agreed to the cancellation, substitution, termination, change or addition described in the IBM notice. If any such cancellation, substitution, termination, change or addition of a Third Party Agreement will have an impact on the operations of users that are outside the scope of the Services and Equifax has notified IBM prior to the expiration of the Equifax response period described above of such impact and IBM elects to proceed, IBM will provide or cause to be provided the products and/or services that are the subject of such Third Party Agreement to the users that are outside the scope of the Services on terms no less favorable than the terms of the applicable Third Party Agreement.
- (b) IBM will perform its obligations and responsibilities as an attorney in fact pursuant to Section 8.3(a) under all Third Party Agreements to which a member of the Equifax Group is a party subject to the provisions of this Agreement, including, without limitation, Section 8.2, this Section 8.3, Section 9.1 and Article 11. Upon Equifax's request, IBM will provide to Equifax all information and documentation related to its activities as the Equifax Group's attorney in fact with regard to such Third Party Agreements. Equifax may terminate or provide additional restrictions on IBM's attorney in fact appointment with respect to any Third Party Agreement to which one or more of the members of the Equifax Group is a party if IBM (i) fails to pay any amount due in a timely manner; (ii) permits an actual default to occur; or (iii) does not diligently pursue the service and financial benefits available to the Equifax Group under such Third Party Agreement.
- (c) Beginning on the Commencement Date and for the Term, the Equifax Group will not enter into any new, or terminate or amend any existing, Third Party Agreement to which one or more members of the Equifax Group is a party that adversely impacts IBM's ability to provide the

34

Services or increases IBM's cost of providing such Services without the prior written consent of IBM.

8.4 Conflicts of Interests

- (a) Each Party recognizes that IBM personnel providing Services to the Equifax Group under this Agreement may perform similar services for others and this Agreement shall not prevent IBM from performing similar services for others subject to the restrictions set forth in Article 11; provided, however, IBM shall not use any of the Equifax Provided Hardware or Equifax Software or Equifax Provided Office Furnishings to perform similar services for others (including IBM), without the prior written consent of Equifax.
- (b) Neither Party, through its personnel at any site covered under this Agreement, shall knowingly, directly or indirectly, solicit any employee of the other Party or their Affiliates at such site during the Term of this Agreement unless otherwise agreed in writing by the Parties and except as provided in Section 12.5(g). Equifax or IBM employee's responses to or employment resulting from general public solicitations will be exempted from this provision.

8.5 Alternate Providers

(a) During the Term, Equifax shall have the right to retain third party suppliers (including suppliers of *) to perform any service, function, responsibility, activity or task that is within the scope of the Services or would constitute a New Service pursuant to Schedule C (Charges), or to perform any such services, functions, responsibilities or tasks (whether all or a part of the Services or the New Services) internally. IBM shall cooperate with any such third party supplier and Equifax

as requested from time to time. Such cooperation shall include, without limitation, (i) providing reasonable physical and electronic access to the Facilities, the Data Center and the books and records in the possession of IBM regarding the Equifax Business and/or the Services; (ii) use of any Machines used by IBM to perform services for the Equifax Group for the Equifax Business; (iii) use of any of the Software (other than any Software where the underlying license agreement does not authorize such access and consent permitting such access and use has not been obtained); (iv) providing such information (subject to an appropriate confidentiality agreement, if appropriate) regarding the operating environment, System constraints, and other operating parameters as is reasonably necessary for the work product of the third party supplier of the Equifax Group to be compatible with the Services or New Services; and (v) such other reasonable cooperation as may be requested by Equifax.

- (b) IBM's obligations hereunder shall be subject to the third party suppliers' compliance with reasonable Facilities and Data Center data and physical security and other applicable standards and procedures, execution of appropriate confidentiality agreements, and reasonable scheduling of computer time and access to other resources to be furnished by IBM pursuant to this Agreement.
- * Information deleted pursuant to Rule 24b-2 of the Exchange Act.

35

- (c) If IBM's cooperation with Equifax or any third party supplier performing work as described in Section 8.5(a) would constitute a New Service, IBM shall promptly so notify Equifax and the provisions of Section 10.1 of Schedule C (Charges) shall apply. The Parties further agree that if in IBM's reasonable, good faith determination, a third party supplier's activities impair IBM's ability to meet the Service Levels or otherwise provide the Services in accordance with this Agreement, IBM will provide written notice to Equifax of such determination. The Parties will cooperate to determine and verify whether such effect is caused by a third party supplier, the extent of such effect, and how to ameliorate any such effect. IBM shall be excused for any inability to meet the Service Levels or otherwise provide any of the Services to the extent, and only for the period, any such third party supplier's activities directly impair IBM's ability to meet any Service Level or otherwise provide any of the Services in accordance with this Agreement.
- (d) Equifax's retention of third party suppliers pursuant to this Section 8.5 to perform services, functions, activities, tasks or responsibilities that are within the scope of the Services shall not relieve Equifax of its obligations set forth in this Agreement to pay IBM the charges applicable to such services, functions, activities, tasks or responsibilities as set forth in this Agreement, unless Equifax is relieved from such charge pursuant to a provision of this Agreement or by the agreement of IBM. For the avoidance of doubt, the preceding sentence shall not be interpreted to require Equifax to pay IBM any amount, pursuant to a PxQ Calculation under Schedule C (Charges), for any Resource Units not actually used or consumed by Equifax except in any particular circumstances in which Equifax would be required to pay such amount to IBM pursuant to Schedule C (Charges), notwithstanding the fact that Equifax did not actually use or consume the Resource Units in question.

8.6 Use of Subcontractors

(a) The Parties will develop and prepare a list of approved subcontractors that the Parties agree may be engaged by IBM to perform and deliver the part or portion of the Services indicated on such list as a subcontractor to IBM (the "Listed Subcontractors"), which will be attached to this Agreement as Schedule R. Affiliates of IBM shall be deemed to be Listed Subcontractors. With respect to subcontractors which are not Listed Subcontractors, IBM shall notify Equifax at least fifteen (15) business days prior to the proposed date of commencement by IBM of any subcontractor's activity with respect to the Equifax Group or the Services, in writing of a decision to delegate or subcontract a function, responsibility or task to a subcontractor, or to change subcontractors for any function, responsibility or task, (i) that could have a material affect on the quality, timing, cost, consistency or performance of the Services or on the operations of any member of the Equifax Group, or (ii) where the subcontractor will interface directly with the members of the Equifax Group. Upon Equifax's request, IBM shall promptly provide to Equifax information regarding the proposed new or replacement subcontractors in order to permit Equifax to determine whether to grant its consent to such delegation or change or subcontract. Such information shall include the scope of the Services to be delegated, and the experience, financial status, resources, and reason for selection of the proposed subcontractors. Subject to IBM's timely provision of the foregoing information to Equifax, Equifax shall be deemed to have accepted such delegation or subcontract or change that is the subject of the notification by IBM to Equifax, if Equifax has not notified IBM in writing of its good faith objections to such delegation or subcontract or or before the fifteenth (15th) business day after receipt of such notice from IBM. IBM shall not delegate or subcontract or change subcontractors unless and until IBM and Equifax shall have resolved any objection tim

36

Equifax to such proposed action by IBM. In addition, IBM shall not disclose any Confidential Information of the Equifax Group to any subcontractor unless and until such subcontractor has agreed in writing to protect the confidentiality of such Confidential Information in a manner equivalent to that required of IBM by Article 11.

- (b) Equifax shall have the right to request that IBM replace a subcontractor if such subcontractor has entered a business that is primarily competitive with Equifax's main business.
- (c) IBM shall remain primarily liable and obligated to Equifax for the timely and proper performance of all of its obligations hereunder even if such obligations are delegated to third party subcontractors (including, without limitation, Affiliates of IBM entering into Local Enabling Agreements with Equifax and/or Affiliates of Equifax), and for the proper and timely performance and actions of any person or entity to which it delegates or subcontracts any such obligation.
- (d) Subcontractors used by IBM in performing the Services shall be subject to the provisions of Section 3.7 and Schedule T (Security Procedures).

8.7 Equifax Approvals and Notification

For those areas of the Services where Equifax (a) has reserved a right-of-approval, consent or agreement, (b) is required to provide notification, and/or (c) is to perform a responsibility set forth in this Agreement, and such approval, consent, notification or performance is delayed or withheld beyond the period provided in this Agreement, without authorization or right and, such delay or withholding is not caused by IBM and affects IBM's ability to provide the Services under this Agreement, IBM will be excused from any failure to meet the Service Levels for any affected portion of the Services to the extent, but only to the extent, such failure is directly caused by such delay or withholding. If not specified otherwise in this Agreement, the period for such approval or notification shall be fifteen (15) business days unless another time period is otherwise agreed by the Parties.

9. CHARGES

Schedule C (Charges) to this Agreement sets forth the pricing, invoicing and payment methodologies and processes for the charges related to the Services. For the avoidance of doubt, Equifax acknowledges that it will receive and be obliged to pay invoices for IBM's charges under the existing IBM-Equifax agreements set forth in Schedule M for Services provided on or before July 31, 2003 thereunder.

10. INTELLECTUAL PROPERTY RIGHTS

IBM, the members of the Equifax Group and their respective contractors and subcontractors may develop, create, modify or personalize (collectively, "Develop," "Developed" or "Developing") certain computer programming code, including source and object code ("Code") and other Materials in order to perform the Services. The provisions of this Article 10 set forth the respective rights of Equifax and IBM in such Code and other Materials.

37

10.1 Ownership of Materials

With respect to any Materials whether Developed solely by IBM or its subcontractors, or jointly by the Equifax Group personnel or their subcontractors and IBM or its subcontractors, ownership will be as follows:

- (a) Equifax Code, Equifax Derivative Code and Equifax Works, and all * other than Equifax Materials Developed by IBM under this Agreement (for example, Equifax Materials developed outside the scope of this Agreement or existing prior to the Commencement Date), shall be owned by Equifax or another member of the Equifax Group, as applicable. During the Term, IBM shall have an irrevocable, nonexclusive, worldwide, paid-up license to use, execute, reproduce, display, perform, operate, distribute, modify, develop, personalize and create Derivative Works from such Materials *.
- (b) IBM Code, IBM Derivative Code, IBM Works and IBM Interfaces, and all patent rights created in the course of Developing IBM Code, IBM Derivative Code, IBM Works and IBM Interfaces, shall be owned by IBM. The Equifax Group shall have an irrevocable, nonexclusive, worldwide, paid-up license to use, execute, operate, reproduce, display, perform, distribute, modify, Develop, personalize and create Derivative Works from such Materials internally, and the right to sublicense third parties to do any of the foregoing, to the extent necessary and for the sole purpose of receiving or using the Services during the Term.
- (c) With respect to any Equifax Materials whether or not Developed under this Agreement, which are or have been Developed solely by the Equifax Group personnel and/or their contractors, such Materials and all patent rights created in the course of Developing such Materials shall be owned by Equifax. IBM shall have an irrevocable, nonexclusive, worldwide, paid-up license to use, execute, operate, reproduce, display, perform, distribute, modify, Develop, personalize and create Derivative Works from such *.

(d) *

- (e) Any ownership or license rights herein granted to either Party or another member of the Equifax Group or any other Authorized Users are limited by and subject to any *, and terms and conditions of * applicable Third Party Providers.
- (f) To the extent that by operation of law any of the Materials, * may not be owned by IBM or the Equifax Group to which ownership has been allocated under this Article 10, each Party agrees to promptly assign, or cause to be assigned, and take such actions and execute and deliver such documents as shall be necessary or appropriate to effect such assignment without further consideration. Each Party hereby assigns, without further consideration, the ownership of all right, title and interest in all U.S. and foreign copyrights, and mask work rights (if any) in the Materials to the other Party as set forth in this Article 10. Such assignee shall have the right to obtain and hold in its own name or transfer patents and copyrights, applications, registrations, renewals and all other rights relating or pertinent thereto.
- * Information deleted pursuant to Rule 24b-2 of the Exchange Act.

38

10.2 Obligations Regarding Materials

- (a) Following the Development of Equifax Code, Equifax Derivative Code or Equifax Works or the creation of any invention (whether or not patentable) in the course of such Development, by IBM (or any of its Affiliates or subcontractors), IBM will follow its standard processes and procedures to assess whether any patentable ideas have been identified or discovered and provide to Equifax any invention disclosure prepared by IBM in accordance with its standard processes and procedures.
- (b) The Parties agree to reproduce copyright legends which appear on any portion of the Materials which may be owned by the Parties and any and all third parties.
- (c) Except as set forth in this Article 10 or in Article 11, this Agreement shall not preclude either Party from Developing materials or providing services which are competitive to the Materials or Services which might be delivered pursuant to this Agreement, except to the extent any of same may infringe any of the other Party's patent rights, copyrights, trade secrets or mask work rights.
- (d) Neither this Agreement nor any disclosure made hereunder grants any license to either Party under any patents rights, copyrights, mask work rights or trade secrets of the other Party, except for the licenses expressly granted under this Article 10 and Section 12.6 hereof.
- (e) Each Party and their respective Affiliates shall have the right to develop commercialize, use, publish and distribute materials and/or intellectual property which may be substantially similar to the Materials (including, without limitation, computer programs and other copyrighted works) for their own use, for third parties or for other purposes provided that such activities are effected without breach of their obligations under this Agreement and do not infringe the intellectual property rights of the other Party and/or its Affiliates.

10.3 Authorized Users

The Authorized Users (excluding the Equifax Group) shall have only such rights to the intellectual property comprising the System as shall be granted by Equifax which in no event shall be greater than any right Equifax may have to such intellectual property. All such rights shall be subject to this Agreement.

11. CONFIDENTIALITY/DATA SECURITY

11.1 Confidential Information

(a) IBM and Equifax each acknowledge that the other Party and/or its Affiliates possesses and will continue to possess information, which has commercial value in such other Party's and/or its Affiliates' business and is not in the public domain. Such information has been created, discovered, developed by such other Party and its Affiliates or provided to it by a third party, and such other Party and/or its Affiliates holds property rights in such information by assignment, license or otherwise. "Confidential Information" means with respect to a Party, any and all proprietary business information of the disclosing Party and/or its Affiliates

and/or of third parties in the possession of the disclosing Party and its Affiliates treated as secret by the disclosing Party and its Affiliates (that is, it is the subject

39

information related to the services and/or business of the disclosing Party and/or its Affiliates and/or of a third party which (a) derives economic value, actual or potential, from not being generally known to or readily ascertainable by other persons who can obtain economic value from its disclosure or use; and (b) is the subject of efforts by the disclosing Party and/or its Affiliates that are reasonable under the circumstances to maintain its secrecy, including without limitation (i) marking any information reduced to tangible form clearly and conspicuously with a legend identifying its confidential or proprietary nature; (ii) identifying any oral presentation or communication as confidential immediately before, during or after such oral presentation or communication; or (iii) otherwise, treating such information as confidential or secret. Assuming the criteria in sections (a) and (b) above are met, Trade Secrets include, but are not limited to, technical and nontechnical data, formulas, patterns, compilations, computer programs and software, devices, drawings, processes, methods, techniques, designs, programs, financial plans, product plans, and lists of actual or potential customers and suppliers. "Company Information" means collectively the Confidential Information, Equifax Data and Trade Secrets. Company Information also includes information which has been disclosed to either Party and/or its Affiliates by a third party which such Party and/or its Affiliates is obligated to treat as confidential or secret.

(b) For the purposes of obligations under Article 11, "Authorized Users" are included within the concept of "Equifax and/or its Affiliates" or "Party and/or its Affiliates".

11.2 Obligations

(a) Equifax and IBM will each refrain from disclosing, will hold as confidential, and will use the same level of care to prevent disclosure to third parties and to hold confidential, the Company Information of the other Party as it employs to avoid disclosure, publication or dissemination of its own information of a similar nature but in no event less than a reasonable standard of care. Notwithstanding the foregoing, the Parties and their Affiliates may disclose Company Information in the case of Equifax and its Affiliates, to members of the Equifax Group or to companies divested by the Equifax Group that elect to receive services hereunder as an Authorized User, and in the case of both Parties and their Affiliates, to companies divested by the Equifax Group that elect to receive services hereunder as an Authorized User or to authorized contractors and subcontractors involved in providing and using the Services under this Agreement where: (i) such disclosure is necessary to permit the members of the Equifax Group or any divested companies of the Equifax Group that receive services hereunder as an Authorized User, or any authorized contractor or subcontractor to perform its duties hereunder or use the Services; (ii) members of the Equifax Group and such divested companies of the Equifax Group that elect to receive services hereunder or any authorized contractor agree in writing to observe the confidentiality and restricted use and disclosure covenants and standards of care set forth in this Article 11 and IBM and Equifax are each third party beneficiaries for all purposes; and (iii) IBM in the case of Equifax Company Information received by IBM and/or its Affiliates and disclosed by them as permitted herein or Equifax in the case of IBM Company Information received by Equifax and/or its Affiliates and disclosed by them as permitted herein, assumes full responsibility for the acts or omissions of its Affiliates, contractors and subcontractors or, in the case of Equifax, its divested companies receiving services h

(b) Neither Equifax nor IBM shall use the Company Information of the other Party except in the case of IBM and its Affiliates and subcontractors, (i) in connection with the performance of the Services and (ii) as otherwise specifically permitted in this Agreement, and in the case of Equifax, its contractors and other members of the Equifax Group, (A) as specifically permitted in this

40

Agreement and (B) in connection with the use of the Services. IBM shall be responsible to ensure that its Affiliates and subcontractors comply with this Section 11.2(b) and Equifax shall be responsible to ensure that the members of the Equifax Group and its contractors comply with this Section 11.2(b).

- (c) Without limiting the generality of the foregoing, neither Party nor their Affiliates will publicly disclose the terms of this Agreement, except to the extent permitted by this Article 11 and to enforce the terms of this Agreement, without the prior written consent of the other. Furthermore, neither IBM nor Equifax nor their Affiliates will make any use of the Company Information of the other Party and its Affiliates except as contemplated by this Agreement; acquire any right in or assert any lien against the other Party's Company Information except as contemplated by this Agreement; or refuse to promptly return, provide a copy of or destroy such Company Information upon the request of the disclosing Party.
- (d) Notwithstanding any other provision of the Agreement, neither Party will be restricted in using, in connection with its business operations, any * ideas, concepts, know-how and techniques which are retained in the unaided memories of employees who have had access to the other Party's Confidential Information, subsequent to the Commencement Date, without deliberate memorization of such Confidential Information for purposes of reuse under this paragraph ("Residual Knowledge") *.

11.3 Exclusions

Notwithstanding the foregoing, this Article 11 will not apply to any information which IBM or Equifax can demonstrate was: (a) at the time of disclosure to it, in the public domain; (b) after disclosure to it, published or otherwise becomes part of the public domain through no fault of the receiving party; (c) without a breach of duty owed to the disclosing party, is in the possession of the receiving party at the time of disclosure to it; (d) received after disclosure to it from a third party who had a lawful right to and, without a breach of duty owed to the disclosing party, did disclose such information to it; or (e) independently developed by the receiving party without reference to Company Information of the disclosing party. Further, either Party may disclose the other Party's Company Information to the extent required by law or order of a court or governmental agency. However, the recipient of such Company Information must give the other Party prompt notice and make a reasonable effort to obtain a protective order or otherwise protect the confidentiality of such information, all at the discloser's cost and expense. It is understood that the receipt of Company Information under this Agreement will not limit or restrict assignment or reassignment of employees of IBM and its Affiliates and the Equifax Group within or between the respective Parties and their Affiliates.

11.4 Loss of Company Information

The receiving Party will immediately notify the disclosing Party, orally or in writing in the event of any disclosure, loss, or use in violation of this Agreement of a disclosing Party's Company Information known to the receiving Party.

11.5 Limitation

The covenants of confidentiality set forth herein (a) will apply after the Commencement Date to any Company Information disclosed to the receiving Party before and after the Commencement Date and (b) will continue and must be maintained from the Commencement Date through termination of the relationship between the Parties and (i) with respect to Trade Secrets, until the earlier of ten (10) years after termination of this Agreement or until such Trade Secrets no longer qualify as trade secrets under applicable law; and (ii) with respect to Confidential Information for a period equal to the shorter of two (2) years after termination of the Parties' relationship under this Agreement, or until such Confidential Information no longer qualifies as confidential under applicable law. Neither Party will be responsible for the security of the Company Information of the other Party during transmission via public communications facilities or for the loss of or damage to such information during transmission, except to the extent that such breach of security or loss or damage is caused by the failure of such Party to perform its obligations under this Agreement, including exercising the standard of care set forth in Section 11.2(a).

11.6 Equifax Data

(a) All of Equifax's Company Information (including, without limitation, data, records and reports related to the Equifax Group, the Equifax Business and the Services) is represented by Equifax to be the exclusive property of Equifax, and/or its Affiliates or the property of third parties licensed to Equifax and/or its Affiliates, and the furnishing of such information, data, records and reports to, or access to such items by, IBM and/or its Affiliates and/or subcontractors will not grant any express or implied license to or interest in IBM and/or its Affiliates and/or subcontractors relating to such information, data, records and reports except as required to perform the Services pursuant to this Agreement. Unless specifically provided otherwise in this Agreement, IBM shall have no responsibility with respect to compliance with laws or regulations applicable to the storage, maintenance, and distribution of Equifax Company Information to the extent that any such activity by IBM is performed or implemented in accordance with Equifax's legal compliance officer's written instruction or direction. Upon request by Equifax at any time and from time to time and without regard to the default status of the Parties under this Agreement, IBM and/or its Affiliates and/or subcontractors shall promptly deliver to Equifax's Company Information (including without limitation all data, records and related reports regarding the Equifax Group, the Equifax Business and the Services) in electronic (tape) format and in such hard copy as existing on the date of the request by Equifax.

(b) IBM personnel shall not attempt to access, or allow access to, any Equifax Data which they are not permitted to access under this Agreement. If such access is attained, IBM shall immediately report such incident to Equifax, describe in detail the accessed Equifax Data and return to Equifax any copied or removed Equifax Data.

11.7 Data Privacy

In carrying out their activities under this Agreement, the Parties shall observe and comply with the data privacy and protection requirements set forth in Schedule P (Data Protection).

42

12. TERMINATION

12.1 Termination By Equifax

Subject to the Termination Charges set forth in Schedule C (Charges), where applicable, Equifax may terminate the applicable portion(s) of this Agreement for the following reasons:

- (a) A material breach of this Agreement by IBM and/or its Affiliates that remains uncured for ten (10) days after receipt of written notice thereof; provided, however, if a material breach of this Agreement by IBM and/or its Affiliates (other than a breach of Article 11 hereof) occurs that by its nature cannot be cured by IBM in such ten (10) day period but IBM submits a commercially reasonable written plan to Equifax within such period to cure such breach after the ten (10) day period (but in no event more than forty five (45) days after such notice of breach), the cure period for such breach shall be extended to the date set forth in the plan (but in no event more than sixty (60) days after the notice of breach). If the material breach by its nature cannot be cured within sixty (60) days, then Equifax may terminate this Agreement by providing notice to IBM; or
- (b) There exists a series of non-material or persistent breaches by IBM and/or its Affiliates that in the aggregate have a material and significant adverse impact (i) on the Services support of the administrative, management, planning, financial reporting or operations functions of the Equifax Group or the portion of the Equifax Group constituting the user group, or (ii) on the management of the Services or the portion of the Services; or
- (c) For convenience upon one hundred eighty (180) days prior notice by Equifax to IBM, Equifax may terminate by Service Tower or by Country Location or this entire Agreement at any time after the third anniversary of the Commencement Date; or
- (d) In the event that (i) another entity, directly or indirectly, in a single transaction or series of related transactions, acquires either Control of Equifax or all or substantially all of the assets of Equifax, or (ii) Equifax is merged with or into another entity, upon one hundred eighty (180) days prior notice by Equifax to IBM, which notice must be given within 180 days after the Change of Control; or
- (e) IBM and/or its Affiliate that has accepted this Agreement by executing a Local Enabling Agreement becomes insolvent or is unable to pay its debts or enters into or files (or has filed or commenced against it) a petition, arrangement, application, action or other proceeding seeking relief or protection under the bankruptcy laws of the United States or any similar laws of the United States or any other country or transfers all or substantially all of its assets to another person or entity; or
- (f) IBM and/or its Affiliate that has accepted this Agreement by executing a Local Enabling Agreement incurs * in excess of the * under the circumstances and resulting from the events described in Section 13.1(a)(i);
- (g) If Equifax's license to use credit data in Spain shall expire or be terminated prior to the end of the Term of this Agreement, Equifax may terminate this Agreement with respect to Services provided in Spain as of a date specified by Equifax in a written notice of termination to IBM, and Equifax will pay IBM's charges due and payable through the termination date. If Equifax elects such termination, Equifax shall pay on account of such termination IBM's Wind-Down Expenses for a period not to exceed one hundred twenty (120) days and shall acquire from IBM, at a cost equal to IBM's then-current net book value, and hardware or software primarily dedicated to performing Services for Equifax that cannot be redeployed by IBM using Commercially Reasonable Efforts.; or

Information deleted pursuant to Rule 24b-2 of the Exchange Act.

12.2 Termination by IBM

In the event, but only in the event, that Equifax fails to pay IBM when due undisputed charges totaling at least one (1) month's charge under this Agreement and fails to make such payment within sixty (60) days after receiving notice from IBM of the failure to make such payment, IBM may, by giving written notice to Equifax, terminate this Agreement as of a date specified in the notice of termination.

12.3 Services Transfer Assistance

- (a) The Parties agree that IBM will cooperate with the Equifax Group to assist in the orderly transfer of the services, functions, responsibilities, tasks and operations comprising the Services provided by IBM and its Affiliates hereunder to one or more members of the Equifax Group itself or another services provider in connection with the expiration or earlier termination of this Agreement for any reason, however described. The Term of this Agreement shall not be deemed to have expired or terminated until the Services Transfer Assistance thereunder is completed. Upon Equifax's request IBM or its Affiliate shall provide transfer assistance in connection with migrating the work of the Equifax Group to the Equifax Group itself or another services provider ("Services Transfer Assistance") commencing up to one (1) year prior to expiration or upon any notice of termination, or of non-renewal of this Agreement. In the event Equifax repeatedly fails to pay any amounts when due and payable under this Agreement within two (2) years of the start of Services Transfer Assistance, with or without an attendant termination for cause by IBM, IBM shall not be required to provide Services Transfer Assistance unless Equifax prepays the applicable total monthly charges due under Schedule C (Charges) for the entire duration of Services Transfer Assistance, if any, and a reasonable projection of other charges due for the entire period Equifax requests Services Transfer Assistance. In no event will Equifax's holding of or escrow of monies in compliance with Section 8.2 (Disputed Charges/Credits) of Schedule C (Charges) be considered a failure by Equifax to pay amounts due and payable hereunder. Further, IBM shall provide the Services Transfer Assistance in accordance with this Section 12.5 even in the event of Equifax's material breach (other than an uncured payment default) with or without an attendant termination for cause by IBM, if Equifax prepays a reasonable projection of the other charges due under this Agreement (other than the total monthly charges due under Schedule C (Charges), which shall be paid monthly as provided in Schedule C (Charges)) for the Services Transfer Assistance for the entire period Equifax desires IBM to provide such services to the Equifax Group or its designees. Services Transfer Assistance shall be provided through the effective date of the expiration or termination of the Services being terminated, and upon request by Equifax, the effective date of such expiration or termination shall be extended for up to one (1) year thereafter pursuant to the terms and conditions of this Agreement and such period shall be considered an extension of the Term, however any such extension shall not affect the payment date or amount of any applicable Termination Charges, which Termination Charges shall be due and payable as of the initially noticed effective date of termination. Services Transfer Assistance shall include, but not be limited to, providing the Equifax Group and their respective agents, contractors and consultants, as necessary, with the services described in Schedule O (Services Transfer Assistance).
- (b) If the provision of any Services Transfer Assistance by IBM would require IBM to perform New Services, the provisions of Section 10.1 (Prices and Charges for New Services) of Schedule C (Charges) shall apply.
- (c) If Equifax exercises its option to prepay the monthly charges due under Schedule C (Charges) and other charges reasonably projected by IBM for Services Transfer Assistance and it is

44

determined that such prepayment is in excess of the actual charges associated with the Services Transfer Assistance, then IBM shall apply such overpayment to monies otherwise due IBM or, if no monies are due IBM, promptly refund such overpayment to Equifax at the end of such Services Transfer Assistance. Conversely, if the amount prepaid by Equifax to IBM for Services Transfer Assistance does not fully reimburse IBM for the actual charges due under Schedule C (Charges) for the provision of Services Transfer Assistance to Equifax, then IBM shall invoice Equifax and Equifax shall promptly pay IBM for such additional amounts as incurred and invoiced to Equifax.

(d) In the process of evaluating whether to undertake or allow termination, expiration or renewal of this Agreement, Equifax may consider obtaining, or determine to obtain, offers for performance of services similar to the Services following termination, expiration or renewal of this Agreement. As and when reasonably requested by Equifax for use in such a process, IBM shall provide to Equifax such information and other cooperation regarding performance of the Services as would be reasonably necessary for a third party to prepare an informed, non-qualified offer for such services. The types of information and level of cooperation to be provided by IBM pursuant to this Section 12.3(d) shall be no less than those initially provided by Equifax to IBM prior to commencement of this Agreement.

12.4 Equitable Remedies

If a court of competent jurisdiction should find that IBM has breached its obligations to provide Service Transfer Assistance, IBM agrees that, *.

12.5 Other Rights Upon Termination

At the expiration or earlier termination of this Agreement for any reason, however described, IBM agrees in each such instance, as applicable:

- (a) Upon Equifax's request, IBM agrees to sell to Equifax or its designee the IBM Machines owned by IBM then currently being used by IBM primarily to perform the Services or the portion of the Services, as applicable, at its then-current unamortized net book market value. In the case of IBM Machines that IBM is leasing and using primarily to perform the Services, IBM agrees to permit Equifax or its designee to either buy-out the lease on the IBM Machines and purchase the IBM Machines from the lessor or assume the lease(s) and secure the release of IBM thereon, subject to the terms of the applicable lease. Equifax shall be responsible for any sales, use or similar taxes associated with such purchase of such IBM Machines or the assumption of such leases.
- (b) To the extent that IBM is using commercially available IBM proprietary software to perform the Services on the termination or expiration of the Agreement, IBM will grant to the members of the Equifax Group and their Affiliates a license (which license shall permit a third party designee to use such software solely for Equifax's benefit) to such software (and any related documentation) on IBM's then standard terms and conditions (other than any one-time charges which shall not be required).

- (c) IBM will provide to the Equifax Group * in accordance with * for use by the Equifax Group as a part of and in connection with the Equifax Business, upon terms and prices to be mutually agreed upon by the Parties (which prices shall not be greater and other terms shall be no less favorable than those then offered to other customers of IBM) or, in the case where no such customers exist, other third parties). At Equifax's option, IBM will recommend a mutually agreeable commercially available substitute, if available, to perform the same function.
- (d) Subject to Section 12.6(e), if IBM has licensed or purchased and is using any generally commercially available Software to provide the Services to the Equifax Group at the date of expiration or termination of this Agreement, Equifax may elect to take a transfer or an assignment of the license for such software (and any attendant maintenance agreement), subject to the terms of such license reimburse IBM for the initial license or purchase charges for such Software in an amount equal to the remaining unamortized cost of such Software, if any, depreciated over a five (5) year life. Equifax shall also pay any transfer fee or charge imposed by the applicable vendor and not the obligation of IBM hereunder, and subject to Equifax's acceptance of any applicable vendor terms and conditions, such licensed Software shall be transferred or assigned to Equifax.
- (e) If IBM has licensed or purchased and is using any generally commercially available Software to provide the Services to the Equifax Group and other IBM customers in a shared environment at the date of expiration or termination of this Agreement, IBM, upon request by Equifax, will assist Equifax in obtaining licenses for such Software (and any attendant maintenance agreement) subject to Equifax's payment of any license fee and other charge imposed by the applicable vendor.
- (f) IBM will use Commercially Reasonable Efforts to negotiate license arrangements with third parties that will minimize the amount of license and maintenance agreement and assignment fees to be paid by Equifax. If IBM is unsuccessful in any such negotiations, it will so notify Equifax prior to executing the affected agreement, in which event Equifax may elect participate in the negotiation of such license and maintenance agreement arrangements. IBM shall provide reasonable advance written notice to Equifax of such anticipated negotiations.
- (g) In the case of the impending expiration or termination of this Agreement for any reason, subject to local law, the Equifax Group shall have the right (or, as required by applicable local law, the duty) to make offers of employment to any or all IBM employees performing material Services for the Equifax Group hereunder, as applicable ("Service Employees"). Promptly after either Party provides the other Party written notice of termination or expiration with the prior consent of each Services Employee (each of whom IBM will notify of Equifax's interest), IBM agrees, subject to the agreement of the Service Employees, to supply Equifax with the names and resumes requested by Equifax for the purpose of exercising its rights under this Section 12.5(g), at no charge. Equifax's rights under this Section 12.5 will take precedence over any IBM/employee employment contract or covenant that may otherwise limit an employee's right to accept employment with the Equifax Group.
- * Information deleted pursuant to Rule 24b-2 of the Exchange Act.

46

(h) Upon Equifax's request, IBM will transfer or assign to Equifax or its designee, on mutually acceptable terms and conditions, any Third Party Agreements not otherwise treated in this Section 12.5, applicable solely to services being provided to Equifax, including, without limitation, Third Party Agreements for maintenance, Disaster Recovery Services and other necessary third party services then being used by IBM to perform the Services subject to the payment by Equifax of any transfer fee or charge imposed by the applicable vendors.

12.6 Effect of Termination/Survival of Selected Provisions

Notwithstanding the expiration or earlier termination of the Services or this Agreement for any reason however described, the following Sections of this Agreement shall survive any such expiration or termination: Section 8.3(b), Article 10, Article 11, Section 12.5, Section 12.6, Article 13, Article 14, Article 15, Section 16.1 and Article 17.

12.7 Savings Clause

Equifax's failure to perform its Equifax Retained Functions or its responsibilities set forth in this Agreement (other than as provided in Section 12.2) shall not be deemed to be grounds for termination by IBM. IBM's nonperformance of its obligations under this Agreement shall be excused if and to the extent (a) such IBM nonperformance results from Equifax's failure to perform its responsibilities; and (b) IBM provides Equifax with reasonable notice of such nonperformance and uses Commercially Reasonable Efforts to perform notwithstanding Equifax's failure to perform (with Equifax reimbursing IBM for its additional out-of-pocket expenses for such efforts).

13. LIABILITY

13.1 Liability Caps

- (a) Except as provided in Section *, the liability of IBM and its Affiliates to Equifax and its Affiliates under this Agreement arising out of or resulting from the performance or non-performance of IBM and/or its Affiliates and/or subcontractors of the Services and its obligations shall be limited in the aggregate for all claims, causes of action or occurrences:
- (i) to Direct Damages incurred by Equifax and its Affiliates equal to the charges paid by Equifax for the Services in the affected country during the * immediately prior to the first event which is the subject of the first claim or if * have not elapsed in the term of this Agreement at the time of the first such event, the estimated charges to Equifax for the Services in the affected country set forth in Schedule C (Charges) during the first * of the Term ("IBM Direct Damages Cap"); and
- (ii) in the event Equifax claims Direct Damages for event(s) which are the subject matter of claim(s) or cause(s) of action which are the basis for and result in Equifax's termination of this Agreement pursuant to Section 12.1(a) for cause or Section 12.1(e) for Bankruptcy, and the * operates to preclude Equifax's recovery of *, then * shall be entitled to recover an *, not to exceed * of *, which amount shall be applied only toward *.
 - (b) Except as provided in Section *, the liability of Equifax to IBM arising out of or resulting from the performance and non-performance of its obligations in each country shall be limited in all cases to Direct Damages which in the aggregate shall not exceed the amounts payable by Equifax upon a termination for convenience under Schedule C (the "Equifax Direct Damages Cap"). The IBM Direct Damages Cap and the Equifax Direct Damages Cap are herein collectively called the "Direct Damages Caps".
- * Information deleted pursuant to Rule 24b-2 of the Exchange Act.

13.2 *

13.3 Direct Damages and Cover Charges

Unless specifically provided to the contrary in this Agreement, neither party shall have any liability whether based on contract, tort (including without limitation, negligence), warranty, guarantee or any other legal or equitable grounds to the other party for any damages other than Direct Damages.

- (a) "Direct Damages" mean actual, direct damages incurred by the claiming Party which include, by way of example but without limitation, (i) the costs to * rendered by *, (ii) the difference in the amounts to be paid * and the * to provide, and/or the costs incurred by *, all or a portion of the Services during any period or periods that * the Services, (iii) the Service Credits, (iv) Transition Cover Costs, and (v) similar damages, but "Direct Damages" shall not include (A) loss of interest, profit or revenue of the claiming Party or (B) incidental, consequential, special or indirect damages suffered by the claiming Party (except as the damages described in (A) and (B) are included as a part of the Termination Charge and the Service Credits or as otherwise provided for in this Agreement) and shall not include punitive or exemplary damages suffered by the claiming Party arising from or related to this Agreement, even if such Party has been advised of the possibility of such losses or damages.
- (b) "Transition Cover Costs" means all costs and expenses incurred by the Equifax Group to Transition to another provider of information management and communications services, and/or take in-house, some or all of such functions, responsibilities, tasks and activities comprising the portion of the Services provided under any terminated portion(s) of this Agreement, after Commercially Reasonable Efforts to mitigate such costs and expenses.

13.4 Dependencies

In no event will IBM or its subcontractors be liable for any damages if and to the extent caused by Equifax's or its Affiliates' or its subcontractors' failure to perform its responsibilities hereunder; provided, however, for the purposes of this Section 13.4, neither IBM nor its Affiliates nor the Third Party Providers shall be considered a subcontractor of Equifax. Neither Equifax nor its Affiliates or contractors shall be liable for any damages if and to the extent caused by any failure to perform by IBM or its Affiliates or subcontractors.

13.5 Remedies

At its option, Equifax may seek all remedies available to it under law and in equity or recover as liquidated damages the Service Credits, subject to the limitations and provisions specified in this Article 13. If IBM's provision of the Services is such that IBM would otherwise owe Equifax a Service Credit and Equifax elects to recover Service Credits, Equifax's recovery of Service Credits shall constitute acknowledgment by Equifax of full satisfaction and release of any claim by Equifax that IBM has breached its obligations under this Agreement with respect to any such event(s) giving rise to the Service Credits. However, within twelve (12) calendar months of the receipt of any Service Credits Equifax received with respect to any action or inaction by IBM upon which Equifax is basing termination for cause under Section 12.1(a) or termination for persistent breaches under Section 12.1(b), Equifax may return such Service Credits and pursue a damage claim against IBM, if any such claim exists.

* Information deleted pursuant to Rule 24b-2 of the Exchange Act.

48

14. INDEMNITIES

14.1 Indemnity by IBM

IBM will indemnify and hold each member of the Equifax Group and their respective officers, directors, employees, agents, successors, contractors and assigns (each an "Indemnitee") harmless from and against any and all Losses incurred by any of them arising from or in connection with:

- (a) any Claims of infringement asserted against an Indemnitee as a result of * on or after the Commencement Date, including the *;
- (b) subject to Section 14.1(a) above, any Claims of infringement of any patent or any copyright, trademark, service mark, trade name, trade secret, or similar property right conferred by contract or by common law or by any law of any country, including without limitation, the United States, and any other applicable jurisdiction or any state, alleged to have been incurred because of or arising out of any aspect of the Services (including without limitation any information technology, information management and communications services, equipment, software or other resources) provided by IBM and/or its Affiliates or subcontractors in its performance of the Services; provided, however, IBM will have no obligation with respect to any Losses to the extent arising from or in connection with * related to the Services committed by an Indemnitee or any employee of an Indemnitee that is not the result of IBM and/or its Affiliates or subcontractors * including, without limitation, obtaining * for which it has responsibility; and provided, further, that IBM will have no obligation with respect to any Losses to the extent arising out of or in connection with * provided by IBM and/or its Affiliates or subcontractors, or an Indemnitee's * provided by IBM and/or its Affiliates or subcontractors, or an Indemnitee's use of * provided by IBM and/or its Affiliates to such Indemnitee * specifically designated in * or a written notice to Equifax from IBM;
- (c) any Claims, however described (including without limitation, failure to obtain Required Consents or arising from IBM's exercise of its rights to terminate, modify or change the Third Party Agreements pursuant to Section 8.3(a)), accruing during the Term (that is, not arising or resulting from a breach by the Equifax Group before the Commencement Date or after the termination date of this Agreement) regarding any Third Party Agreement; provided, however, IBM will have * to the extent arising out of or in connection with Claims for * related to the Services (i) committed by any Indemnitee or any employee of an Indemnitee that is not the result of IBM and/or its Affiliates or subcontractors * for which it has responsibility or (ii) to the extent arising out of or resulting from Equifax * for which it has responsibility;
- (d) any Claims for personal injuries, death or damage to tangible personal or real property of third parties including employees of IBM, its Affiliates, contractors and subcontractors caused by the negligence or willful misconduct of IBM, its employees, Affiliates, contractors or subcontractors; provided that IBM will have no obligation under this part, to the extent the same arise out of or in connection with the negligence or willful misconduct of a member of the Equifax Group;
- (e) any Claims for amounts, including but not limited to taxes, interest and penalties, assessed or claimed against the Equifax Group which are obligations of IBM under this Agreement;
- * Information deleted pursuant to Rule 24b-2 of the Exchange Act.

- (f) any Claim for violation of environmental laws or regulations arising out of this Agreement or as a result of the Services performed at the Facilities, the Data Center or the other Equifax Group sites or locations to the extent IBM or its Affiliates or subcontractors has caused the environmental damage or violation of the environmental laws or regulations from which the Claim arises;
- (g) any Claims directly attributable to IBM's decision to * and Equifax's * and Losses incurred by Equifax associated with *;
- (h) any Claims for penalties, interest and other charges imposed by a taxing authority (except the actual taxes payable by Equifax under the terms of this Agreement) arising out of or resulting from IBM * provided to Equifax in writing regarding * to Equifax;
- (i) any Claims by any Transferred Employees in respect of which IBM is obliged to indemnify Equifax pursuant to Exhibit D-1, D-2, D-3, D-4 or D-5 of Schedule D (Human Resources); and
- (j) any Claims arising out of or resulting from a breach by IBM and/or its Affiliates of * of this Agreement to the extent that such * are of specific application to the provision of the Services by IBM and/or its Affiliates to the Equifax Group under this Agreement.
- (k) In the event and to the extent that a Claim is made against an Indemnitee by an employee of IBM, its contractors or subcontractors providing services, products and/or software hereunder, the Parties agree that IBM shall indemnify and hold harmless the Indemnitee to the same extent as if the Claim was made by a non-employee of IBM, its contractors or subcontractors. IBM's indemnification hereunder shall be *. Accordingly, in addition to other provisions herein, and in order to render the Parties' intent and this indemnification agreement fully enforceable, IBM, in an indemnification claim hereunder, expressly and without reservation * it may have under any applicable * or any other statute or judicial decision * and consents to *. This waiver and consent * is made irrespective of and specifically * under any statute or judicial decision.

14.2 Indemnity by Equifax

Equifax will indemnify and hold harmless IBM, and its respective officers, directors, employees, agents, successors and assigns (each an "IBM Indemnitee") harmless from and against any and all Losses incurred by any of them arising from or in connection with:

- (a) any Claims of infringement asserted against an IBM Indemnitee as a result of * on or after the Commencement Date;
- (b) subject to Section 14.2(a) above, any Claims of infringement of any patent or any copyright, trademark, service mark, trade name, trade secret, or similar property right conferred by contract or by common law or by any law of any country, including without limitation, the United States and any other applicable jurisdiction or any state, alleged to have been incurred because of or arising out of any equipment, materials and other resources (including without limitation information technology, information management and communications services equipment, software or other resources) provided to IBM and/or its Affiliates by the Equifax Group in connection with the performance of the Services; provided, however, Equifax will have no obligation with respect to any Losses to the result of the Equifax Group * including, without limitation, obtaining * for which it has responsibility; and provided, further, that Equifax will have no obligation with respect to any Losses to the extent arising out of or in connection with * provided by the Equifax Group with * by the Equifax Group, or an IBM Indemnitee's * the Equifax Group to such IBM Indemnitee under this Agreement * or a written notice to IBM from Equifax *;
- * Information deleted pursuant to Rule 24b-2 of the Exchange Act.

50

- (c) any Claims accruing before the Commencement Date or after the termination date of this Agreement regarding any Third Party Agreements between members of the Equifax Group and a third party covered by this Agreement, including without limitation, failure to obtain Required Consents but not including Claims arising or resulting from IBM and/or its Affiliates failing to perform its obligations under this Agreement including, without limitation, obtaining any Required Consent for which it has responsibility;
- (d) any Claims for amounts, including without limitation, taxes, interest and penalties assessed or claimed against IBM which are obligations of Equifax under this Agreement;
- (e) any Claims for personal injuries, death or damage to tangible personal or real property of third parties including employees of the Equifax Group caused by the negligence or willful misconduct of the Equifax Group or their employees; provided that Equifax will have no obligation, under this part, to the extent the same arise out of or in connection with the negligence or willful misconduct of IBM, its Affiliates or subcontractors;
- (f) any Claims arising out of or resulting from the operations of the Equifax Group, including the provision of access to the Services pursuant to Section 17.14, to the extent such Claims do not arise out of a breach of this Agreement by IBM and are not the subject of a specific indemnity provided to Equifax by IBM in Section 14.1; provided, however, that Equifax will have no obligation under this item, to the extent the Claims arise out of or result from the negligence or willful misconduct of IBM, its Affiliates or subcontractors;
- (g) any Claim for violation of environmental laws or regulations arising out of the Services performed at the Facilities or other Equifax Group sites or locations to the extent that Equifax or its Affiliates or contractors (other than IBM and its Affiliates and subcontractors) has caused the environmental damage or violation of the environmental laws or regulations from which the Claim arises;
- (h) any Claims by any Transferred Employees in respect of which Equifax is obliged to indemnify IBM pursuant to Exhibit D-1, D-2, D-3, D-4 or D-5 of Schedule D (Human Resources); and
- (i) any Claims arising out of or resulting from the operations of the Equifax Group and arising from acts of Authorized Users.

In the event and to the extent that a Claim is made by * against an IBM Indemnitee, the Parties agree that Equifax shall indemnify and hold harmless the IBM Indemnitee to the same extent as if *. Equifax's indemnification hereunder shall be *. Accordingly, in addition to other provisions herein, and in order to render the Parties' intent and this indemnification agreement fully enforceable, Equifax, in an indemnification Claim hereunder, expressly and without reservation * it may have under any applicable * or any other statute or judicial decision *. This waiver and consent * is made irrespective of and specifically * under any statute or judicial decision.

14.3 Employment Actions

It is agreed that IBM shall be solely and exclusively responsible for personnel decisions affecting IBM's employees, contractors, subcontractors and agents (including without limitation, hiring, promotions, training, compensation, evaluation, discipline, and discharge). Equifax shall be solely and exclusively responsible for personnel decisions affecting employees, contractors, and agents of the members of the Equifax Group (including without limitation, hiring, promotion, training, compensation, evaluation, discipline and discharge).

14.4 Exclusive Remedy

Except as provided in Section 14.5, the indemnification rights of each Indemnitee and IBM Indemnitee (individually an "Indemnified Party") for third party Claims pursuant to Sections 14.1 and 14.2, shall be the sole and exclusive remedy of such Indemnified Party with respect to each such third party Claim to which such indemnification relates.

14.5 Infringement

If any item chosen for use by IBM to provide the Services becomes, or in IBM's reasonable opinion is likely to become, the subject of an infringement or misappropriation claim or proceeding, IBM shall, in addition to indemnifying Equifax as provided in this Article 14 and to the other rights Equifax may have under this Agreement, (a) promptly at IBM's expense secure the right to continue using the item, or (b) if this cannot be accomplished with commercially reasonable efforts, then at IBM's expense, replace or modify the item to make it non-infringing or without misappropriation, provided that any such replacement or modification will not degrade the performance or quality of the affected component of the Services, or (c) if neither of the foregoing can be accomplished by IBM with commercially reasonable efforts, and only in such event, then remove the item from the Services, in which case IBM's charges shall be equitably adjusted to reflect such removal.

14.6 Indemnification Procedures

(a) Written notice shall be given to the Party that is obligated to provide indemnification under Sections 14.1 and 14.2 (the "Indemnifying Party"), if any civil, criminal, administrative or investigative action or proceeding is commenced or threatened by a third party (any of the above being a "Claim") against any Indemnified Party. Such notice shall be given as promptly as practicable but in all events, within a period that will not prejudice the rights of the Indemnified Party under this Agreement to defend the Claim. After such notice, if the Indemnifying Party acknowledges in writing to the Indemnified Party that this Agreement applies with respect to such Claim, then the Indemnifying Party shall be entitled to take control of the defense and investigation of such Claim and to employ and engage attorneys of its sole choice to handle and defend the same, at the Indemnifying Party's sole cost and expense. The Indemnifying Party must deliver written notice of its election of taking such control of the claim to the Indemnified Party not fewer than ten (10) days prior to the date on which a response to such Claim is due or such lesser period as is reasonable given the nature of the Claim and the notice and response time permitted by law or the facts and circumstances. The Indemnified Party shall cooperate in all reasonable respects with the Indemnifying Party and its attorneys in the investigation, trial, defense and settlement of such Claim and any appeal arising therefrom. The Indemnified Party may participate in such investigation, trial, defense and settlement of such Claim and any appeal arising therefrom, through its attorneys or otherwise, at its own cost and expense. No settlement of a Claim that involves a remedy other than

52

the payment of money by the Indemnifying Party shall be entered into without the consent of the Indemnified Party, which consent will not be unreasonably withheld.

(b) After notice to the Indemnified Party of the Indemnifying Party's election to assume full control of the defense of any such Claim, the Indemnifying Party shall not be liable for any legal expenses incurred thereafter in connection with the defense of that Claim by the Indemnified Party. If the Indemnifying Party does not promptly assume full control over and diligently pursue the defense of a Claim as provided in this Section 14.5, the Indemnified Party shall have the right to defend, settle or otherwise resolve the Claim in such manner as it may deem appropriate, at the cost and expense of the Indemnifying Party, and the Indemnifying Party may participate in such defense, at its sole cost and expense. In no event shall any settlement of the Claim pursuant to this Section 5(b) require the consent of the Indemnifying Party.

14.7 Limitation

Notwithstanding anything to the contrary in this Agreement, the provisions of Section * of this Agreement shall not apply to the indemnification obligations of IBM pursuant to Section 14.1(j) of this Agreement.

15. INSURANCE AND RISK OF LOSS

15.1 IBM Insurance

During the Term of this Agreement, IBM and each IBM contractor and subcontractor shall maintain and keep in force, at its own expense, the following minimum insurance coverages and minimum limits:

- (a) workers' compensation insurance, with statutory limits as required by the various laws and regulations applicable to the employees of IBM or any IBM contractor or subcontractor;
- (b) employer's liability insurance, for employee bodily injuries and deaths, with a limit of \$* each accident;
- (c) comprehensive or commercial general liability insurance, covering claims for bodily injury, death and property damage, including premises and operations, independent contractors, products, services and completed operations (as applicable to the Services), personal injury, contractual, and broad-form property damage liability coverages, with limits as follows: (1) occurrence/aggregate limit of \$* for bodily injury, death and property damage per occurrence of \$* combined aggregate; or (2) split liability, without aggregate limits, of (i) \$* for bodily injury per person; (ii) \$* for bodily injury per occurrence; and (iii) \$* per occurrence for property damage;
- (d) comprehensive automobile liability insurance, covering owned, non-owned and hired vehicles, with limits as follows (1) combined single limit of \$* for bodily injury, death and property damage per occurrence; or (2) split liability limits of (i) \$* for bodily injury per person; (ii) \$* for bodily injury per occurrence;

Information deleted pursuant to Rule 24b-2 of the Exchange Act.

(e) all-risk property insurance, on a replacement cost basis, covering the real and personal property of IBM which IBM is obligated to insure by this Agreement. Such real and personal property may include buildings, equipment, furniture, fixtures and supply inventory.

All such policies of insurance of IBM and its contractors and subcontractors shall provide that the same shall not be canceled nor the coverage modified nor the limits changed without first giving thirty (30) days prior written notice thereof to Equifax. No such cancellation, modification or change shall affect IBM's obligation to maintain the insurance coverages required by this Agreement. Except for workers' compensation insurance, Equifax shall be named as an additional insured on all such required policies. All liability insurance policies shall be written on an "occurrence" policy form. Equifax shall be named as loss payee as its interest may appear on the property insurance policies of IBM. IBM shall be responsible for payment of any and all deductibles from insured claims under its policies of insurance. The coverage afforded under any insurance policy obtained by IBM pursuant to this Agreement shall be primary coverage regardless of whether or not Equifax has similar coverage. IBM and its contractors and subcontractors shall not perform under this Agreement without the prerequisite insurance. Upon Equifax's request, IBM shall provide Equifax with certificates of such insurance including renewals thereof. Unless previously agreed to in writing by Equifax, IBM's contractors and subcontractors shall comply with the insurance requirements herein. The minimum limits of coverage required by this Agreement may be satisfied by a combination of primary and excess or umbrella insurance policies. If IBM or its contractors or subcontractors shall fail to comply with any of the insurance requirements herein, upon written notice to IBM by Equifax and a ten (10) day cure period, Equifax may, without any obligation to do so, procure such insurance and IBM shall pay Equifax the cost thereof plus a reasonable administrative fee as designated by Equifax. The maintenance of the insurance coverages required under this Agreement shall in no way operate to limit the liability of IBM to Equifax under the provisions of this Agr

The Parties do not intend to shift all risk of loss to insurance. The naming of Equifax as additional insured is not intended to be a limitation of IBM's liability and shall in no event be deemed to, or serve to, limit IBM's liability to Equifax to available insurance coverage or to the policy limits specified in this Article 14, nor to limit Equifax's rights to exercise any and all remedies available to Equifax under contract, at law or in equity.

15.2 Risk of Property Loss

IBM is responsible for risk of loss of, or damage to, the Software, Machines, Equifax Provided Office Furnishings and Equifax Data in its possession or control, and Equifax is responsible for risk of loss of, or damage to, the Software, Machines and Equifax Data in its possession or control.

15.3 Mutual Waiver of Subrogation

(a) To the extent permitted by law, IBM and its Affiliates, contractors, subcontractors, and their respective directors, officers, employees, agents and insurers hereby waive their rights of subrogation against the member of the Equifax Group and their respective directors, officers, employees, agents, contractors and subcontractors for any loss or damage to the IBM Machines, IBM Software, and other tangible and intangible, real and personal property of IBM and its Affiliates, contractors and subcontractors resulting from operations in connection with this Agreement. Each property and worker's compensation insurance policy of IBM and its Affiliates, contractors and subcontractors shall be endorsed to provide a waiver of any and all rights of

54

subrogation against the Equifax Group and their respective directors, officers, employees, agents, contractors and subcontractors for loss resulting from operations in connection with this Agreement.

(b) To the extent permitted by law, Equifax, the other members of the Equifax Group and their respective directors, officers, employees, agents and insurers hereby waive their rights of subrogation against IBM and its Affiliates, contractors and subcontractors for any loss or damage to the Equifax Provided Hardware, Equifax Software, Equifax Provided Office Furnishings and other tangible and intangible, real and personal property of Equifax and the other members of the Equifax Group resulting from operations in connection with this Agreement. Each property and worker's compensation insurance policy of Equifax shall be endorsed to provide a waiver of any and all rights of subrogation against IBM and its Affiliates, contractors and subcontractors for loss resulting from operations in connection with this Agreement.

16. DISPUTE RESOLUTION

16.1 Dispute Resolution Procedures

- (a) Subject to Schedule L (Governance), any dispute between the Parties either with respect to the interpretation of any provision of this Agreement or with respect to the performance hereunder by IBM or by Equifax or their respective Affiliates shall be resolved as specified in this Section 16.1.
- (i) Upon the written request of either Party, a dispute shall be submitted to the Global Governance Team for resolution.
- (ii) The Global Governance Team shall meet as often as necessary to gather and furnish to each Party all non-privileged information with respect to the matter in issue which is appropriate and germane in connection with its resolution.
- (iii) The Global Governance Team shall discuss the problem and negotiate in good faith in an effort to resolve the dispute without the necessity of any formal proceeding relating thereto.
- (iv) During the course of such negotiation, all reasonable requests made by one Party to the other for non-privileged information reasonably related to this Agreement, will be honored in order that each Party may be fully advised of the other Party's position.
- (v) The specific format for such discussions will be left to the discretion of the Global Governance Team, but may include the preparation of agreed upon statements of fact or written statements of position furnished by each Party to the other Party.
 - (b) If the Global Governance Team does not resolve the dispute within thirty (30) days after the date of receipt by the other Party of a request to submit the dispute to the Global Governance Team as described in Section 16.1(a)(1) (the "Notice"), then the dispute shall be escalated to an officer of Equifax and an officer of IBM, for their review and resolution within forty-five (45) days after receipt of the Notice.

- (c) If the officers referred to in Section 16.1(b) do not resolve the dispute within forty-five (45) days after the Notice, then the dispute shall be escalated to the President of Equifax and the IBM corporate officer in charge of IBM Global Services, for their review and resolution within sixty (60) days after the Notice.
- (d) If the dispute is not resolved by the Parties' representatives identified in Section 16.1(c) within ninety (90) days after the Notice, the Parties agree to try in good faith to resolve the dispute by

55

mediation under the Commercial Mediation Rules of the American Arbitration Association, before resorting to litigation or some other dispute resolution procedure.

- (e) If the dispute is not resolved by mediation within one hundred twenty (120) days after the Notice, then the Parties may initiate formal proceedings; however, formal proceedings for the judicial resolution of any such dispute may not be commenced until the earlier of:
- (i) the designated representatives concluding in good faith that amicable resolution through continued negotiation of the matter in issue does not appear likely; or
- (ii) one hundred twenty (120) days after the Notice; or
- (iii) thirty (30) days before the statute of limitations governing any cause of action relating to such dispute would expire.

Notwithstanding anything to the contrary in this Section 16.1(e), the Global Governance Team shall have the authority to stay the time periods set forth in this Section 16.1 upon unanimous vote of its members to take such action.

(f) Notwithstanding any other provision of this Section 16.1, either Party may resort to court action for injunctive relief at any time if the dispute resolution processes set forth in this Section would permit or cause irreparable injury to such Party or any third party claiming against such Party, due to delay arising out of the dispute resolution process.

16.2 Continued Performance

The Parties agree to continue performing their respective obligations under this Agreement while the dispute is being resolved unless and until such obligations are terminated or expire in accordance with the provisions of this Agreement or unless such performance is prevented by the actions of the other Party; provided that Equifax withholding payment of disputed charges as permitted under Section 8.2 (Disputed Charges/Credits) of Schedule C (Charges) will not considered to be preventing IBM from performing its obligations.

17. GENERAL

17.1 Relationship of Parties

This Agreement shall not be construed as constituting either Party or its Affiliates as partner of the other Party and its Affiliates or to create any other form of legal association that would impose liability upon one Party or its Affiliates for the act or failure to act of the other Party and its Affiliates or as providing either Party or its Affiliates with the right, power or authority (express or implied) to create any duty or obligation of the other Party and its Affiliates, except as provided in Section 8.3. Each Party shall be responsible for the management, direction and control of the employees of such Party and its Affiliates and such employees shall not be employees of the other Party or its Affiliates.

Each Party will submit to the other Party all advertising, written sales promotion, press releases and other publicity matters relating to this Agreement in which the other Party's or its Affiliate's name or mark is mentioned or language from which the connection of said name or mark may be inferred or implied, and will not publish or use such advertising, sales promotion, press releases, or publicity matters without prior written approval of the other Party. However, either Party may include the other Party's and/or its Affiliates name and a factual description of the work performed under this Agreement on employee bulletin boards, in its list of references and in the

56

experience Section of proposals to third parties, in internal business planning documents and in its annual report to stockholders, and whenever required by reason of legal, accounting or regulatory requirements.

17.2 Entire Agreement, Updates, Amendments and Modifications

This Agreement (including the Schedules thereto and any Statements of Work issued under it) constitutes the entire agreement of the Parties and their Affiliates with regard to the Services and matters addressed therein. This Agreement supersedes and replaces, the agreements listed in Schedule M (Existing IBM-Equifax Agreements Superseded by the Agreement). All letters, proposals, discussions and other documents regarding the Services and the matters addressed in this Agreement are superseded by and merged into this Agreement. Updates, amendments and modifications to this Agreement may not be made orally, but shall only be made by a written document signed in the case of this Agreement by duly authorized representatives of both Parties. In the case of Equifax, only the Equifax Inc. Chief Technology Officer or his designee shall have contract signature authority with respect to this Agreement. Any terms and conditions varying from this Agreement on any order or written notification from either Party or its Affiliates shall not be effective or binding on the other Party or its Affiliates.

17.3 Force Majeure

(a) Neither Party shall be liable for any default or delay in the performance of its obligations hereunder provided that the non-performing Party is without material fault in causing such default or delay, except for payment defaults, if and to the extent and while such default or delay is caused, directly or indirectly, by fire, flood, earthquake, elements of nature or acts of God, acts of war, terrorism, riots, civil disorders, rebellions or revolutions, strikes, lockouts, or labor difficulties or any other similar cause beyond the reasonable control of such Party and its Affiliates other than strikes, lockouts, non-performance by third parties for which the non-performing party is responsible (provide that such third parties are not subject to a Force Majeure Event) or labor difficulties initiated by such Party's or its Affiliates or subcontractor's employees; and provided such default or delay could not have been prevented by reasonable precautions and cannot

reasonably be circumvented by the non-performing Party or its Affiliates through the use of alternate sources, work-around plans or other means, (individually, each being a "Force Majeure Event").

- (b) If a Force Majeure Event occurs, the non-performing Party will be excused from any further performance or observance of the obligation(s) so affected for as long as such circumstances prevail and such Party continues to use Commercially Reasonable Efforts to recommence performance or observance whenever and to whatever extent possible without delay. Any Party so delayed in its performance will immediately notify the other by telephone and describe at a reasonable level of detail the circumstances causing such delay (to be confirmed in writing within twenty-four (24) hours after the inception of such delay).
- (c) If any Force Majeure Event substantially prevents, hinders, or delays performance of the Services necessary for the performance of the critical functions of the Equifax users, as identified by Equifax, of such Services for more than three (3) consecutive days, then at Equifax's option:
- (i) Equifax may procure such Services from an alternate source. Provided Equifax has not terminated this Agreement to Section 17.3(c)(iii) and Equifax continues to make payment to IBM under this Agreement and Equifax exerts reasonable efforts to mitigate amounts payable to the alternate source, IBM will directly and timely pay the alternate source the full amount charged by such alternate source for the

57

provision of such Services to Equifax until such time as IBM restores the Services and meets the Service Levels but in no event for more than one hundred eighty (180) days; and/or

- (ii) Equifax may terminate any portion of this Agreement so affected and the charges payable hereunder shall be equitably adjusted to reflect those terminated Services; and/or
- (iii) Until such time as IBM has restored the Services, Equifax may terminate this Agreement as of a date specified by Equifax in a written notice of termination to IBM, and Equifax will pay all charges due and payable through the termination date. If Equifax elects such termination, Equifax shall only pay on account of such termination IBM's Wind Down Expenses for a period not to exceed one hundred twenty (120) days and shall acquire from IBM, at a cost equal to IBM's then-current net book value, and hardware or software primarily dedicated to performing Services for Equifax that cannot be redeployed by IBM using Commercially Reasonable Efforts.
 - (d) This Section 17.3 does not limit or otherwise affect IBM's obligation to provide Disaster Recovery Services in accordance with Section 3.3. In the event of a Force Majeure Event affecting Equifax, this Section 17.3 will not limit or otherwise relieve Equifax's obligation to pay any monies due IBM under the terms of this Agreement, except as provided in Section 17.3(c)(iii) and Section 3.3.

17.4 Waiver

No waiver of any breach of any provision of this Agreement shall constitute a waiver of any prior, concurrent or subsequent breach of the same or any other provisions hereof.

17.5 Severability

If any provision of this Agreement shall be held to be invalid, illegal or unenforceable, the validity, legality and enforceability of the remaining provisions shall not in any way be affected or impaired thereby, and such provision shall be deemed to be restated to reflect the Parties' original intentions as nearly as possible in accordance with applicable law(s).

17.6 Counterparts

This Agreement shall be executed in counterparts. Each such counterpart shall be an original and together shall constitute but one and the same document.

17.7 Governing Law

This Agreement and any and all claims and disputes arising out of or in connection with or related to the relationships and arrangements between the Equifax Group and IBM and its Affiliates described in this Agreement will be governed by and construed in accordance with the laws of the State of Georgia and the United States of America except where local mandatory law applies. The Parties hereby (a) agree that the U.S. District Court for the Northern District of Georgia, Atlanta Division, or if such court does not have subject matter jurisdiction, the appropriate State or Superior Court sitting in Fulton County, Georgia, shall have exclusive jurisdiction over the actions arising out of or related to or in connection with this Agreement and the subject matter of this Agreement, whether in contract, tort, or any other form of action ("Action"); (b) agree to initiate any such Action against the other Party only in such courts; (c) agree that they shall not raise any defense to the lawful jurisdiction of such courts; and (d) agree that they shall not attempt the removal of any Action to any other court, whether local, state or federal courts of the United States or the courts of any other country.

58

17.8 Binding Nature and Assignment

This Agreement will be binding on the Parties and their respective successors and permitted assigns. Except as provided in this Section 17.8, neither Party may, or will have the power to, assign this Agreement without the prior written consent of the other, which consent shall not be unreasonably withheld, except that either Party may assign its rights and obligations under this Agreement, without the approval of the other Party (i) in whole or in part to an Affiliate which expressly assumes such Party's obligations and responsibilities hereunder; or (ii) as part of a merger, acquisition or sale of substantially all of the assets of a Party. The assigning Party shall remain fully liable for and shall not be relieved from the full performance of all obligations under this Agreement. Any attempted assignment that does not comply with the terms of this Section 17.8 shall be null and void. Any Party assigning its rights or obligations to an Affiliate in accordance with this Agreement shall provide written notice thereof to the other Party together with a copy of the assignment document, within three (3) business days of such assignment Notices.

17.9 Notices

(a) Whenever one Party is required or permitted to give legal notice to the other Party under this Agreement, such legal notice will be in writing unless otherwise specifically provided herein and will be deemed given when delivered in hand, one (1) day after being given to an express courier with a reliable system for tracking delivery, or five (5) days after the day of mailing, when mailed by United States mail, registered or certified mail, return receipt requested, postage prepaid, or when sent if delivered by facsimile. Operational notices may be sent via email.

(b) Legal notifications, including Contract Change notices, will be addressed as follows:

In the case of Equifax:
Global Program Manager
1525 Windward Concourse
Alpharetta, Georgia 30005
Facsimile: 770-740-7952
with a copy to:
Equifax Chief Legal Officer
1600 Peachtree Street, N.W.
Atlanta, Georgia 30309
Facsimile: 404-885-8682
and to:
AVP — Technology Contract
Management
Equifax Inc.
1525 Windward Concourse 30005
Alpharetta, Georgia

(c) Operational notifications will be addressed to the IBM Regional PE and the Equifax Regional PM in the applicable Country Location.

59

Either Party hereto may from time to time change its address for notification purposes by giving the other prior written notice of the new address and the date upon which it will become effective.

Fax: 770-740-7213

17.10 No Third Party Beneficiaries

The Parties do not intend, nor will any Section hereof be interpreted, to create for any third party beneficiary rights with respect to either of the Parties, except (a) each member of the Equifax Group and each IBM Affiliate shall be a third party beneficiary under this Agreement with respect to enforcement of any rights such member of the Equifax Group or IBM Affiliate may have under Article 10, Article 11, or Article 14 of this Agreement, and (b) an Affiliate of the Parties to which this Agreement has been assigned and accepted, will have the rights and benefits described this Agreement, and (c) the third parties identified in Article 14 will have the rights and benefits described in that Article.

17.11 Other Documents

Upon request of the other Party, on or after the Execution Date and amendments or revisions to any of the foregoing, each Party shall furnish to the other such certificate of its Secretary, certified copy of resolutions of its Board of Directors, or opinion of its counsel as shall evidence that this Agreement or any amendment or revision hereto has been duly executed and delivered on behalf of such Party or its Affiliates.

17.12 Consents and Approvals

The Parties agree that in any instance where a consent, approval or agreement is required of a Party in order for the other Party to perform under or comply with the terms and conditions of this Agreement, then such Party will not unreasonably withhold or delay such consent, approval or agreement and where consent, approval or agreement cannot be provided, the Party shall notify the other Party in a timely manner.

17.13 Headings

All headings herein and the table of contents are not to be considered in the construction or interpretation of any provision of this Agreement. The Agreement was drafted with the joint participation of both Parties and shall be construed neither against nor in favor of either, but rather in accordance with the fair meaning thereof. In the event of any apparent conflicts or inconsistencies between the provisions of this Agreement, the Exhibits, the Schedules or other attachments to this Agreement, such provisions shall be interpreted so as to make them consistent to the extent possible, and if such is not possible, the provisions of the Agreement shall prevail.

17.14 Remarketing

Equifax may not remarket all or any portion of the Services provided under this Agreement, or make all or any portion of the Services available to any party that is not a member of the Equifax Group, without the prior written consent of IBM; provided, however, that Equifax may sell or make available to Authorized Users and persons or entities acquiring portions of the

60

Equifax Business from Equifax or its Affiliates access to elements of the Services under this Agreement ("Elements of the Services") subject to the following limitations:

- (a) Equifax shall independently set its own pricing and policies in connection with any such access to Elements of the Services;
- (b) Equifax does not utilize IBM's name as part of its marketing efforts regarding any such access to Elements of the Services;
- (c) Equifax discloses to its customers or Authorized Users accessing Elements of the Services that IBM is running the Systems but that IBM has no liability of

any kind to such customers;

- (d) if Equifax's activities for a customer or Authorized User accessing Elements of the Services cause IBM to fail to meet a Service Level, IBM shall be excused from such failure to the extent, and only to the extent, IBM demonstrates that the failure was caused directly by such customer's or Authorized User's activities and to the extent such failure was not caused by IBM's failure to satisfy its obligations under this Agreement; and
- (e) Use of Elements of the Services by Authorized Users in connection with the Equifax Business and for the benefit of the Equifax Group is included in IBM's charges under Schedule C (Charges) except to the extent, if any, that such use of elements of the Services requires the IBM to perform New Services pursuant to Section 10.1 (Prices and Charges for New Services) of Schedule C (Charges).
- (f) Nothing herein may be construed to limit or hinder Equifax or the other members of the Equifax Group from (i) marketing, selling or performing its services to and for its customers or potential customers and/or (ii) from providing any portion of the Services to its Affiliates.

17.15 Commencement of Actions

Neither party may bring an action, regardless of form, arising out of this Agreement more than two (2) years after the later to occur of the date on which the cause of action has arisen or the date such cause of action was or should have been discovered.

17.16 IBM Logo Products Warranties

Nothing in this Agreement is intended to replace, supercede or vitiate the warranties and attendant rights and remedies granted to members of the Equifax Group by IBM and/or its Affiliates with respect to IBM Logo Products as set forth in any applicable lease, purchase and/or license arrangement.

17.17 Local Enabling Agreements

- (a) It is the intention of the Parties that this Agreement shall provide for the provision of Services at the Country Locations provided herein, including those which may be located outside of the United States. Unless this Agreement expressly provides otherwise, Services to be provided hereunder to Equifax Affiliates outside of the United States shall be provided by the IBM Affiliate in the same country in which the Equifax Affiliate is located.
- (b) IBM's applicable local Affiliate shall enter into an enabling agreement in substantially the form agreed to by the Parties (a "Local Enabling Agreement"). Such Local Enabling Agreements may supplement the Schedules to this Agreement as appropriate (e.g., by updating the list of Systems Software to include any different or additional Systems Software used at the newly included site or

61

by the newly included Affiliate) and may include country-specific terms and conditions that are necessary to comply with local law.

(c) Without Equifax's consent, no Local Enabling Agreement shall alter the net amounts to be received by IBM, expand the geographic scope, or otherwise alter the Parties' respective rights and obligations under this Agreement. No Local Enabling Agreement shall be effective unless it is approved by the Equifax Global Program Manager and the IBM Global Project Executive. Each Party shall have joint and several liability for the respective obligations of its Affiliates under Local Enabling Agreements.

(main body of Agreement ends here)

62

QuickLinks

AGREEMENT FOR OPERATIONS SUPPORT SERVICES TABLE OF CONTENTS

CERTIFICATION PURSUANT TO RULE 13a-14(a) OR 15d-14(a) OF THE EXCHANGE ACT

- I, Thomas F. Chapman, Chairman and Chief Executive Officer of Equifax Inc., certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of Equifax Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 12, 2003	/s/ THOMAS F. CHAPMAN
	Thomas F. Chapman Chairman and Chief Executive Officer

QuickLinks

CERTIFICATION PURSUANT TO RULE 13a-14(a) OR 15d-14(a) OF THE EXCHANGE ACT

CERTIFICATION PURSUANT TO RULE 13a-14(a) OR 15d-14(a) OF THE EXCHANGE ACT

- I, Donald T. Heroman, Chief Financial Officer of Equifax Inc. (the "registrant") certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of Equifax Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - c) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 12, 2003	/s/ DONALD T. HEROMAN	_
	Donald T. Heroman Chief Financial Officer	-

QuickLinks

CERTIFICATION PURSUANT TO RULE 13a-14(a) OR 15d-14(a) OF THE EXCHANGE ACT

CERTIFICATION PURSUANT TO 18 U. S. C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Company's Quarterly Report on Form 10-Q for the quarterly period ended September 30, 2003 (the "Report"), I, Thomas F. Chapman, Chairman and Chief Executive Officer of Equifax Inc. (the "Company") do hereby certify, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350, that:

- (i) the Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (ii) the information contained in the Report fairly represents, in all material respects, the financial condition and results of operation of the Company.

/s/ THOMAS F. CHAPMAN

Thomas F. Chapman Chairman and Chief Executive Officer

Dated: November 12, 2003

A signed original of this statement required by Section 906 has been provided to Equifax Inc. and will be retained by Equifax Inc. and furnished to the Securities and Exchange Commission or its staff upon request.

QuickLinks

CERTIFICATION PURSUANT TO 18 U. S. C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

CERTIFICATION PURSUANT TO 18 U. S. C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Company's Quarterly Report on Form 10-Q for the quarterly period ended September 30, 2003 (the "Report"), I, Donald T. Heroman, Chief Financial Officer of Equifax Inc. (the "Company") do hereby certify, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350, that:

- (i) the Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (ii) the information contained in the Report fairly represents, in all material respects, the financial condition and results of operation of the Company.

/s/ DONALD T. HEROMAN

Donald T. Heroman Chief Financial Officer

Dated: November 12, 2003

A signed original of this statement required by Section 906 has been provided to Equifax Inc. and will be retained by Equifax Inc. and furnished to the Securities and Exchange Commission or its staff upon request.

QuickLinks

CERTIFICATION PURSUANT TO 18 U. S. C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002